The Real Effects of Money Growth in the Long Run

An Analysis of a Schumpeterian Growth Model with Nominal Price Rigidity

Inaugural dissertation

zur

Erlangung des Doktorgrades

 der

Wirtschafts- und Sozialwissenschaftlichen Fakultät

 der

Universität zu Köln

2006

vorgelegt von

Dipl.-Volksw. Bettina Kromen

aus Köln

Referent: Prof. Dr. Peter Funk

Korreferent: Prof. Dr. Clemens Fuest

Tag der Promotion: 18.01.2007

Danksagung

Die Anfertigung einer Doktorarbeit ist ohne die Mithilfe und vielfältige Unterstützung von vielen Personen nicht möglich.

Mein besonderer Dank gebührt meinem Doktorvater Prof. Dr. Peter Funk. Er hat mir die Anregung zur Beschäftigung mit dem Thema der Arbeit gegeben und mich während der Arbeit an der Dissertation nach Kräften gefördert und gefordert. Für die fortwährende aktive Unterstützung unter anderem durch viele hilfreiche Freitagsdiskussionen danke ich ihm sehr.

Prof. Dr. Clemens Fuest danke ich für die Übernahme des Zweitgutachtens. Weiterhin danke ich Prof. Dr. Achim Wambach für den Vorsitz der Disputation.

Meinen ehemaligen Kollegen am Lehrstuhl, Anna Bode, Peter Kojetinsky und Dr. Thorsten Vogel, möchte ich für die angenehme Atmosphäre und hilfreiche Diskussionen danken.

Ohne die moralische Unterstützung meiner Familie wäre diese Arbeit nicht beendet worden. Ich danke besonders meinen Eltern herzlich für die immerwährende Unterstützung während meiner gesamten Ausbildungszeit. Meinen Brüdern Dr. Wolfgang Kromen und Dr. Winfried Kromen sowie meiner Schwägerin Angela Westdorf danke ich sowohl für seelische Unterstützung als auch für Hilfe beim Korrekturlesen der Arbeit. Meinen Brüdern gebührt darüber hinaus Dank für hilfreiche Diskussionen über mathematischstatistische Probleme der Arbeit.

Nicht zuletzt möchte ich meinem Freund Thomas Hemmelgarn danken, der die Höhen und Tiefen meiner Promotion wohl am intensivsten miterlebt hat und während der ganzen Zeit immer mein Fels in der Brandung war.

Köln, im März 2007

Bettina Kromen

1	The	e real e	ffects of money – an overview	1
	1.1	Introd	uction	1
	1.2	The re	eal effects of money – some definitions	3
	1.3	Money	r and the levels of output and employment	4
		1.3.1	Money in the business cycle	4
		1.3.2	Inflation, employment and output in the long run: A	
			non-vertical Phillips curve	17
	1.4	Money	r growth, inflation and the growth rate of output \ldots .	23
		1.4.1	Survey of the theoretical literature	23
		1.4.2	Empirical evidence	31
	1.5	Money	growth, price rigidity and growth	33
		1.5.1	Attempting a long-run synthesis	33
		1.5.2	Short outline of the remaining chapters	37
2	Money growth and Innovation-driven growth			42
	2.1	Introd	uction \ldots	42
	2.2	The m	odel	46
		2.2.1	Final good sector	47
		2.2.2	Intermediate goods sector	48
		2.2.3	Price level and real wage	56
		2.2.4	Patents and the R&D sector	58
		2.2.5	Labour market equilibrium	60
		2.2.6	Production side equilibrium	61
		2.2.7	Public Sector	61

		2.2.8	Assets and Households	62	
	2.3	Gener	al equilibrium	64	
		2.3.1	Closing the model	64	
		2.3.2	Existence and uniqueness of the steady state equilibrium	66	
	2.4	Comp	arative statics	68	
		2.4.1	Economy without price rigidity: Superneutrality of money	69	
		2.4.2	Economy with price rigidity: Negative growth effect of		
			money growth	70	
		2.4.3	Economy with price rigidity: Growth effect of changes		
			in the level of rigidity	74	
	2.5	Calibr	ating the model	75	
	2.6	Conclu	usion	78	
	2.A	Appen	ndix to chapter $2 \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	79	
		2.A.1	Derivation of a new firm's market value	79	
		2.A.2	Derivation of labour demand $L^{X}(\tau)$	81	
		2.A.3	Derivation of equation $2.13 \dots \dots \dots \dots \dots \dots$	83	
		2.A.4	Derivation of the growth rate of the Quality index	83	
		2.A.5	Proofs of propositions	85	
3	Loc	ocal Stability 8			
	3.1	Introd	uction	87	
	3.2	Key e	quations for the economy's off-steady-state behaviour	88	
		3.2.1	Firms	88	
		3.2.2	Labour market equilibrium	93	
		3.2.3	Households	95	
		3.2.4	Money market equilibrium	96	
	3.3	The li	inearised dynamical system and local stability of the		
		steady	r state equilibrium $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	96	
	3.A	Appen	ndix to chapter $3 \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	99	
		3.A.1	Linearising the optimal price	99	
		3.A.2	A new intermediate good firm's market value	102	
		3.A.3	Deriving the linearised zero-profit-condition	103	
		3.A.4	Eliminating $\widetilde{P}^{*}(\tau)$ from the system	108	

		3.A.5	Eliminating $\frac{m(\tau)}{Q(\tau)}$ from the system $\ldots \ldots \ldots \ldots$	109
		3.A.6	Rewriting the system	110
		3.A.7	Elements of the linearised system's matrix	114
4	Lim	mit Pricing 116		
	4.1	Introdu	uction \ldots \ldots \ldots \ldots \ldots \ldots \ldots	116
	4.2	The m	odel with limit pricing	118
		4.2.1	The limit price \ldots \ldots \ldots \ldots \ldots \ldots	118
		4.2.2	A new firm's market value at entry	119
		4.2.3	Labour demands and labour market equilibrium	120
		4.2.4	The zero-profit-condition in general equilibrium \ldots	120
	4.3	Exister	nce and uniqueness of steady state equilibrium under	
		limit p	ricing	121
	4.4	Compa	arative statics and a calibrated example \ldots \ldots \ldots	122
		4.4.1	Effect of money growth on output growth \ldots	123
		4.4.2	Effect of price rigidity on output growth	124
		4.4.3	Calibrated example	124
	4.5	Conclu	sion	125
	4.A	Appen	dix to chapter 4	126
5	Money growth, employment and growth 128			.28
	5.1	Introdu	uction \ldots \ldots \ldots \ldots \ldots \ldots \ldots	128
	5.2	The m	odel	133
		5.2.1	Final good sector	133
		5.2.2	Intermediate goods sector	134
		5.2.3	Patents and the R&D sector	135
		5.2.4	Public Sector	136
		5.2.5	Consumption and money demand \ldots \ldots \ldots \ldots	136
		5.2.6	Labour supply	137
	5.3	Steady	state equilibrium	138
		5.3.1	Households	139
		5.3.2	Money market equilibrium	139

		5.3.3	Behaviour of the aggregate quality index $Q(\tau)$ and the	
			growth rate	139
		5.3.4	Equilibrium in the market for intermediate goods	140
		5.3.5	Equilibrium in the final good market	145
		5.3.6	Labour market equilibrium given the innovation rate μ_{-}	147
		5.3.7	Research market equilibrium given employment L	151
	5.4	Existe	nce and uniqueness of the steady state equilibrium \ldots .	158
	5.5	Comp	arative statics	159
		5.5.1	Money growth and employment \ldots \ldots \ldots \ldots	160
		5.5.2	Money growth and output	163
		5.5.3	Money growth and economic growth	163
		5.5.4	Limited trade-off for monetary policy between employ-	
			ment and growth	166
		5.5.5	A digression: Explaining the different growth effect of	
			deflation in the models of chapters 2 and 5	167
		5.5.6	Comparative statics regarding the level of price rigidity	169
	5.6	Conclu	usion	170
	5.A	Appen	$ model{eq:model} modele{eq:model} modele{eq:$	171
		5.A.1	A new firm's market value	171
		5.A.2	Total intermediate good production	172
		5.A.3	Total final good production	172
		5.A.4	Proofs of propositions	174
6	Con	cludin	g remarks 1	178

iv

List of Figures

2.1	Steady state equilibrium research intensity μ
2.2	Comparative statics for $\beta < \infty$: money growth rate 69
2.3	Function $\gamma(\psi)$ in baseline calibration
5.1	Partial equilibrium research intensity μ given employment $\overline{L}~$. 153
5.2	Effects on the partial equilibrium research intensity given em-
	ployment $\mu\left(\psi, \overline{L}, \beta\right)$ of an increase in ψ when inflation is pos-
	itive
5.3	Existence and uniqueness of general steady state equilibrium
	research intensity mu in our leading example
5.4	Partial and total effect of money growth on employment \ldots 162
5.5	Output as a function of the money growth rate
5.6	Economic growth rate as a function of the money growth rate 165
5.7	Trade-off between employment and growth

List of abbreviations

- NNS New Neoclassical Synthesis
- RBC Real Business Cycles

List of Tables

2.1	Parameter values used for calibration	77
3.1	Parameter values used in local stability analysis	98
5.1	Parameter values used for calibration of the model with endo-	
	genous labour supply	160

Chapter 1

The real effects of money – an overview

1.1 Introduction

The following quotation from David Hume's essay "Of money" from the year 1752 nicely summarises a fundamental puzzle that has engaged the thoughts of many generations of economists: How is it that changes in money – an intrinsically valueless good that serves as a unit of account and medium of exchange in transactions – have an effect on the real level of activity in an economy? In the words of Hume:

"It is indeed evident, that money is nothing but the representation of labour and commodities, and serves only as a method of rating or estimating them. Where coin is in greater plenty; as a greater quantity of it is required to represent the same quantity of goods; it can have no effect, either good or bad, taking a nation within itself; any more than it would make an alteration on a merchant's books, if, instead of the Arabian method of notation, which requires few characters, he should make use of the Roman, which requires a great many. (...) But notwithstanding this conclusion, (...) we find, that, in every kingdom, into which money begins to flow in greater abundance than formerly, every thing takes a new face: labour and industry gain life; the merchant becomes more enterprising, the manufacturer more diligent and skilful, and even the farmer follows his plough with greater alacrity and attention."¹

Since the days of Hume, an enormous number of theories have been put forward to explain this puzzle.² The goal of the present thesis is to contribute to this literature by enhancing the understanding of money's effects on the real side of the economy in the long run. In particular, the following chapters will analyse the influence of the growth rate of money supply on an economy's growth rate of output and on its levels of output and employment in a framework of a Schumpeterian model of innovation-driven growth augmented with a New Keynesian specification of nominal price rigidity.

In the context of this project, the aim of the present chapter is threefold: The first aim is to present some empirical results about the relationship between monetary variables, in particular inflation, the level and growth rate of money supply, and real variables, in particular the levels of employment and output and the latter's growth rate. The second goal is to introduce the reader to the main explanations offered by the modern theoretical literature for the correlations found in the data between the mentioned nominal and real variables. Although the analysis of this thesis is restricted to the longrun influence of money, some stylised facts on the short-run influence of money and a presentation of the modern theoretical approach to analysing money's role in the business cycle are included. The reason for doing so is related to the third aim of the present chapter, which is to demonstrate that the subsequent analysis is actually a synthesis of elements from the New Keynesian literature dealing with the short-term effects of money and from long-run oriented endogenous growth theory. The reader will along the way be introduced to most of the central building blocks of the subsequent chapters models' while also learning about the methodological differences between the approach of this thesis and the existing literature.

¹Hume [1955] p. 37, emphasis omitted.

²See Blanchard [1990] and Lucas [1996] for recent surveys.

The remainder of this chapter is organised as follows: Section 1.2 presents some helpful definitions. In section 1.3, the reader is introduced to the analysis of the relationship between the money supply and its growth rate on the one hand and the *levels* of output and employment on the other. In particular, in section 1.3.1 some stylised facts about money's role in the business cycle are presented and the New Keynesian business cycle model is introduced which as stated above contains several of the building blocks of the models of this thesis. Money's level effects in the long run are analysed in section 1.3.2 with an emphasis on the New Keynesian literature. Some empirical results complement the theoretical analysis. Section 1.4 is concerned with the effects of the growth rate of money supply on the *growth rate* of output. Theoretical explanations based on neoclassical and endogenous growth theory frameworks are discussed in section 1.4.1. Section 1.4.2 presents recent empirical findings. Finally, section 1.5 introduces the reader to the approach taken in this thesis. Section 1.5.1 discusses the motivation for the intended synthesis and presents some evidence on its empirical relevance. Section 1.5.2 gives an overview of the remaining chapters of this thesis.

1.2 The real effects of money – some definitions

The title of this chapter refers very generally to the real effects of "money". We now introduce some central definitions which help to clarify which specific changes in nominal variables are conjectured to have an effect on real variables.

If changes in both the level and the growth rate of the nominal money supply affect only nominal variables such as inflation and the nominal interest rate, money is said to be both neutral and superneutral.³ Conversely, any changes in real variables caused by variations in the *level* of money supply are said to be evidence of money's *non-neutrality*, whereas any effects on

 $^{^{3}}$ See Orphanides and Solow [1990]. The nominal money supply is in the context of the literature understood to be a narrow monetary aggregate under direct control of the monetary authority.

the real side of the economy brought about by changes in the *growth rate* of money supply are manifestations of money's *non-superneutrality*.

Given that inflation is considered to be a monetary phenomenon entirely determined by money growth in the long run,⁴ a shortcut is sometimes taken and the effects of inflation on the real economy are directly discussed. In particular, the term "*Phillips curve*" is used to refer to a conjectured effect of inflation on unemployment.⁵

Some theory and evidence concerning all these neutrality propositions will be discussed in this chapter.

In the models to be presented in the subsequent chapters of this thesis, money is neutral in the long run.⁶ Consequently, our analysis focuses on the non-superneutrality of money with respect to the economy's output growth rate and on the effects of the money growth rate and inflation on the economy's levels of employment and output, where the latter relations will be discussed under the heading of a long-run Phillips curve.

1.3 Money and the levels of output and employment

1.3.1 Money in the business cycle

Various measures of money supply are positively correlated with contemporaneous output and employment in the short run. Also, output is positively correlated with lagged values of money supply, indicating that money leads output in the business cycle.⁷

⁴There is a nearly complete long-run correlation between various measures of the money supply growth rate and the inflation rate. E.g., McCandless and Weber [1995] report correlations between 0.92 and 0.96. Besides, money growth Granger-causes inflation, see Crowder [1998].

⁵Somewhat confusingly, inflation rate and growth rate of money supply are sometimes used synonymously in the literature studying whether money is superneutral.

⁶In the context of an empirical study, Fisher and Seater [1993] show that neutrality is a necessary but not sufficient condition for the superneutrality of money.

⁷See, e.g. Walsh [2003], pp. 12-13.

These are two of the well-documented empirical regularities that characterise business cycle fluctuations in real economies.⁸

By themselves, these correlations say nothing about the underlying causal relationship between money and output, i.e. about the neutrality of money. Starting with the seminal study of Friedman and Schwartz [1963], researchers have used a variety of methodological approaches to show that causality runs mainly from money to output.⁹ While there are still some voices arguing for "reverse causation", i.e. changes in the money supply being caused by output,¹⁰ there is now a broad consensus that money is non-neutral in the sense that increases in money supply cause a hump-shaped reaction in output over the following quarters.^{11,12} In particular, the vector autoregression (VAR) framework has been used in recent years to investigate the influences of monetary policy shocks on output.¹³ Christiano et al. [1999] (p. 69) summarise the consensus in this literature about the short-run non-neutrality of money:¹⁴

"there is considerable agreement about the qualitative effects of a monetary policy shock (...): after a contractionary monetary policy shock, short term interest rates rise, aggregate output, employment, profits and various monetary aggregates fall, the aggregate price level responds very slowly, and various measures of wages fall, albeit by very modest amounts."

⁸See Stock and Watson [2000] for a detailed presentation of the stylised facts of the US business cycle. Cooley/Hansen [1995] focus on those regularities concerning nominal variables.

⁹See the survey and discussion in chapter 1 of Walsh [2003].

 $^{^{10}\}mathrm{See}$ e.g. King and Plosser [1984].

¹¹See e.g. Sims [1992] for the US and a number of European countries.

¹²In contrast, most empirical studies find support for the neutrality of money in the *long-run*, see Weber [1994], King and Watson [1997], Serletis and Koustas [1998] and the references in Bullard [1999].

¹³By identifying exogenous shocks to monetary policy, one excludes systematic variations in monetary policy that may be endogenous reactions to current or expected future economic conditions. For an overview of the VAR approach to the analysis of monetary policy and the business cycle see Christiano et al. [1999].

¹⁴While Christiano et al. [1999] summarise studies for the US, Peersman and Smets [2003] find very similar characteristics for the Euro area.

Turning to the theoretical side, the goal of modern business cycle theory is to explain the observed pattern of co-movements of real and nominal variables in the business cycle.¹⁵

In doing so, Real Business Cycle (RBC) theory, which was the dominant approach to the study of economic fluctuations in the 1980s assigned a very limited role to money and other nominal influences since the theory sought to explain business cycles as resulting from the optimal reactions of firms and households to *real* shocks in a Walrasian setting with flexible prices. In fact, the bulk of the literature abstracted from money entirely.¹⁶

Despite this, RBC theory is important for the modern explanation of the short-term effects of money in that it is one of the central building blocks of the New Keynesian or New Neoclassical Synthesis (NNS) literature,¹⁷ the modern paradigm of business cycle analysis which in turn does attribute an important role to nominal variables in the explanation of cyclical movements. The New Neoclassical Synthesis literature owes its name to the fact that it integrates elements of Keynesian flavour into an optimisation-based Walrasian RBC model with rational expectations and infinitely-lived house-holds.¹⁸ These key New Keynesian elements are first, monopolistic competition in the goods or labour market. Second and most important, money matters in NNS models due to the introduction of nominal price rigidity as a source of non-neutrality.

Since the models to be presented in this thesis share the central building blocks of the New Neoclassical synthesis, these are discussed in some detail in

¹⁵Our presentation is focussed on the modern New Keynesian paradigm since it is beyond the scope of this introduction to give a complete account of the historical development of modern business cycle theory and money's role in it. For an exposition on this history see e.g. Mankiw [1990].

¹⁶If money was at all present in these models, in the absence of other nominal frictions the only source of non-neutrality was the "inflation tax on labour supply" which will be discussed in section 1.4.1.2. Simulations showed that money was not an important influence on the business cycle under these assumptions, see Cooley and Hansen [1989, 1995].

¹⁷The two terms are used synonymously in the literature.

¹⁸In the words of Goodfriend and King [1997] who first used the term, "Building on new classical macroeconomics and RBC analysis, it incorporates intertemporal optimization and rational expectations [...]. Building on New Keynesian economics, it incorporates imperfect competition and costly price adjustment [...]" (p. 255).

section 1.3.1.1. Since the Phillips curve relationship has long been considered one of the central trade-offs facing monetary policy makers, the idea is briefly discussed and the variant of the Phillips curve implied by the supply side of the NNS is introduced in section 1.3.1.2. Further, the main monetary transmission channel responsible for money's non-neutrality in the general equilibrium of NNS models is discussed in section 1.3.1.3.

1.3.1.1 The basic New Keynesian Business Cycle model

This section presents central elements of the standard New Keynesian or NNS business cycle: The assumption leading to a well-defined demand for money in equilibrium, the structure of the intermediate goods market with monopolistic competition and price rigidity, and the specification of monetary policy.

For the sake of better comparability with the models of the following chapters, the ensuing presentation of the New Keynesian model is set in continuous time while NNS models are usually presented in discrete time.

Money in the utility function Starting with the motive why money is held in equilibrium, a standard way to introduce money into the New Keynesian model is to assume that households derive utility from holding cash.¹⁹ As shown by Feenstra [1986], for the case of non-separable utility for consumption and real balances assumed in later chapters of this thesis, this can be viewed as a shortcut to modelling the transaction cost reducing services of money. In NNS models, a simpler separable form of preferences is usually assumed:

$$u(c,m,l) = \frac{c^{1-\sigma}}{1-\sigma} + a\frac{m^{1-b}}{1-b} - d\frac{(1-l)^{1+\eta}}{1+\eta}$$
(1.1)

¹⁹This representation is used e.g. by Chari, Kehoe and McGrattan [2000], Kim [2000], Erceg, Henderson and Levin [2000]. A frequently used alternative is the assumption that a subset of goods can only be paid for with cash ("Cash in advance", CIA) which will be discussed in section 1.4.1.2. It is used in NNS models by e.g. Yun [1996].

where c denotes consumption, m = M/P is real money balances held by the household, l is leisure l and σ , a, b, d, η are positive constants. The representative infinitely lived household chooses intertemporal paths for c, m and l to maximise the expected present value of utility subject to his budget constraint.²⁰ This results in the following conditions for the household's optimum at time τ :

$$d\frac{\left[1-l\left(\tau\right)\right]^{\eta}}{c\left(\tau\right)^{-\sigma}} = \frac{w\left(\tau\right)}{P\left(\tau\right)}$$
(1.2)

$$\frac{E_{\tau}\left\{c\left(\tau\right)\right\}}{c\left(\tau\right)} = \frac{r\left(\tau\right) - \rho}{\sigma}$$
(1.3)

$$a\frac{m(\tau)^{-b}}{c(\tau)^{-\sigma}} = i(\tau)$$
(1.4)

where E is the expectations operator. These equations constitute the demand side of the New Keynesian model. Equations (1.2) and (1.3) are two standard efficiency conditions implying that the marginal utility ratio between leisure and consumption is equalised to the real wage w/P and that the expected intertemporal allocation of consumption depends on the difference between the real interest rate r and the household's discount factor ρ . Most importantly, equation (1.4) states that the marginal utility ratio between real balances and consumption in equilibrium equals the nominal interest rate i which is the opportunity cost of holding money: By holding money instead of real assets, the household foregoes interest income r. Further, the real value of nominally denominated cash depreciates at the rate of deflation. Thus the difference in return between holding real assets and money is r minus the deflation rate, i.e. $r + \pi = i$ where π is the inflation rate. This implies that an increase in inflation ceteris paribus has an effect on the household's optimal ratio of consumption to real balances.

²⁰The only source of uncertainty are monetary policy shocks that will be discussed below. They determine the size of the seigniorage-financed lump-sum transfer the household receives.

Monopolistic Competition Households consume the economy's final good Y which is produced using a continuum of horizontally differentiated intermediate goods x_j according to the production function

$$Y(\tau) = \left[\int_{j=0}^{1} x_j(\tau)^{\frac{\alpha-1}{\alpha}} dj\right]^{\frac{\alpha}{\alpha-1}}$$

following Dixit and Stiglitz [1977]. The elasticity of substitution between any two types of intermediate good is $\alpha > 1$ implying that the types are imperfect substitutes in production, so that their producers act in an environment of monopolistic competition. This implies that the producer of any intermediate good type j can optimally choose a price subject to the demand function for his good

$$x_{j}(\tau) = \left[\frac{p_{j}(\tau)}{P(\tau)}\right]^{-\alpha} Y(\tau)$$

derived from the final good producers' cost minimisation, which is downward sloping in the good's relative price $\frac{p_j(\tau)}{P(\tau)}$.²¹ For the sake of simplicity, production of intermediate goods is assumed to be linear in labour.²² Optimal pricing then involves setting a positive mark-up over marginal cost, i.e. over the wage. Given positive marginal profits, firms are willing to increase supply at given prices in the face of an increase in nominal demand, creating the possibility for nominal shocks to affect output.²³

Nominal Price Rigidity The key Keynesian assumption allowing money to have real effects is the introduction of nominal price rigidity in the goods market: It is assumed that intermediate good producers can only change their prices infrequently. The standard specification of price rigidity used is due to Calvo [1983] and implies that firms can adjust their prices whenever they receive a reset signal. Since this reset signal follows a Poisson distribution, a firm's price is fixed for an interval of random length.

The main consequence of price rigidity is that instead of setting the mark-

²¹For details on the derivation of the demand function see, e.g. Walsh [2003].

 $^{^{22}{\}rm The}$ models abstract from capital.

²³At the same time, monopolistic competition implies that equilibrium is not Pareto optimal and that fluctuations are asymmetric in their welfare consequences.

up that maximises current profits, $\alpha/(\alpha - 1)$, firms, whenever they have the chance to readjust their price, choose a forward-looking mark-up that takes account of the expected future development of marginal cost and revenue while their price will be temporarily fixed. In particular, firms solve the problem

$$\max_{p_j} E_{\tau} \int_{s=\tau}^{\infty} e^{-\int_{t=\tau}^{s} [\chi(t)+\beta]dt} \left[p_j - w\left(s\right) \right] \left[\frac{p_j}{P\left(s\right)} \right]^{-\alpha} Y\left(s\right) ds$$

where the wage w(s) is the firm's marginal cost and where future profits are discounted with the factor $\chi(s)$. The weight given to profits of the future period s further decreases in the flow probability of receiving a price adjustment signal β since only those profits are relevant for the maximisation that accrue to the firm *before* its next price adjustment opportunity. The optimal price $p^*(\tau)$ is then a weighted average of the optimal prices that maximise expected profits in the future periods s where the price is fixed:

$$0 = E_{\tau} \int_{s=\tau}^{\infty} e^{-\int_{t=\tau}^{s} [\chi(t)+\beta]dt} \left[p^*\left(\tau\right) - \frac{\alpha}{\alpha-1} w\left(s\right) \right] \left[\frac{p^*\left(\tau\right)}{P\left(s\right)} \right]^{-\alpha} Y\left(s\right) ds$$

which can be solved for the forward-looking optimal mark-up under price rigidity

$$p^{*}(\tau) = \frac{\alpha}{\alpha - 1} \frac{E_{\tau} \int_{s=\tau}^{\infty} e^{-\int_{t=\tau}^{s} [\chi(t) + \beta] dt} \frac{w(s)}{w(\tau)} P(s)^{\alpha} Y(s) ds}{E_{\tau} \int_{s=\tau}^{\infty} e^{-\int_{t=\tau}^{s} [\chi(t) + \beta] dt} P(s)^{\alpha} Y(s) ds} w(\tau)$$
(1.5)

Given perfect competition in the final goods market, firms there make zero profits so that the output price $P(\tau) = \left[\int_{j=0}^{1} p_j(\tau)^{1-\alpha} dj\right]^{\frac{1}{1-\alpha}}$ is a weighted average of the intermediate good firms' prices. Calvo's specification of price rigidity is widely used because it facilitates aggregation of individual prices: Given that an intermediate good's effective price is the optimal price set when the firm last received a price adjustment signal and that the Poisson flow probability β of receiving a signal is independent of how long the price has been fixed, the output price can equivalently be expressed as a weighted average of past optimal prices, where the weights are determined by the Chapter 1. The real effects of money – an overview

Poisson distribution:

$$P(\tau) = \left[\int_{s=-\infty}^{\tau} \beta e^{-\beta(\tau-s)} p^*(s)^{1-\alpha} ds\right]^{\frac{1}{1-\alpha}}$$
(1.6)

Together, equations (1.5) and (1.6) form the supply side of the New Keynesian model.

Monetary policy The standard New Keynesian model can be reduced to a system of two equations: A demand side relationship derived from the household's Euler equation and a supply side relationship known as the New Keynesian Phillips curve. The model is then closed by the specification of a monetary policy rule. Besides its positive goal – understanding the business cycle and money's role in it - the NNS literature has also made important contributions to the normative debate about the optimal conduct and goals of monetary policy in the presence of nominal price rigidity.²⁴ Since most real world monetary authorities' policy is aimed at control of a short-term nominal rate rather than a monetary aggregate, most NNS models describe monetary policy as following an interest rate rule.²⁵ In NNS models not concerned with optimal policy design, a stochastic autoregressive process for money growth is instead chosen.²⁶ For our discussion of the monetary transmission mechanism in section 1.3.1.3, we make the latter assumption: $\frac{dM_t}{dt}\frac{1}{M_t} = \psi + \vartheta_t$ where the disturbance ϑ_t is independently and identically distributed.

In the context of the deterministic models of the subsequent chapters of this thesis, which all contain the simple deterministic relationship between money demand and the nominal interest rate described by equation (1.4), monetary policy can equivalently be described by an interest rate rule or a money growth rule.²⁷

 $^{^{24} \}rm See$ Woodford [2003], Walsh [2003], Clarida et al. [1999], Galí [2003], Erceg et al. [2000] and Goodfriend and King [1997].

²⁵See, e.g. Woolford [2003], Walsh [2003] and Clarida et al. [1999].

²⁶See e.g. Chari, Kehoe and McGrattan [2000].

²⁷See Ireland [2006], Clarida et al. [1999].

The models to be presented in this thesis are optimisation-based models with sticky prices set by intermediate goods firms acting in an environment of monopolistic competition and where households' utility depends positively on real money balances- i.e. the models share the central building blocks of the New Neoclassical synthesis we have just presented. However, some features of the methodological approach of the NNS literature sharply distinguish it from the approach of this thesis despite the mentioned similarities: The NNS literature is concerned with the explanation and replication of the fluctuations experienced by real economies in the short run. Concentration on the short-run implies firstly, that analysis focuses on models' behaviour off the steady state. In particular, the equations describing the economy's dynamic behaviour are linearly approximated around a zero-inflation steady state. Secondly, the analysis abstracts from secular growth, output is constant in the model's steady state. Thirdly, to be able to replicate real economies' fluctuations, both nominal and real shocks are introduced into the model which hit the economy in the short run.²⁸ The analysis usually does not involve explicitly solving the model but proceeds through the use of simulations. The reasons for this procedure are twofold: First, Lucas [1980] argued that the comparison of second moments generated by simulations of the model to those known from real world time series data is the preferable method for the empirical validation of a business cycle model.²⁹ Second, even if one preferred to proceed differently, New Keynesian Dynamic Stochastic General Equilibrium models are often too complex to lend themselves to non-numerical methods. The drawback to this methodological approach is that many "results" of the NNS literature are only shown with the help of simulations, meaning they only apply to a range of parameter calibrations considered plausible.

In contrast to the NNS literature, the focus of this thesis is on the effects of money on economies' long-run development in the absence of shocks at

 $^{^{28}\}mathrm{At}$ the same time, the model's equations are approximated around the model's deterministic steady state.

²⁹Although widely used, the approach is by no means universally accepted, see Hansen and Heckman [1996] for a critical discussion of calibrated simulations as a method of empirical validation.

the aggregate level. Results about the comparative static properties of our models' steady state equilibria are derived analytically; calibrated examples only serve to illustrate them. The issue of linearised dynamics in the neighbourhood of the steady state is only addressed in order to analyse the latter's stability properties.

Having described the New Keynesian model's building blocks, we now give a very brief description of the idea underlying the Phillips curve concept and show that the NNS business cycle model implies a forward-looking Phillipscurve relationship.

1.3.1.2 The Phillips curve and its New Keynesian variant

In an influential paper, Phillips [1958] described a strong negative relation between unemployment and the growth rate of nominal wages (wage inflation) for the UK.³⁰ By tying price movements to wage movements through firms' mark-up calculations, Samuelson and Solow [1960] reformulated the relationship in terms of inflation and output and emphasised the trade-off it involved for policy makers. The existence of a *long*-run trade-off was disputed by Friedman [1968] and Phelps [1967] who introduced the expectationsaugmented Phillips Curve: In deciding about their nominal wage claims, workers take into account the expected price level. Generating surprise inflation can therefore in the short-run lower real wages, increasing employment and output. Yet in the long run, Friedman and Phelps argued, there is no money illusion: workers' expectations about future prices adapt to the monetary authority's behaviour, making the long-run Phillips Curve vertical – any mean rate of inflation is then compatible with the long-run "Natural Rate" of unemployment and output.³¹ The long-run vertical Phillips curve or Natural Rate hypothesis has been a widely accepted tenet of Macroeconomics ever

³⁰It is beyond the scope of this introduction to give an extensive overview of the development of the Phillips curve discussion. Such an overview of the development (prior to the NNS literature) is given by McCallum [1990], King and Watson [1994] and Romer [2001]. King and Watson [1994] also include an overview of some empirical studies.

³¹Holding employment and output above their natural rates is only possible by allowing for ever accelerating inflation.

since. Yet the reader will in section 1.3.2 be introduced to a small strand of literature originating in the NNS framework that has recently disputed the vertical slope of the long-run Phillips Curve in a world with nominal price rigidity. Chapter 5 contributes to this small literature in showing that the relationships between money growth and both employment and output are hump-shaped in the Schumpeterian growth model with price rigidity.

In contrast, the existence of a short-run trade-off between (unexpected) inflation and output has never been seriously questioned. Recently, interest in the Phillips Curve has augmented with the surge of the NNS literature which for all widely used specifications of price rigidity³² implies a modern variant of the Phillips Curve reminiscent of the expectations-augmented formulation of Friedman and Phelps. We will now derive a continuous time version of this New Keynesian Phillips curve and briefly describe its properties.

Linearising the output price equation (1.6) around steady state equilibrium with zero inflation yields³³

$$\left[\pi\left(\tau\right) - \pi\right] = \beta\left[\widetilde{p^*}\left(\tau\right) - \widetilde{p^*}\right] \tag{1.7}$$

where $\pi(\tau) = P(\tau) / P(\tau)$, $\tilde{p}^*(\tau) = p^*(\tau) / P(\tau)$ and $x(\tau) - x$ is the current deviation of the variable $x(\tau)$ from its steady state value.

Linearising the optimal price (1.5) yields

$$(\beta + \chi) \left[\widetilde{p^*}(\tau) - \widetilde{p^*} \right] = E_\tau \left\{ \left[\widetilde{p^*}(\tau) - \widetilde{p^*} \right] \right\} + \frac{\alpha \left(\beta + \chi\right)}{\alpha - 1} \left[\widetilde{w}(\tau) - \widetilde{w} \right] + \left[\pi \left(\tau\right) - \pi \right]$$
(1.8)

where $\widetilde{w}(\tau) = w(\tau)/P(\tau)$ and where $\dot{x} = dx/dt$ is the time derivative of x. Finally using (1.7) to eliminate $\widetilde{p^*}(\tau)$ from equation (1.8), we have a continuous time version of the New Keynesian Phillips Curve (NKPC):

$$\chi \left[\pi \left(\tau \right) - \pi \right] = E_{\tau} \left[\pi \left(\tau \right)^{\cdot} - \pi \right] + \frac{\alpha \beta \left(\beta + \chi \right)}{\alpha - 1} \left[\widetilde{w} \left(\tau \right) - \widetilde{w} \right]$$
(1.9)

³²Roberts [1995] shows that NNS models based on the staggered contracts models of Taylor [1980] or Calvo [1983] or the quadratic price adjustment model of Rotemberg [1982] give rise to a common formulation of the Phillips Curve.

 $^{^{33}}$ Details on the linearisation of slightly more complicated versions of (1.5) and (1.6) can be found in chapter 3 which deals with the local stability of the steady state equilibrium.

Note first that in contrast to the traditional Phillips curve, the NKPC is forward-looking: Inflation today increases because future inflation is expected to increase, due to the forward-looking nature of optimal price setting under rigidity. Second, the driving force for the evolution of inflation are changes in marginal cost $\tilde{w}(\tau)$ since monopolistically competitive firms as seen above optimally set their price as a mark-up over marginal cost. In the presence of a positive correlation between marginal cost and output $[\tilde{w}(\tau) - \tilde{w}] =$ $g[Y(\tau) - Y]$ with $g'[Y(\tau) - Y] > 0$,³⁴ the NKPC can be recast in terms of an inflation-output relationship:

$$\chi \left[\pi \left(\tau\right) - \pi\right] = E_{\tau} \left[\pi \left(\tau\right) - \pi\right] + \frac{\alpha \left(\beta + \chi\right)}{\alpha - 1} \frac{\beta}{g} \left[Y \left(\tau\right) - Y\right] \tag{1.10}$$

Like traditional Phillips-curve specifications, equation (1.10) involves a short-run trade-off between inflation and output or equivalently, employment, since higher output is associated with higher employment in the model:³⁵ Inflation rates higher than their steady state level are for a given expected change in inflation associated with an output (employment) level above the steady state level since the coefficients on both variables are positive.

1.3.1.3 Transmission of monetary impulses: The interest rate channel

Putting together the Phillips curve and the demand side of the New Keynesian model, how are changes in the money supply transmitted to the real side of the economy in the NNS framework? The main transmission channel for monetary policy shocks in NNS models is the interest rate channel: An unexpected change in the nominal money supply or the nominal short-term

 $^{^{34}}$ A positive correlation can be established due to the fact that households are only willing to supply the labour needed for an increase in output under a higher wage. In the present context, see equation (1.2).

³⁵However, the focus has been on the New Keynesian Phillips Curve as a description of inflation dynamics. Empirical evaluations have focussed on this aspect, see e.g. Galí and Gertler [1999], Galí, Gertler, and López-Salido [2001], Sbordone [2002] and Eichenbaum and Fisher [2003] Fuhrer [1997] and Bils and Klenow [2004] present a more critical view. For an overview of the development of the theoretical and empirical discussion about the Phillips Curve see King and Watson [1994].

Chapter 1. The real effects of money – an overview

interest rate under price rigidity causes a change in the real short-term interest rates which in turn affects agents' optimal decisions.³⁶ The effects of a monetary policy shock are usually derived in NNS models with the help of simulations. Thus even fundamental insights about transmission channels may only be valid for certain parameter constellations considered plausible. Insofar the intuition that will be given here applies to the transmission of a serially uncorrelated negative money shock in a standard NNS model with "typical" calibration as presented e.g. by Walsh [2003] pp. 247ff.³⁷

The intuition can be illustrated with the help of equations (1.2) - (1.4)and (1.9). Given that the price level is predetermined in that only part of intermediate goods producers can change their prices at any given time, a decrease in the nominal money supply translates into a decrease of the real money supply m. By equation (1.4), for the household to be willing to reduce his real money demand, the nominal interest rate must rise.³⁸ Assuming that the rise in the nominal interest rate entails an increase in inflation that is less than proportionate, the real interest will rise.³⁹ Via an intertemporal substitution effect, this will lead the household to postpone consumption, i.e. current consumption falls relative to future consumption. For standard calibrations, the negative substitution effect on current consumption dominates the positive income effect of the interest rate increase so that current consumption decreases. By equation (1.2), this lowers demand for leisure (raises labour supply) at a given real wage. Since the fall in consumption entails a fall in equilibrium employment, the equilibrium real wage must fall. By equation (1.9), this decrease in marginal cost causes a decrease in the optimal price and in the inflation rate at given future inflation. Given that uncorrelated monetary policy shocks are examined, the money supply in the

 $^{^{36}}$ Regarding this transmission channel's empirical importance, Angeloni et al. [2003] find in a recent empirical survey of studies analysing monetary transmission channels in the Euro area that "while not dominant on the whole, the IRC [interest rate channel] is still a prominent channel in the transmission". (p. 25)

³⁷See also Galí [2003] and Christiano, Eichenbaum and Evans [2005].

 $^{^{38}}$ Galí [2003] discusses the transmission mechanism when money supply shocks are highly correlated: Here the nominal interest rate may fall (absence of "liquidity effect") but the real interest rate will still rise.

³⁹The effect on inflation is described below.

future will return to its original value, causing an increase in future inflation $(E_{\tau} [\pi(\tau) - \pi] > 0)$ that dampens the fall in current inflation.

Contrasting this intuition with the stylised facts about the effects of a monetary policy shocks cited at the beginning of this section shows that the baseline NNS model does quite well at capturing the non-neutrality of money at business cycle frequencies.

1.3.2 Inflation, employment and output in the long run: A non-vertical Phillips curve

As stated in section 1.3.1.2, it is the mainstream view in economics that inflation raises employment in the short-run but the Phillips curve is vertical in the long run. In the words of Taylor [1999b]:

"Theoretical and empirical research on the relationship between inflation and unemployment has left little doubt that there is no long-run trade-off between the rate of inflation and the rate of unemployment. In other words, the unemployment rate will average about the same amount, whether the inflation rate is zero percent or, say, ten percent." (p. 31)

Recently, however, a variety of models has been put forward which contain mechanisms allowing money to affect output and employment in the long run. Put differently, the models imply the existence of a non-vertical Phillips Curve. As stated in section 1.3.1.2, Friedman [1968] and Phelps [1967] argued that in the long run, the Phillips curve has to be vertical due to the absence of long-run money illusion. So to reintroduce a long-run influence of inflation on real variables, one has to either argue that money illusion does exist in the long run or put forward arguments why money might have an influence despite perfectly rational expectations. We begin by presenting one paper taking the first approach in section 1.3.2.1 and then in section 1.3.2.2 concentrate on the second idea which is closer to the focus of this thesis since it has been explored in the sticky-price framework of the NNS business cycle models presented in section 1.3.1. Further, some empirical evidence on the long-run inflation-unemployment relationship will be presented in section 1.3.2.3.

1.3.2.1 Non-vertical long-run Phillips curve: Money illusion

Turning to the first strand of the literature, in Akerlof, Dickens and Perry [2000], the flexibility of both prices and wages is unrestricted. The existence of a non-linear long-run Phillips curve in their model is due to agents being subject to a kind of money illusion even in the long run.⁴⁰ The authors argue that this departure from rationality will vanish not over time as one moves from the short run to the long run but moving from low inflation rates - where the costs associated with the deviation from perfect rationality are low – to high inflation rates where the costs increase. In particular, deviation from perfect rationality is modelled in the form of agents either ignoring inflation or giving inflation less than the adequate weight in the price / wage setting process. As a consequence, at low inflation rates where inflation is not a "salient" (p. 39) factor in decision making, price and wage setting will respond less than proportionately to expected inflation: Near-rational firms may not adjust nominal wages enough to keep real wages constant and near rational workers may not realise their real wage has fallen. Prices are a mark-up over unit labour cost, so that near rational firms' relative prices fall in inflation, raising demand and employment. As inflation rises, so does the cost of near-rational behaviour for agents. Threshold levels above which an agent switches to fully rational behaviour are normally distributed, leading to a non-linear form of the long-run Phillips curve with employment being a hump-shaped function of inflation. At high inflation, all agents behave rationally and employment returns to the Natural Rate.

1.3.2.2 Non-vertical long-run Phillips curve: Sticky prices

A handful of papers analyse the long-run relationship between money growth and the levels of employment and output by comparing the steady state

 $^{^{40}}$ The authors themselves do not use the term money illusion but it is suggested by Blinder [2000] in his discussion of the paper.

equilibria of their NNS models associated with different constant money growth rates.⁴¹ The pivotal assumption that generates the money nonsuperneutrality in these models is the existence of temporary stickiness in nominal wages or prices. As is apparent from equation (1.9), the mere presence of a positive discount rate χ used by economic agents is sufficient for the existence of a long-run Phillips-type trade-off in the standard NNS model.⁴²

The existence of the long-run inflation-unemployment trade-off is shown and its properties are analysed by Graham and Snower [2004] (henceforth: GS) in a model with flexible prices and rigid wages. In their model, households that are monopolistically competitive suppliers of differentiated labour types are faced with nominal wage rigidity. In particular, following Taylor [1980], each period the *nth* part of all households get to set a nominal wage for the next n periods. During the interval when a household's nominal wage is fixed, inflation steadily erodes its real wage. Taking this into account, a household that readjusts its wage chooses a nominal wage that will on average lead to an optimal real wage while the nominal wage is fixed. Positive discounting implies that the average of real wages formed by households is weighted, with weights that decrease over time. This implies that erosion of the real wage has not proceeded much in the periods that receive the highest weight. Consequently, the optimal nominal wage set by the household is not high enough to offset the effect of inflation on its mean wage during the n periods. As all households set wages this way, the aggregate real wage is lowered by inflation. The reduction in the real wage raises demand for labour, equilibrium employment and output. Thus by reducing the monopolistic distortion that households' optimal wage setting implies, the time discounting effect allows for a substantial employment-increasing effect of inflation – a

⁴¹See Ascari [1998, 2004], Devereux and Yetman [2002], Graham and Snower [2002, 2004], Karanassou, Sala and Snower [2003,2005] and King and Wolman [1998].

⁴²Defining the long-run by the fact that expected changes in inflation are zero, the short-run and long-run slopes of the Phillips curve in equations (1.9) or (1.10) would be equal. Note, however, that this is an artifact of the continuous time specification. The equivalent of equation (1.10) in discrete time is $\hat{\pi}(\tau) = \tilde{\chi} E_{\tau} \hat{\pi}(\tau+1) + \tilde{\alpha} \hat{Y}(\tau)$ where a hat over a variable denotes deviation from its steady state value. In the long-run characterised by $\hat{\pi}(\tau) = E_{\tau} \hat{\pi}(\tau+1) = \hat{\pi}$, we have $\hat{\pi} = \frac{\tilde{\alpha}}{1-\tilde{\chi}} \hat{Y}(\tau)$, implying that the Phillips curve is steeper in the long-run than in the short-run.

non-vertical long-run Phillips-Curve.⁴³ The simulations in Karanassou, Sala and Snower [2005] show that the trade-off may be substantial even at small discount rates when there is what Ball and Romer [1990] call "real rigidity" of wages, i.e. when the responsiveness of optimal wages to changes in real variables is small.⁴⁴

In contrast to the monotone inflation-employment relationship,⁴⁵ the longrun relationship between inflation and *output* is non-monotone in the model of GS: Given staggered wage setting under price rigidity, there is dispersion in real wages which distorts demand towards labour with a relatively low real wage, i.e. whose wage was set in the past and has since been eroded by inflation. Given that the labour types are imperfect substitutes in output production, this reduces their average productivity and hence, output. An increase in inflation thus ceteris paribus increases output via the increase in employment, but labour is used more and more inefficiently as inflation and real wage dispersion rise, ceteris paribus reducing output. The interaction of both effects makes output a hump-shaped function of inflation.⁴⁶

Interestingly, the authors' simulations also show that at all but extreme inflation rates, the qualitative relationship between inflation and real variables is unchanged when the contract length is endogenously chosen by

⁴⁴Karanassou, Sala and Snower [2003] argue that the sluggish adjustment of prices and wages may be complementary in generating the long run inflation-unemployment trade-off.

⁴⁶In a similar model with two period wage staggering à la Taylor [1980], Ascari [1998] engages in a number of numerical exercises on the output and welfare effects of a reduction in steady state money growth. His sensitivity analysis shows that the output and welfare gains increase strongly in the degree of non-linearity of the model, in particular the degree of increasing marginal disutility of labour (households' preference for smooth labour supply paths) and the non-linearities in production which make real wage dispersion inefficient (imperfect substitutability of intermediate goods, decreasing returns to labour-types).

 $^{^{43}}$ GS solve for the steady state numerically and explicitly derive results only for the steady state of the linearised economy in the simplest case n = 2.

⁴⁵GS also find the inflation-*employment* relationship to be hump-shaped when introducing a second transmission channel for inflation, the empirical relevance of which GS themselves consider "doubtful" (p. 16): Given real wage dispersion, the demand for a household's labour is distributed unevenly over the contract period, increasing as the wage is eroded by inflation. Assuming that households have a preference for smoothing out leisure over time, they demand a higher wage for a given average amount of labour to be supplied, the more unevenly this labour is distributed over time. Inflation that raises real wage dispersion thus raises the wage and reduces employment. Together with the time discounting effect, a hump-shaped relationship emerges.

agents.47

The paper of King and Wolman [1998] is a complement to Graham and Snower [2004] and the approach of this thesis in that it analyses the effects of positive steady state inflation in a NNS model with flexible wages where monopolistic producers using labour to produce intermediate goods are faced with *price* rigidity à la Taylor [1980] where each period half of all firms get to set prices for two periods. The authors do not explicitly derive the function relating output to inflation but argue that output is maximised at positive rather than zero inflation. This is due to the net effect of inflation via its influence on relative prices and on monopolistic mark-ups. For the model of chapter 5, we are able to analytically show that employment and output are hump-shaped functions of output in a broader framework where a more general specification of price rigidity is used.⁴⁸

Summing up, money under rigid prices or wages is non-superneutral due to inflation's effects on monopolistic distortions and on production efficiency. While the model presented in chapter 5 of this thesis shares this central idea with the papers presented here, there are several important differences that make the approach taken in this thesis stand out: First, the papers abstract from secular output growth. Hence, in contrast to this thesis they present no analysis of the interaction of the effects of inflation on employment and growth. Second, the literature mainly relies on simulation exercises instead of the analytical derivation of results whereas the approach of this thesis is analytical. Third, the bulk of the literature uses a particularly simple specification of price rigidity, whereas the more general specification used in this thesis allows for richer effects of money.⁴⁹

⁴⁷Graham and Snower endogenise price rigidity by allowing households to choose the length of their wage contract period assuming there is a fixed cost to changing wages. Their finding of a qualitatively unchanged relationship is confirmed by simulations of Devereux and Yetman [2002] of a model with Calvo [1983]-type price rigidity on the goods market.

⁴⁸Ascari [2004] builds a model with capital that like ours features Calvo [1983] price rigidity in the goods market. Relying exclusively on simulations, he shows that the steady state level of output depends negatively on the inflation rate.

⁴⁹In particular, the average mark-up is a non-linear function of the money growth rate under our specification of price rigidity. Among other things, this makes employment a hump-shaped function in our model, in contrast to the result of Graham and Snower [2004].

1.3.2.3 Non-vertical long-run Phillips curve: Empirical evidence

A number of studies investigate the slope of the long-run Phillips-curve for the US or Europe.⁵⁰ The evidence is very mixed: In many cases a vertical long-run Phillips curve cannot be rejected,⁵¹ while some evidence points to a long-run trade-off between inflation and unemployment. For example, in the study of Watson and King [1994], different identifying assumptions about the short-run behaviour of the variables lead to either a vertical long-run Phillips curve or a substantial trade-off for the post-war US.⁵² Koustas and Serletis [2003] find the same dependence of the long-run slope estimate on Keynesian versus Monetarist identifying assumptions using European data. Karanassou, Sala and Snower [2003] find support for a negative slope of the inflation-unemployment relationships using European data and give the following quantification: A "10 percent increase in long-run money growth (equal to long-run inflation) is associated with a 3.18 percentage point fall in the EU unemployment rate". At the same time, a limited number of recent studies report a positive slope of the long-run inflation-unemployment relationship: Bever and Farmer [2002] and Russell and Banerjee [2006] find evidence of a significantly positive long-run relationship between inflation and unemployment in the US. In the latter paper, the effect is quantified as implying that "an increase in inflation of around 5 percentage points [...] would be associated with an increase in unemployment in the long-run of about $1 \frac{1}{2}$ percentage points" (p. 14).

Summing up, the empirical relationship between inflation and unemployment in the long run is ambiguous, making it hard to reject any of the theories presented in section 1.3.2 or the approach of chapter 5.

 $^{^{50}\}mathrm{Relevant}$ recent studies are mostly based on two influential papers by Watson and King [1994, 1997].

⁵¹See e.g. Watson and King [1994, 1997] for the US and Weber [1994] on evidence for the G7 countries including Germany.

 $^{^{52} \}mathrm{Setterfield}$ and Leblond [2003] find support for a long-run trade-off in postwar US data.

1.4 Money growth, inflation and the growth rate of output

In this section, we first present some of the most prominent theoretical explanations of the non-superneutrality of money in the neoclassical and endogenous growth frameworks. The mechanisms allowing for monetary nonsuperneutrality in these literatures differ fundamentally from the assumption of nominal price rigidity made in this thesis and in the NNS literature. We nevertheless review the explanations in some detail in section 1.4.1 because they share the central question about the long-run *growth* effect of money which is one of the principal concerns of this thesis. Subsequently, we review some recent results of the empirical literature investigating the superneutrality of money with respect to output growth in section 1.4.2.

1.4.1 Survey of the theoretical literature

Two different literatures have contributed to the theoretical literature on inflation and growth. The first and older strand of the literature analyses the effects of money growth in the neoclassical growth model of Solow [1956] and Swan [1956] or Ramsey [1928], Cass [1965] and Koopmans [1965]. Technically, this literature might as well be included in section 1.3.2, since-given that steady state output growth is due to exogenous technical progress in the neoclassical framework- the influence of inflation on capital accumulation is at steady state restricted to effects on the *level* the capital-labour ratio. Yet due to the fact that this literature's focus is on the effects on capital accumulation which is also the focus of the modern literature on inflation and growth, the neoclassical monetary growth models are generally subsumed to the inflation-growth literature. Since this literature also paved the way for modern analyses of the inflation-growth nexus in models with an endogenously determined output growth rate which are the main focus of this section, section 1.4.1.1 starts with the presentation of the most influential contributions from this literature. Section 1.4.1.2 then turns to endogenous growth theory.

1.4.1.1 Long-run effects of money growth in neoclassical monetary growth models

While the steady state output growth rate in the neoclassical growth model is exogenously given, money does influence the *levels* of real variables in steady state.⁵³ In particular, the issue of money's superneutrality can be evaluated by comparing the steady-state levels of the capital-labour ratio and of output across steady states associated with different money growth rates. We sketch the main arguments of the three most influential articles from this literature.⁵⁴

Tobin [1965] in the neoclassical model with constant savings rate of Solow [1956] and Swan [1956] argued that the increase in inflation implied in the long run by an increase in money growth raises the opportunity cost for households of holding nominally denominated currency. As explained in section 1.3.1.1, the opportunity cost of holding money is the difference between the rates of return on capital and on money which in turn is given by the sum of the real interest rate and the inflation rate, i.e. the nominal interest rate. Assuming money and real capital are substitutes in the household's asset portfolio, an increase in the nominal interest rate induced by an increase in inflation then makes the household shift its portfolio towards higher capital holdings. Via this portfolio composition effect, an increase in money growth causes an increase in real capital investment that raises the steady-state levels of the capital-labour ratio and output. Although the *Tobin-effect* of inflation has received widespread attention, the model has been criticised for not making clear why households would choose to hold money in the first place, since money as a store of value is dominated by capital for any non-negative inflation rate.

To make explicit the preferences underlying the household's portfolio choice, Sidrauski [1967] integrated money into the Ramsey [1928]-Cass [1965]-Koopmans [1965] framework by assuming that households derive utility from

 $^{^{53}\,}Outside\,$ steady state, money growth for standard classes of utility functions does influence the growth rate of consumption and output, see e.g. Fischer [1979] for an analysis of the Sidrauski [1967] model.

⁵⁴See Orphanides and Solow [1990] for a thorough survey of the literature.

consumption and from holding real money balances. He showed that when households optimally choose their intertemporal consumption path, the steady state real capital stock in efficiency units and therefore, the level of output in steady state are determined independently of the money growth rate. Intuitively, money is superneutral since the marginal productivity of capital at steady state is unaffected by inflation.⁵⁵ Since an increase in inflation raises the optimal level of consumption relative to money holdings, the positive portfolio composition effect of inflation on the capital stock is exactly offset by a negative effect on total savings.

Finally, Stockman [1981] instead of introducing money into households' utility function modelled the role of money as a means of exchange by following Clower [1967] in assuming that households are subject to a Cash-inadvance constraint.⁵⁶ In his specification, this means that households must pay for their consumption good purchases and capital investments with cash instead of using credit services. In this framework, an increase in inflation that raises the opportunity cost of money holdings also raises the cost of investment in real capital. While the productivity of capital is unaffected, the return on investment is reduced. The "inflation tax" on investment therefore lowers the steady state capital stock in efficiency units and the output level in stark contrast to the result of Tobin [1965].

Summing up, different assumptions about the role of money in the economy led to vastly different results concerning the level effect of the money growth rate on the steady state capital intensity.

⁵⁵Sidrauski's specification of the utility function excluded leisure. Superneutrality continues to hold at steady state when leisure enters the utility function in such a way that the marginal rate of substitution between consumption and leisure is independent of money holdings. But as noted by Blanchard [1990] among others, even if little weight is given to money in the utility function, the superneutrality result is only an approximation since the change in the opportunity cost of money does influence real money holdings and therefore, utility.

⁵⁶The CIA-framework will be described in more detail in section 1.4.1.2.

1.4.1.2 Long-run growth effects of money growth in endogenous growth models

In the new generations of growth models,⁵⁷ the output growth rate of the economy at steady state is endogenously determined by economic agents' decisions and is thus subject to the influence of policy measures that change agents' incentives. The models to be reviewed below therefore compare the output growth rate across steady state equilibria associated with different inflation rates caused by policy makers' exogenous choice of a constant money growth rate. In the light of the recent empirical findings on the inflationgrowth nexus to be reviewed in section 1.4.2, the focus is on finding plausible channels through which inflation harms long-run growth. The literature's models share a similar structure made up by the following elements: A money demand function is generated by assuming either that households derive utility from holding money or that some purchases can only be made with cash (Cash in advance constraint).⁵⁸ If necessary, a further assumption is made to allow for an influence of inflation on real variables. In contrast to the NNS models discussed in section 1.3, the money supply process is modelled in the simplest possible way: The monetary authority expands the nominal money supply at a constant rate and distributes the proceeds lump-sum to households. Finally, the accumulation of physical and/or human capital which is not subject to diminishing returns is the mechanism driving growth in nearly all examined models.⁵⁹ The arguments inducing non-superneutrality of money in this framework can be divided into two categories: First, mechanisms that allow inflation to affect the marginal productivity of the capital form whose accumulation drives growth. Second, mechanisms that while leaving the marginal productivity of capital unaltered

⁵⁷For an introduction to endogenous growth theory, see Barro and Sala-i-Martin [2003] or Aghion and Howitt [1998].

⁵⁸Two less common alternatives are to introduce money as an input in production or to explicitly model a transaction cost reducing role for money.

⁵⁹The exception is the model by Marquis and Reffett [1994] who introduce an additional financial services sector (see section 1.4.1.2) into a Romer [1990]-type growth model with an increasing variety of intermediate goods to study the interaction of inflation, the economy's system of payments and growth.
influence the net real return of capital investment and thus, the incentive to accumulate capital.

Inflation affects the marginal productivity of capital A variety of arguments has been put forward why variations in the long-run inflation rate might influence the marginal productivity of either human or physical capital. In models where the accumulation of capital is the engine of growth, the resulting change in the incentive to invest automatically implies a growth effect of inflation. The two most important arguments in this context are the inflation tax on labour supply and the absorption of productive resources in the process of avoiding the cost of inflation.⁶⁰ Both arguments imply that inflation has an effect on the level of employment in the sector central to growth, and that this level effect in turn causes inflation to also affect the economy's growth rate. We briefly discuss each idea in turn.⁶¹

Inflation tax on labour supply Given standard preferences over consumption and labour, the introduction of a cash-in-advance (CIA) constraint on households' consumption purchases induces households to shift their consumption profile away from activities requiring cash (i.e., consumption) towards activities not subject to the cash-in-advance constraint (i.e., leisure). To see this, make the standard CIA assumption that in each period, consumption c takes place before households receive labour and rental income, so that money M for shopping has to be carried over from the previous period:

$$P_t c_t \le M_{t-1}$$

⁶⁰Alternatively, real money balances can be introduced as an input in the production process. Inflation then increases the cost of the input, so firms economise on their money holdings. In a two sector growth model with constant returns to physical and human capital in output production, this allows inflation to influence growth if money is essential in the production of the limiting factor human capital but only has a level effect if human capital production is unaffected by money, see Wang and Yip [1992], Pecorino [1995] and Chang [2002].

⁶¹Gillman and Kejak [2005a] present a detailed comparison of a number of models integrating these two effects in a variety of different endogenous growth models.

As argued above, since money in contrast to capital does not bear any real interest r and further, the real value of money depreciates at the inflation rate π , the opportunity cost of transferring one unit of goods across time in the form of money instead of capital is given by the nominal interest rate $R = r + \pi$. Then the first order condition governing the static optimisation concerning leisure l and consumption c from the maximisation of utility U(c, l) subject to a standard budget constraint is

$$\frac{U_l}{U_c} = \frac{w}{1+R}$$

where U_x is marginal utility from consumption of x and w is the real wage.⁶² The term 1 + R is the effective cost of consumption under the CIA-constraint given by the goods price plus the opportunity cost of holding the money necessary for the transaction. As the effective cost of consumption rises in the inflation rate π , an increase in π decreases the optimal ratio U_l/U_c , which for standard preferences implies that leisure is increased relative to consumption.⁶³

The "inflation tax on labour supply" then results in a negative effect of inflation on growth in any model where the following three conditions $hold:^{64,65}$ Capital accumulation results in sustained growth, the marginal

 $^{^{62}}$ For details see, e.g. Walsh [2003].

⁶³Although we here follow the literature in introducing money via a CIA-constraint, the substitution from consumption to leisure induced by the rising opportunity cost of holding money with a utility can also be modelled in the money in the utility framework of section 1.3.1: A more general specification of utility than (1.1) that would allow the ratio U_l/U_c in the optimality condition (1.2) to depend negatively on money holdings at given c, l, would imply that demand for leisure increases at a given real wage when inflation rises. Using the CIA constraint makes it possible to avoid introducing such a more complex utility function.

⁶⁴See De Gregorio [1992,1993], Jones and Manuelli [1995], Dotsey and Ireland [1996], Gillman, Harris and Mátyás [2004] and Gillman and Kejak [2005b]. Gomme [1993] also features the inflation tax on labour supply but focuses on the short run inflation-growth relationship off steady state.

⁶⁵Note that in the model of Stockman [1981], there was a negative level effect of inflation through intertemporal substitution of consumption because the CIA constraint extended to purchases of consumption *and* investment goods. In the models of the present section, CIA on consumption alone is sufficient to generate a negative growth effect of inflation via the substitution of consumption and leisure given that labour supply is endogenous. The "inflation tax on investment"-setting of Stockman is extended to an endogenous growth

productivity of capital increases in equilibrium employment (i.e., labour has a scale effect on growth) and a decrease in labour supply results in a decrease in equilibrium employment.⁶⁶

This is most straightforward in the setup of Jones and Manuelli [1995], where long-run growth of per capita production y is possible since returns to physical capital k and effective labour nh are constant. The output production function is $y = k^{\alpha} (hn)^{1-\alpha}$ where $0 < \alpha < 1$ and n is employment. Output can be costlessly transformed into either physical capital or human capital h for accumulation purposes. At any given steady state capital intensity k/h,⁶⁷ accumulated human capital is used less intensely when employment decreases, lowering the marginal products of both types of capital⁶⁸ and thus lowering the incentive to invest and the steady state growth rate. Therefore, economic growth is maximised when monetary policy authorities follow the Friedman rule:⁶⁹ Contracting the money supply at the rate that results in a zero nominal interest rate as prescribed by the Friedman rule eliminates the opportunity cost of holding money and thus does away with households' substitution towards leisure.

Existence of a resource-absorbing credit sector In a number of papers, households can endogenously determine the fraction of goods they want to pay for with cash while the rest is paid for with credit the production of which requires input of productive resources.⁷⁰ In Dotsey and Ireland [1996], for example, credit is produced using labour. An increase in inflation

model e.g. by Marquis and Reffett [1995].

⁶⁶The latter condition does not generally hold in the models examined by Fukuda [1996] and Itaya and Mino [2003] where under strongly increasing returns to scale in production due to labour externalities, a decrease in labour supply may under certain transaction technologies and parameter constellations cause an increase in equilibrium employment and hence, a positive growth effect of inflation.

 $^{^{67}}$ To simplify matters, assume that the depreciation rates of h and k are identical. Then the steady state value k/h is determined by the technology parameter α independently of inflation.

 $^{{}^{68}\}frac{\partial y}{\partial k} = \alpha n^{1-\alpha} \left(\frac{k}{h}\right)^{\alpha-1}, \ \frac{\partial y}{\partial h} = (1-\alpha) n^{1-\alpha} \left(\frac{k}{h}\right)^{\alpha}.$ ${}^{69}\text{See Friedman [1969]}.$

⁷⁰See Marquis and Reffett [1994], Dotsey and Ireland [1996], Gillman, Harris and Mátyás [2004] and Gillman and Kejak [2005b]. See Temple [2000] for references on the interaction of inflation, the financial sector and growth.

raises the cost of using money relative to credit and thus raises demand for credit. The resulting reallocation of labour to the financial sector reduces the resources available for real production activities. Therefore as in the previous section, in any model of growth through capital accumulation where the marginal productivity of capital depends on employment (or more generally, on the amount used in production of the resource that also produces credit), an increase in inflation entails a lower output growth rate due to the reallocation of resources to the credit sector. Again, the steady state output growth rate is maximised when monetary authorities follow the Friedman rule.

Influence of inflation on the return to investment in spite of unchanged marginal productivity of capital The second group of arguments explains why inflation might affect the return on investment, and hence, capital accumulation and the growth rate, even when the marginal productivity of capital is independent of inflation. E.g., the growth effect of inflation via nominal rigidity in the tax system is analysed by Chari, Jones and Manuelli [1996] and Jones and Manuelli [1995]. They investigate the consequences of a non-indexed tax system with nominal depreciation allowances. By raising the nominal interest rate, inflation reduces the present value of depreciation allowances, raising the effective tax rate and reducing the after tax return on investment. Similarly, they show the spread between borrowing and lending rates of banks caused by cash reserve requirements on bank deposits increases in inflation. Assuming that part of capital investment has to be financed with bank loans, rising inflation thus again reduces the return on investment.⁷¹

The approach to monetary analysis taken in this thesis shares with this literature the way money demand generated by money in the utility function and money supply are modelled, and the methodology of comparative statics regarding deterministic steady state equilibria. What distinguishes our approach are both the modelling of endogenous growth as being due to

⁷¹In both the non-indexed tax system approach and the model with reserve requirements for banks, setting the growth rate of money supply according to the Friedman rule again maximises the rate of economic growth.

stochastic research and development activities instead of capital accumulation and the introduction of nominal price rigidity as a friction that allows money to have real effects. As our literature review has shown, both elements have thus far been neglected in the study of inflation and endogenous growth.

1.4.2 Empirical evidence on money growth and output growth

The effect of inflation or money growth on the real output growth rate has been investigated using a variety of methods and approaches.⁷² While some early studies using mostly cross-country data found no significant correlation between inflation and growth,⁷³ some cross-sectional studies report a significantly negative effect of inflation on growth. E.g., Motley [1998] reports that "the coefficients [...] imply that in long-run steady state, a ten percent inflation rate will reduce annual per capita growth in an average country by about 1/4 percentage point" compared to a situation with zero inflation (p. 22).⁷⁴ The results of cross-country studies have however been criticised for their lack of robustness with regard to changes in the country samples, time period and regression specification.⁷⁵ Research has therefore turned to the use of panel data sets hoping to get more robust results by exploiting the information contained in the data's time series dimension.⁷⁶ Using 5-year averages, Barro [1996] finds a linear negative effect of inflation similar to that reported by Motley [1998]. In reaction to the critique that this finding may

⁷²Temple [2000] contains a survey of recent empirical contributions as well as a discussion of the methodological difficulties involved. Summaries or overviews of empirical investigations are also given in Gillman and Kejak [2005a], Gylfason and Herbertsson [2001], Ghosh and Phillips [1998], Bruno and Easterly [1998] and Ragan [1998].

⁷³The cross-country study of McCandless and Weber [1995] is a good example and contains further references. More recently, Judson and Orphanides [1999] find no significant relation in cross-country data but a negative relation when panel data are used. In a time–series setup, Geweke [1986] finds support for the superneutrality hypothesis using a century of annual U.S. data.

 $^{^{74}{\}rm The}$ results of Fischer [1993] are of a similar magnitude.

⁷⁵See Levine and Renelt [1992], Levine and Zervos [1993] and Clark [1997].

⁷⁶Cf. e.g. to Barro [1996], Judson and Orphanides [1999], Gylfason and Herbertsson [2001] and Gillman, Harris and Mátyás [2004].

be mainly due to the effect of high-inflation countries, most recent studies try to both allow for non-linearity of the inflation-growth relationship and differentiate between industrialised and developing countries. While a negative relationship has been confirmed for most inflation rates, non-linearity seems indeed to be present in the data along two dimensions: Firstly, the negative inflation-growth relationship seems to be convex: there is evidence of decreasing marginal cost of inflation at high inflation levels.⁷⁷ Secondly, using spline techniques, several studies have allowed the coefficient on inflation to differ for observations above or below certain thresholds. There is indeed some evidence that below a certain threshold value, the influence of inflation on growth may be insignificant or even positive. E.g., Khan and Senhadji [2001] report that the inflation-growth relationship is weakly but significantly positive for industrialised countries below an inflation threshold of 1%. The threshold value found for industrialised countries in other studies are somewhat bigger.^{78,79}

While the investigation of the non-linearity of the inflation-growth relationship seems promising, a general caveat concerning all mentioned studies is that the long-run growth rate of output is of course not observable in the data: Evidence from panel studies using 15-, 10-, 5-year averages or even annual data is used to make claims about the nature of the long-run inflationgrowth relationship. While the main message- at least medium to large values of inflation are detrimental to growth- is mostly preserved when moving from high to lower frequency data, the levels of significance are sometimes lower.⁸⁰ So additional efforts need to be made to further disentangle the medium-run and long-run efforts of inflation on growth.

Despite these difficulties encountered in the empirical investigation of the inflation-growth relationship, the results can be summed up as follows:

⁷⁷See, e.g. Gylfason and Herbertsson [2001] and Gillman, Harris and Mátyás [2004] or Ghosh and Phillips [1998].

⁷⁸Ghosh and Phillips [1998] report a threshold value of 2.5%, Sarel [1996] finds a threshold at 8% and Gylfason and Herbertsson [2001] reported values of 10% and a rather high 20% for two different data sets.

⁷⁹The threshold for developing countries seems to be somewhat higher than for industrialised countries, see e.g. Khan and Senhadji [2001].

⁸⁰See e.g. Ghosh and Phillips [1998].

There is considerable evidence that money is non-superneutral with respect to the growth rate of real GDP. The relationship seems to be non-linear: The influence of inflation on output growth is negative at large inflation rates, while being possibly insignificant or positive at low levels of inflation.

1.5 The long-run influence of money on innovationdriven growth and the level of employment under nominal price rigidity

1.5.1 Attempting a long-run synthesis

As the reader might have gathered from the preceding sections, the framework of analysis to be used in this thesis will be a synthesis of the approaches presented in this introduction: With the New Keynesian literature we share the conviction that price rigidity is a realistic friction that is central to the transmission of nominal impulses to the real side of the economy.

With the endogenous growth literature, we share the conviction that the output growth rate can be directly influenced by policy and that in particular, the impact of monetary policy is not restricted to the short run. Therefore close in spirit to the New Neoclassical Synthesis literature that introduced Keynesian frictions into the Standard Business Cycle model we attempt to build a long-run synthesis by introducing nominal price rigidity into one of the workhorse models of endogenous growth theory, the Schumpeterian quality ladder model of Grossman and Helpman [1991] and Aghion and Howitt [1992].

1.5.1.1 Price rigidity in a long-run model?

At first sight, the reader might be surprised by our conjecture that nominal price rigidity – which is usually assumed to be relevant at business cycle frequencies – should matter for an economy's long-run growth performance – given the elements' different time dimension, their joint analysis might seem disproportionate. To see that this is not the case and why we should on the

contrary expect short-term price rigidity to indeed have an impact on longterm growth, consider the following points: First, although prices are fixed for only short periods of time, price rigidity is a permanent feature of the economy. In its presence, in the short run relative prices are constantly distorted at any non-zero inflation rate. Since relative prices steer the short-run allocation of real sources in the economy, the latter is consistently influenced by price rigidity. Thus, permanent price rigidity does influence the level of real economic variables in the short and long run. Accepting this level effect inevitably leads one to expect that price rigidity allows for an influence of inflation on the growth rate of output, too: Long-term growth is but repeated short-run growth originating from the investment decisions of economic agents. These decisions are in turn based on current and expected prices and levels of real variables. Thus there is no reason to believe that a feature of the economy that influences short-run levels would not influence long-run economic growth as well. Therefore, the inflation-growth literature's negligence of price rigidity as a transmission channel for monetary impulses on output growth is an unjustified gap that the present thesis aims to fill.

1.5.1.2 Why innovation-driven growth?

In the framework used in this thesis, economic growth is fuelled by stochastic research and development activities that lead to innovations. We now give a short description of this framework and then argue why it is particularly suitable for our analysis.⁸¹

There are three productive sectors in the economy: The economy's final good is produced in a perfectly competitive sector using a large number of differentiated intermediate goods that are imperfect substitutes. Each intermediate good is of a certain quality level (position on the "quality ladder") which determines its productivity. Since intermediate goods are produced one for one with labour, increases in the quantity of intermediates are limited. However, long-run output growth is possible due to increases in the quality of intermediate goods brought about by innovations of the Research

 $^{^{81}{\}rm For}$ a more detailed presentation see Aghion and Howitt [1998] or Barro and Sala-i-Martin [2003] which is closest to our exposition.

and Development (R&D) sector. Every intermediate good type is produced by one firm which holds the exclusive right to produce this type thanks to the acquisition of the relevant patent from the R&D sector. Firms in the intermediate goods sector earn positive quasi-profits since they act in an environment of monopolistic competition due to the fact that intermediate good types are imperfect substitutes in production. The prospect of positive profits leads to buyers' competition in the market for patents. The price of a patent is then determined by the present value of profits to be made from sales of the corresponding innovative intermediate good, which replaces its predecessor of inferior quality in the market for intermediate goods. The patent price in turn determines the incentive to engage in research activities. There is free entry to the R&D sector, where an increase in the input of real resources raises the flow probability of making an innovation. As the incentive to innovate increases, so does the amount of resources devoted to research and hence, the economy's output growth rate.

This framework is suitable for our purposes for various reasons: First, as explained above growth is driven by research activities leading to the design of innovative goods – the market entry of new intermediate goods with corresponding prices is a permanent feature of the economy. Thus, pricing decisions - and therefore, price rigidity - are intimately and naturally connected to the incentive to innovate and the very heart of the economy's growth process. Second, price setting under rigidity is easily integrated into the framework and its results are well comparable to the NNS models of sections 1.3.1 and 1.3.2 since the structure of monopolistic firms selling differentiated intermediate goods to a competitive final good sector is identical. Third, it is not the aim of this thesis to analyse the inflationary problems in developing countries – where governments' financing constraints and underdevelopment of the financial sector might be more central to the transmission of monetary impulses. But in industrialised countries which are thus the focus of our attention, the generation of new ideas and goods is probably more important for growth than the accumulation of capital that the inflation-growth literature has concentrated on so far. For all of these reasons, the quality ladder model is an adequate framework for the analysis of the long-run consequences

of money growth.

1.5.1.3 Empirical evidence on price rigidity

The results of a model based on nominal price rigidity can only be considered substantial if this transmission channel for monetary policy is empirically relevant, i.e., if there is evidence that quantities do change faster than prices. Summing up the relevant empirical literature, this is indeed the case: It is the consistent finding of all major studies investigating price setting behaviour that producer and consumer prices are fixed for periods that are not negligible both in the Euro area and the US.⁸² Disagreement only arises with respect to first, the average duration of the period for which prices are fixed and second the determinants of price rigidity, i.e. the motives firms have for not changing their prices immediately in response to shocks.⁸³ Regarding the duration of price rigidity, Taylor [1999a] in a survey of the literature reports an average length of one year for the US. According to the influential study of Blinder et al. [1998], prices in the US are changed on average once in 9 months, while Bils and Klenow [2004] concentrating on consumption goods found a median length of less than 6 months.⁸⁴ According to a recent survey, price rigidity seems to be somewhat more severe in the Euro area where the average price duration is reported to be "close to one year" by Álvarez et al. [2006].⁸⁵ In order not to overstate the effects of price rigidity, the parameter choices for the calibration of examples in the subsequent chapters are in line with the smallest mentioned estimates – and indeed we find that adding a small

⁸²See e. g. the survey of Taylor [1999a].

⁸³It is beyond the scope of this exposition to discuss the underlying causes of price rigidity. Surveys of the theoretical explanations for price rigidity and their empirical relevance are presented by Wolman [2000], and by Álvarez et al. [2006] for the Euro area. Blinder et al. [1998] conducted an influential survey in the US.

⁸⁴The median length was 4.3 months or 5.5 months excluding temporary price cuts, i.e. sales. Bils and Klenow note that the difference in results between their study and the one of Blinder et al. [1998] might be due to the fact that price rigidity might be more important for intermediate goods rather than consumer goods producers since "firms in the Blinder et al. [1998] survey sell mostly intermediate goods and services (...) rather than consumer items" (pp. 953-954).

⁸⁵Summarising recent studies, they report an average (median) duration of 13.0 (10.3) months from micro data on consumer prices and 10.8 months from firm surveys.

amount of price rigidity is sufficient to generate a quantitatively significant degree of non-superneutrality of money in our Schumpeterian model.

A related question is how price rigidity should be modelled – is the price setting process of firms best described by a time dependent or state dependent schedule, i.e. do firms change prices when a certain amount of time has passed or based exclusively on the state of demand and other economic variables. The results reported by Álvarez et al. [2006] for the Euro area indicate that a time dependent schedule is a reasonable approximation: 34% percent of firms use pure time-dependent price reviewing rules, while among the rest, "firms that mainly follow time dependent rules, but change prices in the case of specific events dominate" (p. 581).⁸⁶ We therefore use the time-dependent pricing rule of Calvo [1983] that as discussed in section 1.3 is standard in the modern New Keynesian business cycle analysis.

1.5.2 Short outline of the remaining chapters

The remainder of this thesis is dedicated to the analysis of the long-run effects of money growth in the described Schumpeterian growth model with price rigidity. To concentrate fully on the effects of price rigidity, money is introduced into the model in a way consistent with the superneutrality of money under flexible prices. In particular, we follow the approach of Sidrauski [1967] traced out in section 1.4.1 in assuming that households derive utility from the consumption of goods and from holding real balances.

The survey of the inflation-growth literature in section 1.4.1 has shown that effects of employment are often important in the analysis of the inflationgrowth nexus. In **chapter 2**, we abstract from this channel entirely and show that increases in money growth and inflation are detrimental to growth in the presence of price rigidity even at constant labour supply and employment. A brief introduction is given in section 2.1. The model is then presented in section 2.2, whereas section 2.3 shows existence and uniqueness of the steady state rational expectations equilibrium. The analysis of the steady

⁸⁶Without referring to a specific study, the authors also report that "the share of firms following time-dependent rules is 40%" for the US (Álvarez et al. [2006], p. 581).

state's comparative static properties in section 2.4 shows that money growth at the steady state equilibrium has three effects on innovation driven growth under price rigidity: First, the initial relative price of an intermediate good under price rigidity increases in the money growth rate. This is due to money growth's effect on the real wage and on the optimal mark-up chosen by the good's producer: As seen in section 1.3.1.1, since firms anticipate their inability to offset the erosion of their relative price through inflation under price rigidity, they optimally set an initial mark-up that increases in the growth rate of marginal cost. Since the latter is in equilibrium identical to the money growth rate, the optimal mark-up and the initial relative price increase in the money growth rate. Demand for an intermediate good and hence, the firm's profits, decrease in the firm's relative price. Since profits from sales of an intermediate good determine the market value of a patent, inflation ceteris paribus reduces the patent price and hence, the incentive to innovate and economic growth. In contrast, erosion of the firm's relative price through inflation raises demand and profits while the nominal price is fixed. Via this second effect, an increase in money growth ceteris paribus raises growth. Third, as seen in section 1.3.2, inflation also has an effect on the efficiency with which resources whose relative prices are distorted are used. In particular, demand is distorted towards intermediate goods whose nominal prices have been fixed for a long time, implying their relative prices are low. Given the constant elasticity of substitution between intermediate goods in final good production, this reduces their average productivity and hence, profits. Via this channel, any deviation of money growth from zero reduces growth because it raises the dispersion of intermediate goods prices. Taking together the three influences, the economic growth rate decreases monotonically in the money growth rate. The model thus shares the policy recommendation given by the growth models reviewed in section 1.4.1.2 that from a growth perspective, the monetary policy authority should follow the Friedman rule, i.e. contract money supply at a rate that at the limit makes the nominal interest rate zero.⁸⁷ The comparative static results concerning

⁸⁷This is the lower bound on admissible money growth rates in those models and ours: Equation (1.2) shows that equilibrium is only well-defined for positive interest rates.

the effects of money growth and of price rigidity are illustrated with a calibrated example where the quantitative effect of money growth is compatible with empirical estimates reported in section 1.4.2 of the present chapter.

To check whether the properties of the steady state that our analysis focuses on are indeed relevant as a description of the economy's long-run behaviour, chapter 3 investigates whether the economy converges to steady state in the long run. To this end, the linearised economy's behaviour in the neighbourhood of the steady state is examined. As explained in the introductory section to the chapter (section 3.1), analysing the steady state's local stability turns out to be a challenging task due to the dynamic system's high dimensionality. We therefore restrict the analysis to the steady state's local stability properties in a large number of calibrated examples. The key equation determining the economy's behaviour off steady state are described in section 3.2. The following section 3.3 presents the reduced dynamical system and the results of the local stability analysis: Without any exception, the steady state is found to be locally stable, which justifies the concentration of our long-run analysis on steady state outcomes. Interestingly, the steady state equilibrium under price rigidity at the same time turns out to be indeterminate, i.e. the convergent solution is not unique.

In the previous steps of the analysis, it was assumed that parameters are such that remaining in the market is unprofitable for the incumbent intermediate good firm if a firm entering the market with an improved version of the same type sets the monopoly price. Relaxing this assumption, chapter 4 analyses the equilibrium under limit pricing: Innovative firms set a limit price to force the incumbent out of the market. After a brief introduction in section 4.1, section 4.2 presents the model. It is shown that there is a unique steady state equilibrium where money growth has an additional negative effect on intermediate firms' profits and the incentive to innovate: Innovators cannot choose a forward-looking initial price to optimally offset the erosion of their intermediate goods' relative price through inflation because they set the limit price that makes staying in the market immediately unprofitable for the incumbent firm. This further raises the negative impact of an increase in money growth on economic growth and turns out to be of importance quantitatively in a calibrated example. Thus, money growth is more detrimental to output growth when innovations are non-drastic.

In chapter 5, we relax the assumption that employment is independent of money growth and inflation. Instead of introducing a consumption-leisure choice in the household's utility maximisation problem, we directly introduce an aggregate labour supply function which increases in the level of the real wage in efficiency units. Introducing labour supply in this simple way helps to keep the model tractable and allows us to speak about involuntary unemployment. A short introduction is given in section 5.1. In section 5.2, the model is presented with a slight modification in the sectoral structure: Labour is used together with intermediate goods in the production of the economy's final good, whereas the input used in the intermediate goods sector and in the R&D sector is output. Steady state equilibrium is discussed in section 5.3 and its existence and uniqueness are shown in section 5.4. Section 5.5 presents comparative static results on the interaction of money growth, employment and output growth in steady state equilibrium that are illustrated with the help of a calibrated example. While it turns out that the growth rate has no significant effect on employment, an increase in employment does in turn raise the rate of economic growth.

Employment is influenced by money growth and inflation via two channels because the productivity of labour increases in the total amount of intermediate goods used in production and in the efficiency of their use. The total amount of intermediates used is influenced by the money growth rate since the latter affects the average mark-up charged by intermediate good producers, i.e. the degree of monopolistic distortion in the economy. Efficiency of the usage of intermediate goods decreases in the degree of relative price distortion which in turn increases in the absolute value of the inflation rate. We show that the interaction of the two effects gives rise to a non-linear long-run Phillips curve: Given the research intensity, we prove that the effects make employment a hump-shaped function of money growth peaking at a money growth rate associated with a positive inflation rate. Given that employment is approximately invariant to the growth rate, the result carries over to general equilibrium. The same applies to the hump-shaped form of the output level as a function of the money growth rate.

Given employment, output growth is a hump-shaped function of the money growth rate peaking at zero inflation.⁸⁸ In general equilibrium, employment's positive effect on the research intensity and the fact that employment increases in money growth at small inflation rates bring about an increase in the growth maximising money growth rate: Output growth is maximised at a positive inflation rate.

A monetary authority interested in stimulating employment or economic growth should therefore fight high inflation which reduces both variables, whereas tolerating a very moderate amount of inflation promotes both its goals.

Finally, chapter 6 presents some concluding remarks.

⁸⁸The difference concerning the growth effects of deflation between the results of chapter 2 and 5 is due to the different influence of the degree of monopolistic distortions. Section 5.5.5 elaborates on this.

Chapter 2

Money growth and Innovation-driven growth

2.1 Introduction¹

In this chapter, the Schumpeterian growth model with nominal price rigidity is presented and the unique steady state equilibrium's comparative static properties are discussed. In particular, the issue of money's (non-) superneutrality with respect to the output growth rate and the latter's dependence on the degree of price rigidity are examined.

The real side of the model is the quality ladder model following Aghion and Howitt [1992] and Grossman and Helpman [1991] which was briefly introduced in the last section of chapter 1. In this framework, sustained output growth is possible due to the growing quality (i.e., productivity) of a large number of differentiated intermediate goods that are imperfect substitutes in the production of the economy's final good. Quality improvements are the result of the efforts of researchers in many small firms in the research and development sector where research success occurs in a random manner governed by a Poisson process. In case of a success, the research firm sells the patent for its invention to a firm that then holds the exclusive right to produce the innovative good. The incumbent firm producing the old inferior

¹Parts of this chapter are based on Funk and Kromen [2005].

version of the intermediate good type is forced out of the market by the innovator's entry. Intermediate goods are produced one for one with labour. Due to the fact that intermediate goods are imperfect substitutes, producers act in a monopolistically competitive environment and optimally set their price as a markup over marginal cost, i.e. the wage. Under perfectly flexible prices, this profit maximising mark-up is constant across time and depends only on the degree of monopoly power.

The survey of the literature on endogenous growth and inflation in section 1.4.1.2 of chapter 1 showed that variations in employment may be an important channel for the transmission of inflation's effects on growth. In this chapter, we abstract from this channel to focus on the direct effect of price rigidity and relative price changes on the return to $R\&D.^2$

The nominal side of the model corresponds to that of a standard New Keynesian model as presented in section 1.3 of chapter 1, with a simplified specification of the money supply rule. Money is introduced into the model by assuming that households derive utility from holding real money balances, following Sidrauski [1967]. Money matters in the model because we assume the existence of nominal price rigidity in the intermediate goods market: When entering the intermediate goods market with a new good, the innovator chooses a price. Once in the market, firms can only change their prices infrequently, where price rigidity follows the standard structure of Calvo [1983] and Kimball [1995]³. Following the literature on inflation and endogenous growth presented in section 1.4.1.2 of chapter 1, it is assumed that the money supply grows at an exogenously fixed constant rate so that there are no shocks at the aggregate level. We analyse Rational Expectations Equilibria in this model, where we restrict our attention to steady state equilibria with constant output growth.

To get an intuition for the non-superneutrality of money in this frame-

 $^{^{2}}$ Labour supply will be endogenised and the interaction of money growth, employment and growth will be examined in chapter 5 of this thesis.

³In Kimball's variation of Calvo's model, firms are assumed to set prices so as to maximise the present value of profits, instead of following a rule of thumb as modelled by Calvo.

work, note first that the growth rate of intermediate good firms' nominal cost, i.e., the wage inflation rate, depends positively on (in fact, is equal to) the growth rate of money supply in this model. Therefore, an intermediate goods producer would like to constantly change his price – but can only do so infrequently under price rigidity – at any non-zero money growth rate. The changes in the firm's effective mark-up and relative price brought about by non-zero money growth influence the profits which can be made selling an innovative intermediate good. These level effects of money growth lead to an influence on the output growth rate of the economy given that the present value of profits from sales of an intermediate good. Since the patent price determines the incentive to innovate, investment in R&D and economic growth are influenced by the money growth rate.

In particular, we identify three distinct effects of non-zero money growth and inflation⁴ on the steady state research intensity.

First, non-zero money growth under price rigidity implies dispersion in the effective prices of intermediate goods. The optimal price grows at the constant money growth rate while each firm's effective price is the price that was optimal at the stochastic point in time the firm was last allowed to readjust its price. Both absolute and relative prices of intermediate goods differ, distorting demand towards goods with low relative prices. Given the constant elasticity of substitution between all intermediate good types in final good production, the distortion of demand reduces the average productivity of intermediate goods. This reduces aggregate demand for intermediate goods. Given positive marginal profits, the reduction in demand for the innovative intermediate good entails lower profits for the innovator. Via this *price dispersion effect*, an increase in the absolute value of the money growth rate that raises price dispersion lowers the return to R&D.

Second, money growth influences the incentive to innovate via the *rel*ative price erosion effect: Given that the intermediate firm's price is tem-

⁴We discuss the effects of an increase in the exogenous money growth rate which in equilibrium entails an increase in the endogenous inflation rate.

porarily fixed under price rigidity whereas its marginal cost grows at the money growth rate, the firm's mark-up is eroded by inflation. Mark-up erosion lowers the firm's relative price, thereby raising demand for its good and therefore raising profits. The extent of mark-up erosion increases in the money growth rate ψ , so that an increase in ψ ceteris paribus raises profits, the value of a patent and hence, the research intensity.

Third, the incentive to innovate is influenced by money growth via the *initial relative price effect*: The intermediate goods producer's initial relative price depends on the real wage level and on the firm's choice of an initial mark-up given the real wage. Firms anticipate that their nominal price will be temporarily fixed and that consequently, their mark-ups and relative prices will be eroded by inflation. To offset this, every firm optimally chooses a forward-looking mark-up that increases in the money growth rate whenever it gets the chance to readjust its price. At the same time, the real wage is also influenced by money growth. Taking account of both effects, under inflation and price rigidity the high initial mark-up translates into a high relative price that reduces demand for the intermediate producer's good and hence, his profits. The resulting decrease in the return to R&D caused by an increase in the money growth.

It is shown that the negative effects of an increase in money growth on the incentive to innovate always dominate so that the output growth rate decreases in the money growth rate ψ at all admissible values of ψ . A realistically calibrated example illustrates the results and shows that adding an empirically plausible degree of nominal price rigidity to our Schumpeterian growth model is sufficient for the generation of a significant negative effect of inflation on growth which is quantitatively in line with estimates from the empirical literature. So indeed the effects of money growth and price rigidity are not restricted to the short run. Instead, the negative level effects on production efficiency and relative prices that inflation has under price rigidity translate into long-run growth effects via their influence on the incentive to innovate. The remainder of the chapter is organised as follows: Section 2.2 describes the model. Section 2.3 presents the steady state equilibrium. The comparative static properties of the steady state equilibrium are discussed in section 2.4. Section 2.5 presents a calibrated example. Section 2.6 concludes.

2.2 The model

Intermediate goods are the only input in the production of the final good, while labour is used in the production of intermediate goods and in research. We first derive the final good sector's optimal demand for intermediate goods and the intermediate goods sector's optimal prices and resulting labour demand in a monopolistically competitive environment under price rigidity. This labour demand function depends on intermediate firms' current prices, which in the presence of price rigidity depend on the past distribution of price resetting signals. Analogously, the following calculation of the market value of a new intermediate goods producer at the time of his market entry takes account of the influence of the stochastic timing of future price resetting opportunities on his relative price and profits. The resulting market value is the price for a patent from the research sector. Free entry implies expected profit from research must be zero. Using the information about the patent price, the zero profit condition gives us the optimal research intensity μ as a function of the size of the final good sector. Using optimal labour demands, we then determine the size of the final good sector compatible with labour market equilibrium. With this information, production side equilibrium gives us the optimal research intensity as a function of L and other variables firms take as given. Analysis of the public sector that controls the money supply, of the asset market and of the optimal behaviour of households, which hold utility-yielding real balances, yields the remaining conditions necessary to discuss the general equilibrium.

2.2.1 Final good sector

A perfectly competitive final good sector assembles the economy's final good $Y(\tau)$ from a large number N of differentiated intermediate goods according to a constant-elasticity-of-substitution aggregator following Dixit and Stiglitz [1977]

$$Y(\tau) = \left[\sum_{j=1}^{N} \left(q^{k_j(\tau)} x_{k_j}(\tau)\right)^{\frac{\alpha-1}{\alpha}}\right]^{\frac{\alpha}{\alpha-1}}$$
(2.1)

where we assume that the elasticity of substitution between intermediate goods α is larger than unity and where x_{k_j} is the amount of sector j type k_j intermediate good used, and q^{k_j} is this type's productivity. We assume that only the highest quality k_j currently available in sector j is used in production.⁵ Profits in the sector are given by

$$\Pi^{Y}(\tau) = P(\tau)Y(\tau) - \sum_{j=1}^{N} p_{k_j}(\tau)x_{k_j}(\tau)$$
(2.2)

where $P(\tau)$ is the final good price and $P_{k_j}(\tau)$ is the price charged for one unit of type k_j sector j intermediate good. Cost minimization leads to firms' optimal demand for intermediate good x_{k_j} , which depends negatively on the type's relative price and positively on its productivity $q^{k_j(\alpha-1)}$ and on aggregate demand $Y(\tau)$.

$$x_{k_j}(\tau) = \left(\frac{p_{k_j}(\tau)}{P(\tau)}\right)^{-\alpha} q^{k_j(\alpha-1)} Y(\tau)$$
(2.3)

Constant returns to scale and perfect competition prevent firms from making positive profits. Optimal demand for intermediate goods and the zero profit condition determine the final good price as a quality-weighted

⁵This can be achieved by assuming that entry of the innovator terminates the incumbent's production. Alternatively and less drastically, the incumbent will voluntarily exit the market if production is not profitable for him once the innovator enters the market. See footnote 8 for a discussion of the conditions which ensure that this holds when the innovator sets the monopoly price.

average of intermediate goods' prices:

$$P(\tau) = \left[\sum_{j=1}^{N} \left(\frac{p_{k_j}(\tau)}{q^{k_j(\tau)}}\right)^{1-\alpha}\right]^{\frac{1}{1-\alpha}}$$
(2.4)

For later use, we define the economy's aggregate technology index, $Q(\tau)$, as the weighted sum of the productivities $q^{k_j(\tau)}$ associated with each sector's intermediate good

$$Q(\tau) = \left[\sum_{j=1}^{N} q^{(\alpha-1)k_j(\tau)}\right]^{\frac{1}{\alpha-1}}$$
(2.5)

2.2.2 Intermediate goods sector

Each of N intermediate goods is produced by one firm that bought the blueprint from the corresponding research firm. Good k_j is produced using a linear technology with labour as the only input:

$$x_{k_i}(\tau) = L_{k_i}(\tau) \tag{2.6}$$

2.2.2.1 An intermediate good producer's pricing problem

The instantaneous profit of a firm producing type k_j consists of the difference between its price P_{k_j} and marginal costs, given by the wage rate $w(\tau)$ determined in the perfectly competitive labour market, times the number of units sold.

$$\Pi_{k_j}(\tau) = \left\lfloor p_{k_j}(\tau) - w(\tau) \right\rfloor x_{k_j}(\tau)$$

As intermediate goods are imperfect substitutes in the final good production function, intermediate goods producers act in an environment of monopolistic competition and can choose an optimal price subject to the final good sector's demand function.

We assume the existence of nominal rigidities in the intermediate goods markets: Producers can only adjust their prices infrequently. Following the standard assumption in the literature due to Calvo [1983] and Kimball [1995], firms in the market at each point in time with an exogenous flow probability β receive a signal allowing them to readjust their prices, where the signal is generated by a stochastic Poisson process with parameter β . In addition, in our model a firm naturally chooses a price for its new product when entering the intermediate goods market with an innovative good. As the arrival of inventions in the research sector is governed by a Poisson process with endogenously determined parameter μ , this does not change the basic structure of price resetting signals received by firms as modelled in the literature while partly endogenising the parameter in a natural way.

Knowing they will not be able to readjust their prices for some time, firms thus solve an intertemporal problem to find the optimal price: At the time of its market entry $\tau = t_{k_j}$ as well as at each later date τ where it receives a price resetting signal, firm j sets it price such as to maximise the expected present value $E[V(p_j, \tau)]$ of future profits. Only those future profits are relevant for the maximisation that are generated before the firm's next opportunity to readjust its price arises and before the next innovative good enters the market in sector j, which we assume drives the incumbent out of the market.

Therefore, profits of all future periods s are weighted with the probability that as of time s, the firm has not received a signal allowing to readjust prices and has not been replaced by a successor. This leads to a discount factor given by the sum of the nominal interest rate, i, the flow probability of being replaced by another firm, $\mu_{k_j}(\tau)$ and the flow probability of receiving a reset signal for the price, β .⁶ In steady state equilibrium, the replacement probability will be constant and the same for all firms, so we set $\mu_{k_j}(\tau) = \mu(\tau) = \mu$. The resulting value is

$$E\left[V\left(p_{j},\tau\right)\right] = \int_{\tau}^{\infty} \widetilde{B}e^{-(i+\mu+\beta)(s-\tau)}\left[p_{j}-w(s)\right] \left(\frac{p_{j}}{P(s)}\right)^{-\alpha} q^{k_{j}(\alpha-1)}Y(s)ds$$
(2.7)

 \tilde{B} is a constant from the integration of the probability distribution of the price reset signal which will drop out during maximisation.⁷ Maximising the

⁶While the parameter β is constant by definition, the interest rate *i* is constant at steady state.

⁷The discount factor at time τ for profits from future period s contains the product of the cumulative probability densities $B(\tau, s) = \tilde{B}e^{-\beta(s-\tau)}$ and $D(\tau, s) = \tilde{D}e^{-\mu(s-\tau)}$ for

value given in (2.7) with respect to the price p_j subject to the final good producing firm's demand function (2.3) and using that at steady state, the final good $Y(\tau)$, price level $P(\tau)$ and wage $w(\tau)$ grow at rates γ , π and ω , respectively, leads to the following expression for the optimal price at time τ :⁸

$$p_j^*(\tau) = \frac{\alpha}{\alpha - 1} \frac{i + \mu + \beta - \alpha \pi - \gamma}{i + \mu + \beta - \alpha \pi - \gamma - \omega} w(\tau)$$
(2.8)

Note that $p_j^*(\tau)$ is the optimal price for firm j whenever it can set a new price, whether this is at market entry or at a later date when it receives a price resetting signal.⁹ Note also that the optimal price is independent of a firm's position on the quality ladder, so we will rename it $p^*(\tau)$. The optimal price is a mark-up over marginal cost $w(\tau)$. In a world without rigidities $(\beta \to \infty)$, a monopolistically competitive firm would optimally charge a price $p_{flex}^*(\tau) = \frac{\alpha}{\alpha-1}w(\tau)$, where the mark-up $\frac{\alpha}{\alpha-1}$ reflects the degree of monopoly

⁹The maximisation problem has a well-defined solution for $i + \mu + \beta - \alpha \pi - \gamma > 0$, $i + \mu + \beta - \alpha \pi - \gamma - \omega > 0$. Assumption (2.38) and $\eta \ge 1$ from the household's problem guarantee that these inequalities hold in equilibrium.

the probabilities that between τ and s, no price resetting is received and the firm is not replaced by the next innovation, respectively. The firm that sets a price at market entry at time τ knows that $D(\tau, \tau) = 1$ so $\tilde{D} = 1$ given that the probability of two innovations occurring in an infinitesimal time interval is negligible. However, since the timing of the last price resetting signal is unknown, so is \tilde{B} . The situation is reversed when the firm later resets its price due to a price resetting signal.

⁸Knowledge of the optimal monopolistic mark-up allows us to discuss the conditions under which monopoly pricing will prevail if the incumbent is not forced to terminate production upon the entry of an innovator: Taking the innovation as a price resetting signal for the incumbent firm, too, only the latest quality is available in each sector if the innovator's mark-up is such that production is not profitable for the incumbent. Given that his good is one quality rung below the innovator's, the incumbent can at most charge $\frac{1}{q}p_j^*(\tau)$. Given the steady state optimal monopolistic mark-up (2.8), $\frac{1}{q}p_j^*(\tau) - w(\tau) \leq 0$ holds when the sufficient conditions $1 \geq \frac{1}{q} \frac{\alpha}{\alpha-1} \frac{\rho+\beta+(1-\alpha)\psi}{\rho+\beta-\alpha\psi}$ and $1 \geq \frac{1}{q} \frac{\alpha}{\alpha-1}$ are jointly satisfied. While the second condition, which is identical to the condition from the underlying real model, ensures that the incumbent makes non-positive profits under negative money growth rates, the first condition must hold for $\psi = \omega > 0$ where firms set an additional forward-looking mark-up $\frac{i+\mu+\beta-\alpha\pi-\gamma-\omega}{i+\mu+\beta-\alpha\pi-\gamma-\omega} = \frac{\rho+\beta+(1-\alpha)\psi+[(\alpha-2+\eta)\frac{q\alpha-1}{\alpha-1}+1]\mu}{\rho+\beta-\alpha\psi+[(\alpha-2+\eta)\frac{q\alpha-1}{\alpha-1}+1]\mu} > 1$. Since this term decreases in μ for $\psi > 0$, $1 \geq \frac{1}{q} \frac{\alpha}{\alpha-1} \frac{\rho+\beta+(1-\alpha)\psi}{\rho+\beta-\alpha\psi}$ is a sufficient condition. The conditions are satisfied at all examined money growth rates in our calibrated examples in

this term decreases in μ for $\psi > 0$, $1 \ge \frac{1}{q} \frac{\alpha}{\alpha-1} \frac{p+p+(1-\alpha)\psi}{p+\beta-\alpha\psi}$ is a sufficient condition. The conditions are satisfied at all examined money growth rates in our calibrated examples in section 2.5. If production continued to be profitable for the incumbent under monopoly pricing, the innovator would set the limit price to drive the incumbent out of the market. The equilibrium under limit pricing is examined in chapter 4.

power and decreases in the elasticity of substitution between intermediate goods, α .

Under price rigidity, positive growth of marginal costs ($\omega > 0$) will erode the firm's relative price in future periods while its price is fixed. Anticipating this, the firm chooses an initial mark-up that is higher than the flexible optimum $p_{flex}^*(\tau)$ under these circumstances.¹⁰ Given that at steady state the wage inflation rate ω equals the growth rate of money supply, ψ ,¹¹ we therefore have that the initial mark-up increases with money growth, allowing it to influence real activity.

The size of the additional mark-up further depends on the importance of future relative to current profits in the firm's optimisation at time τ . Increases in factors that reduce the weight of future profits reduce the markup, drawing it closer to the static optimum $\alpha/(\alpha - 1)$.¹² In contrast, an increase in variables that accelerates future profit growth raises the initial mark-up, reducing the deviation from optimal price in future periods.¹³

Equilibrium in the market for intermediate goods Supply in the market for intermediate goods equals demand as production of each good j is determined by the final good sector's demand function for good j given in equation (2.3).

2.2.2.2 An intermediate good producer's market value at market entry

The value of a patent developed in the R&D-sector will be equal to the market value $E(V_{k_i}(\tau) | t_{k_i} = \tau)$ at the time of market entry t_{k_i} of the firm

¹⁰In New Keynesian business cycle models without real growth, the optimal price at steady state would be $p_j^*(\tau) = \frac{\alpha}{\alpha-1} \frac{i+\beta-\alpha\psi}{i+\beta-(\alpha+1)\psi} w(\tau)$.

¹¹This will be shown in section 2.3.1.1.

¹²These variables are the nominal interest rate i, the rate of obsolescence μ and the probability of receiving a price reset signal β . Remember that only profits obtained while this price has not been replaced by a newer one count in the choice of today's optimal price. This is why the price resetting probability β reduces the weight of future profits.

¹³Future demand and profits increase in the future growth rate of aggregate demand, γ , and inflation π . Remember from equation (2.3) that demand for the intermediate good is ceteris paribus a function $[p_j(s)/P(s)]^{-\alpha}$ of its relative price, which under price rigidity erodes at rate $\alpha\pi$.

using the patent. This market value is the expected present value at time τ of all future profits of the type k_j , given that $t_{k_j} = \tau$. Taking into account stationary growth of Y, P and w, the probability of obsolescence before time $s, e^{-\mu(s-\tau)}$, and the development of the firm's price, this value is:

$$E\left(V_{k_{j}}\left(\tau\right)\left|t_{k_{j}}=\tau\right.\right) = A\left(\tau\right)E\int_{\tau}^{\infty}\left[e^{-\chi\left(s-\tau\right)}p_{j}\left(s\right)^{1-\alpha} - w\left(\tau\right)e^{-(\chi-\omega)\left(s-\tau\right)}p_{j}\left(s\right)^{-\alpha}\right]ds$$

$$(2.9)$$

where $A(\tau) = q^{k_j(\alpha-1)}Y(\tau)P(\tau)^{\alpha}$ and $\chi = i + \mu - \alpha\pi - \gamma$. Note that unlike in the optimal pricing problem leading to equation 2.8, here all future profits enter the maximisation. Thus the calculation involves expectations not only about the life-span of the firm, but also about its price at any future date. This makes the analysis considerably more complicated than in the model without money. In particular, in any future period $s > \tau$ where it is still active, the firm's price is still $p^*(\tau)$ if no price reset signal has been received between periods τ and s, which has probability $\left(1 - \int_{\tau}^{s} \beta e^{-\beta(s-\theta)} d\theta\right)$. Otherwise, $p_j(s)$ is equal to $p^*(\theta)$ where θ is the last period where a reset signal was received. θ can take any value between τ and s, weighted with the corresponding probability $\beta e^{-\beta(s-\theta)}$. Using this and evaluating the integrals, the firm's market value is¹⁴

$$E\left(V_{k_{j}}\left(\tau\right)\left|t_{k_{j}}=\tau\right.\right) = \frac{q^{k_{j}\left(\alpha-1\right)}Y\left(\tau\right)\left(\frac{p^{*}\left(\tau\right)}{P\left(\tau\right)}\right)^{-\alpha}\frac{w\left(\tau\right)}{\alpha-1}}{\left[i+\mu-\alpha\pi-\gamma-\left(1-\alpha\right)\omega\right]\frac{i+\mu+\beta-\alpha\pi-\gamma-\omega}{i+\mu+\beta-\alpha\pi-\gamma-\left(1-\alpha\right)\omega}}$$
(2.10)

Like the expected market value of a firm with totally flexible prices, $q^{k_j(\alpha-1)}Y(\tau) \left(\frac{p^*(\tau)}{P(\tau)}\right)^{-\alpha} \frac{w(\tau)}{\alpha-1}/[i + \mu - \alpha\pi - \gamma - (1 - \alpha)\omega]$, the firm's value under rigidity can be interpreted as the present value of an infinite stream of profits growing at a constant rate.¹⁵ The flex-price firm's instantaneous profit is $q^{k_j(\alpha-1)}Y(\tau) \left(\frac{p^*(\tau)}{P(\tau)}\right)^{-\alpha} \frac{w(\tau)}{\alpha-1}$, while its discount rate is the difference

¹⁴Derivation of the market value is described in more detail in Appendix 2.A.1.

¹⁵Remember that the present value at time τ of a an infinite stream of steadily growing profits starting at τ is $\Pi(\tau)/(R-x)$, where $\Pi(\tau)$ is instantaneous profit at τ , x is the profit growth rate and R is the interest rate.

of the obsolescence-adjusted interest rate $i + \mu$ and its profit growth rate $(\alpha \pi + \gamma + (1 - \alpha)\omega)$. The flex-price firm's profit growth rate is determined by the growth rates of aggregate demand γ , of its price and wages ω , and of demand for the good at rate $\alpha (\pi - \omega)$ caused by the change of its relative price at rate $(\omega - \pi)$.¹⁶

The market value of a firm under price rigidity differs from this value in two respects: First, as explained in the discussion of equation 2.8, in anticipation of the effects of price rigidity, the firm's initial price $p^*(\tau)^{-\alpha}$ contains an additional mark-up that increases in ω (ψ), reducing demand and instantaneous profits. Second, due to the infrequent price adjustment, the growth rate of profits and hence the discount rate is different from the flex-price case. The flex-price discount rate is therefore corrected with the factor $(i + \mu + \beta - \alpha \pi - \gamma - \omega) / (i + \mu + \beta - \alpha \pi - \gamma - (1 - \alpha) \omega)$

which accounts for the difference in profit growth rates between periods where prices are fixed versus flexible.¹⁷ This influence of rigidity on profit growth is one of the mechanisms that allow money growth and inflation to influence the incentive to innovate in general equilibrium.¹⁸

Note that in contrast, the effect of inflation and relative price erosion on the firm's profit *per unit* is completely offset by the firm's optimal choice of its initial mark-up: From Appendix 2.A.1, the average size of profits per unit sold is $\frac{p^*(\tau)}{i+\mu+\beta-\alpha\pi-\gamma} - \frac{w(\tau)}{i+\mu+\beta-\alpha\pi-\gamma-\psi}$, which is of course also influenced by inflation and money growth. The denominators reflect the different growth rates of revenues $(\alpha \pi + \gamma)$ and costs $(\alpha \pi + \gamma + \psi)$. Yet this difference in growth rates is taken account of in the endogenous choice of optimal price by the firm: The initial mark-up $\frac{\alpha}{\alpha-1} \frac{i+\mu+\beta-\alpha\pi-\gamma}{i+\mu+\beta-\alpha\pi-\gamma-\psi}$ over wages is chosen such that the present value of revenues is identical to what it would have been if revenues had grown at the same constant rate as costs.

 $^{^{16}}$ See equation (2.3) for the determinants of demand for the firm's good.

¹⁷Note that both discount rates featured in the correction factor increase in the frequency of the price adjustment signal β , such that as $\beta \to \infty$, the discount rate reduces to the flex-price rate.

¹⁸A detailed intuition is given in section 2.4 where we discuss the equilibrium's comparative static properties.

2.2.2.3 The Intermediate Goods sector's labour demand

We next derive the sector's labour demand which will later be used to determine the size of the final good sector compatible with labour market equilibrium. Due to the linear production function (2.6), firm j's labour demand is equal to the final goods sector's demand for the firm's good. Inserting the optimal price into the demand function (2.3) and aggregating leads to intermediate good producers' labour demand

$$L^{X}(\tau) = \frac{Y(\tau)}{Q(\tau)} \sum_{j=1}^{N} \left(\frac{p_{j}(\tau)}{P(\tau)Q(\tau)}\right)^{-\alpha} \left(\frac{q^{k_{j}}}{Q(\tau)}\right)^{\alpha-1}$$
(2.11)

Obviously, aggregate demand for intermediate goods depends negatively on the average relative price of intermediate goods (in efficiency units). This average price effective at time τ can be expressed as a weighted average of past optimal prices set by firms at the last (stochastic) point in time *s* where they could readjust their prices. The weights $f(s, \tau)$ thus refer to the probability that a price valid at time τ has not been changed since time *s*:

$$\sum_{j=1}^{N} p_j(\tau)^{-\alpha} = \int_{-\infty}^{\tau} f(s,\tau) \left[p^*(s) \right]^{-\alpha} ds$$

More specifically, the weights reflect the probability that a price reset signal was received or an innovation made at time s, $(\mu + \beta)$ and that no such event took place between times s and τ , $e^{(\mu+\beta)(s-\tau)}$. Thus, we have $f(s,\tau) = (\mu + \beta) e^{(\mu+\beta)(s-\tau)}$. Using this and evaluating the integral, we have¹⁹

$$L^{X}(\tau) = \frac{Y(\tau)}{Q(\tau)} \left[\frac{p^{*}(\tau)}{P(\tau)Q(\tau)} \left(\frac{\mu + \beta}{\mu + \beta - \alpha\omega} \right)^{-\frac{1}{\alpha}} \right]^{-\alpha}$$

 $^{^{19}}$ A more detailed derivation of equation (2.12) can be found in Appendix 2.A.2. Note that convergence of the integral is ensured by condition (2.38), see footnote 40.

Inserting the general equilibrium value of the current the optimal price from equation (2.15) gives

$$L^{X}(\tau) = \frac{Y(\tau)}{Q(\tau)} \left[\left(\frac{\mu + \beta}{\mu + \beta - (\alpha - 1)\omega} \right)^{\frac{1}{\alpha - 1}} \left(\frac{\mu + \beta}{\mu + \beta - \alpha\omega} \right)^{-\frac{1}{\alpha}} \right]^{-\alpha}$$
(2.12)

Given the constant elasticity of substitution between intermediate goods in final good production, the quality-weighted amounts of all intermediate goods used should be equal. Only in this case is production efficient and the amount of intermediate goods used corresponds exactly to the amount of output in efficiency units to be produced, $L^X(\tau) = Y(\tau)/Q(\tau)$. The term $\left[\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\omega}\right)^{\frac{1}{\alpha-1}}\left(\frac{\mu+\beta}{\mu+\beta-\alpha\omega}\right)^{-\frac{1}{\alpha}}\right]^{-\alpha} \geq 1$ is therefore a measure of static production inefficiency caused by *price dispersion*: The fact that intermediate goods have different effective relative prices under price rigidity causes them to be demanded and used in different amounts. Since this causes inefficient production, the higher price dispersion, the more intermediate goods $L^X(\tau)$ are needed to produce a given level of output in efficiency units $Y(\tau)/Q(\tau)$. This static production inefficiency is completely unrelated to the inefficiency caused by monopolistic producers' positive mark-ups since it is the dispersion in mark-ups and relative prices, not the absolute level of the mark-up, that causes the inefficiency.²⁰

Going into more detail, note that price dispersion is caused by the concurrence of two elements: Price rigidity and non-zero growth of marginal cost. In the absence of price rigidity, all firms charge the same price, so there is no price dispersion. Under rigidity, each intermediate producer effectively

²⁰Regarding the components of the price dispersion measure, remember that as seen above the term $\left[\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\omega}\right)^{\frac{1}{\alpha-1}}\left(\frac{\mu+\beta}{\mu+\beta-\alpha\omega}\right)^{-\frac{1}{\alpha}}\right]^{-\alpha}$ is the average relative price charged by intermediate goods producers: While $\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\omega}\right)^{\frac{1}{\alpha-1}}$ is the current optimal relative price, $\left(\frac{\mu+\beta}{\mu+\beta-\alpha\omega}\right)^{-\frac{1}{\alpha}}$ is a measure of the deviation of the average relative price from the optimal value. If prices could be reset each period, all firms would charge the same optimal price and both terms would equal unity. The average relative price would then equal the common optimal relative price and therefore be unity. Under price rigidity, firms' actual prices differ and hence the average relative price in equation (2.12) differs from one. charges the price that was optimal at the time when he could last readjust prices. Given that optimal prices are a mark-up over marginal cost, past optimal prices only differ from the current one when marginal cost has a non-zero growth rate. Since the growth rate of marginal cost ω equals the money growth rate ψ as will be shown in equation (2.34) in section 2.3.1.1, money growth influences price dispersion. In particular, price dispersion and production inefficiency increase in the distance of the money growth rate ψ from zero because the larger the absolute value of the money growth rate, the faster is the change in optimal prices and the bigger the difference in prices between goods with old prices and to goods with new prices.²¹ It is also intuitive price that dispersion decreases in μ and β because the higher these variables, the more often prices are adjusted.²²

2.2.3 Price level and real wage

Given that firms in the final good sector make zero profit in equilibrium, the output price (2.4) is a quantity-weighted average of the prices of the intermediate goods that are the only input in production. Analogously to the procedure in section 2.2.2.3, this average intermediate good price can be rewritten as a weighted average of past optimal prices, since every effective intermediate good price is a past optimal price dating from the time the firm could last readjust its price. The weights are again given by $f(s,\tau) =$ $(\mu + \beta) e^{(\mu+\beta)(s-\tau)}$ from section 2.2.2.3. Using these weights and the growth rate ω of optimal prices the weighted average of past optimal prices can be

rewritten as^{23}

$$P(\tau) = \left(\frac{p^*(\tau)}{Q(\tau)}\right) \left[\frac{\mu + \beta}{\mu + \beta - (\alpha - 1)\omega}\right]^{\frac{1}{1 - \alpha}}$$
(2.13)

Equivalently,

$$1 = \left(\frac{\frac{\alpha}{\alpha-1}\frac{i+\mu+\beta-\alpha\pi-\gamma}{i+\mu+\beta-\alpha\pi-\gamma-\omega}w(\tau)}{P(\tau)Q(\tau)}\right) \left[\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\omega}\right]^{\frac{1}{1-\alpha}}$$
(2.14)

The last equation shows that the average real intermediate good price must equal one. It determines the real wage in efficiency units as one over the average intermediate goods mark-up $\frac{\alpha}{\alpha-1} \frac{i+\mu+\beta-\alpha\pi-\gamma}{i+\mu+\beta-\alpha\pi-\gamma-\omega} \left[\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\omega}\right]^{\frac{1}{1-\alpha}}$. The average mark-up is the higher and the real wage is the lower, first, the lower the initial mark-up chosen by intermediate good firms, $\frac{\alpha}{\alpha-1} \frac{i+\mu+\beta-\alpha\pi-\gamma}{i+\mu+\beta-\alpha\pi-\gamma-\omega}$, and second, the higher the term $\left[\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\omega}\right]^{\frac{1}{1-\alpha}}$. This term captures the fact that the average mark-up is smaller (higher) than the current optimal mark-up under price rigidity and positive (negative) growth of the optimal price: At $\omega > 0$ ($\omega < 0$), past optimal mark-ups are lower (higher) than the current value and since prices are temporarily fixed due to price rigidity, a lot of weight is put on past values. Since $\omega = \psi$ at steady state equilibrium, an increase in money growth influences the average mark-up and the wage in countervailing ways via its effects on the initial mark-up and the deviation term. The wage increases in the money growth rate at $\psi \leq 0$ and thus takes its maximum value at some $\psi > 0.^{24}$

Finally, note that rewriting equation 2.13 as

$$\frac{p^*(\tau)}{P(\tau)Q(\tau)} = \frac{\alpha}{\alpha - 1} \frac{i + \mu + \beta - \alpha\pi - \gamma}{i + \mu + \beta - \alpha\pi - \gamma - \omega} \frac{w(\tau)}{P(\tau)Q(\tau)} = \left[\frac{\mu + \beta}{\mu + \beta - (\alpha - 1)\omega}\right]^{1/(\alpha - 1)} \tag{2.15}$$

shows that regardless of the offsetting effects, the firm's initial relative price under price rigidity always increases in $\omega = \psi$.²⁵

²³Details can be found in Appendix 2.A.3. ²⁴ $\frac{\partial \frac{w}{PQ}}{\partial \psi} = \frac{\frac{w}{PQ}}{\mu+\beta-(\alpha-1)\psi} \left[-\frac{\rho+\beta+\tilde{\eta}\mu}{\rho+\beta+\tilde{\eta}\mu-\alpha\psi} \frac{\mu+\beta-(\alpha-1)\psi}{\mu+\beta-(\alpha-1)\psi+\rho+(\tilde{\eta}-1)\mu} + 1 \right] > 0$ for $\psi \leq 0$. ²⁵Note that price dispersion is necessary for this effect: If there were no price dispersion

2.2.4 Patents and the R&D sector

There is free entry to R&D. Research is undertaken by small profit maximising research firms trying to improve the quality of existing intermediate goods. The flow probability of an invention being made is governed by a Poisson process with parameter $\mu_{k_j}(\tau)$ for the firm trying to improve intermediate good k_j . For a given quality rung k_j (i.e., current position of sector j), the probability of success depends linearly on the amount of research labour $L_j^R(\tau)$:

$$\mu_{k_j}(\tau) = \phi(k_j(\tau))L_j^R(\tau) \tag{2.16}$$

For any given level of research, the probability of success decreases in the number of innovations that have already been made in that particular sector. This idea is captured by assuming $\phi'(k_j(\tau)) < 0$. In case of success for research firm j, the design of the new, improved good is sold to a new intermediate goods firm replacing the incumbent in sector j. A potential producer's maximum willingness to pay is the present value of all future profits at market entry $E\left[V_{k_j+1}(\tau) | t_{k_j} = \tau\right]$, as given in equation (2.10). Due to free entry, the patent price will be equal to this expected value.

Thus, sector j research firm's expected profit at time τ is

$$E\left[\Pi_{j}^{R}(\tau)\right] = \mu_{k_{j}}(\tau)E\left[V_{k_{j}+1}\left(\tau\right)\left|t_{k_{j}}=\tau\right] - w(\tau)L_{j}^{R}(\tau)\right]$$

Because of free entry into the research sector, firms' expected profit is zero at every instant which using (2.16) implies that either no research is undertaken $(L_i^R(\tau) = 0)$ or

$$\phi(k_j(\tau))E\left[V_{k_j+1}(\tau) \, \left| t_{k_j} = \tau \right] - w(\tau) = 0 \tag{2.17}$$

holds. Thus expected profit from research per researcher equals the wage.

For any given research effort, the probability of making an innovation

⁽i.e., the deviation term $\left[\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\omega}\right]^{\frac{1}{1-\alpha}}$ were one), all firms would charge the same price, $p^*(\tau)/Q(\tau)$. The relative price of any firm is then $p^*(\tau)/[P(\tau)Q(\tau)] = 1$ regardless of the level of the mark-up charged uniformly by all firms. The relative price would thus be independent of money growth's influence on the chosen mark-up.

decreases in the sector's position on the quality ladder. We choose a specification for $\phi(k_j(\tau))$ that implies the existence of spillovers in research: The lower the sector's quality level in comparison to aggregate quality, i.e. the further away the sector is from the research frontier, the easier is making an innovation:

$$\phi(k_j(\tau)) = (\alpha - 1) \frac{1}{\lambda} \left(\frac{q^{k_j + 1}}{Q(\tau)}\right)^{1 - \alpha}$$
(2.18)

where $1/\lambda$ is the productivity of labour in research. This standard razor's edge specification is chosen to make sure that the optimal research intensity μ can be constant and independent of a sector's position.

2.2.4.1 The R&D sector's labour demand

Research firm k_j 's labour demand is found by rearranging (2.16) and inserting $\phi(k_j(\tau))$ as defined in equation (2.18). Aggregating over all research firms, total demand for research labour thus is $L^R = \sum_{j=1}^N L_{k_j}^R(\tau) = \sum_{j=1}^N \frac{\mu}{\frac{1}{\lambda} \frac{\alpha-1}{q^{\alpha-1}} \left(\frac{q^{k_j(\tau)}}{Q(\tau)}\right)^{1-\alpha}}$ or

$$L^R = \mu \lambda \frac{q^{\alpha - 1}}{\alpha - 1} \tag{2.19}$$

2.2.4.2 Behaviour of the aggregate quality index $Q(\tau)$ and the growth rate

The innovations made in the R&D sector determine the evolution of the quality index defined in equation (2.5). In case of an innovation occurring in sector j, the sector's quality increases from $q^{k_j(\tau)}$ to $q^{k_j(\tau)+1}$, which increases the sector's contribution to the aggregate quality index from $q^{(\alpha-1)k_j(\tau)}$ to $q^{(\alpha-1)[k_j(\tau)+1]}$. The expected change in sector j's contribution can be found by weighting the quality improvement in sector j with the flow probability that an innovation will occur there at time τ . Assuming again that the flow probability of an innovation occurring, μ , is constant and equal across sectors, the expected growth rate of the quality index (2.5), γ_Q , can be calculated from the sum of these expected contributions. Further, the law of large numbers implies that the actual steady state growth rate of $Q(\tau), \gamma_Q$,

equals its expected value for $N \to \infty$.²⁶ Thus we have:

$$\gamma_Q = \mu \frac{q^{\alpha - 1} - 1}{\alpha - 1} \tag{2.20}$$

It will be shown in section (2.2.5) that at steady state, the growth rate of output γ equals the growth rate of the aggregate quality index, so we have

$$\gamma = \frac{q^{\alpha - 1} - 1}{\alpha - 1}\mu\tag{2.21}$$

2.2.5 Labour market equilibrium

The variables determining the model's production side equilibrium, most notably expected profit in the research sector, depend on the size of the final good sector. Using equilibrium in the labour market we now pin down this variable as a function of endogenous variables and employment L and then proceed to the production side equilibrium.

Equilibrium in the labour market requires that the sum of the labour demands of the intermediate goods sector, L^X , and of the research sector, L^R , equal labour supply L:

$$L = L^X + L^R \tag{2.22}$$

Inserting the labour demands from the intermediate goods sector, (2.12), and the research sector, (2.19), determines the size of the final good sector²⁷

$$\frac{Y(\tau)}{Q(\tau)} = \frac{L - \mu \lambda \frac{q^{\alpha-1}}{\alpha-1}}{\left[\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\omega} \right)^{\frac{1}{\alpha-1}} \left(\frac{\mu+\beta}{\mu+\beta-\alpha\omega} \right)^{-\frac{1}{\alpha}} \right]^{-\alpha}}$$
(2.23)

Note that for a given amount of labour employed in the intermediate goods sector, price dispersion reduces the amount of output in efficiency unit pro-

 $^{^{26}}$ For details see Appendix 2.A.4.

²⁷Note that from Appendix 2.A.3, $\frac{p^*(\tau)}{P(\tau)Q(\tau)}$ is constant. Hence, we have that for constant μ , $\frac{Y(\tau)}{Q(\tau)}$ is constant and $\gamma = \gamma_Q$ at steady state, as asserted above.

duced $Y(\tau)/Q(\tau)$.

2.2.6 Production side equilibrium

Inserting $\phi(k_j(\tau))$ from equation (2.18) and a new firm's expected market value

 $E\left(V_{k_j+1}(\tau) | t_{k_j} = \tau\right)$ (2.10) into the zero profit condition (2.17), we have the zero profit condition in the production side equilibrium:

$$\frac{\frac{1}{\lambda} \frac{Y(\tau)}{Q(\tau)} \left[\frac{p^*(\tau)}{P(\tau)Q(\tau)}\right]^{-\alpha} w(\tau)}{\left[\chi - (1-\alpha)\omega\right] \frac{\chi + \beta - \omega}{\chi + \beta - (1-\alpha)\omega}} = w(\tau)$$
(2.24)

with $\chi = i + \mu - \alpha \pi - \gamma$ and where final good production $Y(\tau)/Q(\tau)$ is the value we have determined in the equation describing labour market equilibrium, (2.23). This equation determines the research intensity μ which makes current research firms indifferent with regard to the amount of research labour used.²⁸ As can be seen, the resulting research intensity is the same for all firms making an innovation at time τ , regardless of the sector's current position on the quality ladder, consistent with our assumption in section 2.2.2.

2.2.7 Public Sector

As in earlier monetary growth models, the public sector is modelled choosing the most parsimonious specification: The state expands the money supply at a constant rate ψ and distributes seigniorage to the households in form of a lump-sum transfer.²⁹ In particular, the independent central bank perfectly controls the money supply, $M^{s}(\tau)$, by setting the constant exogenous rate ψ :

$$\frac{M^s(\tau)}{M(\tau)} = \psi \tag{2.25}$$

²⁸Note that the firm's value is $E\left(V_{k_j+1}(\tau) | t_{k_j} = \tau\right)$ because it will produce the *next* quality, $k_j + 1$ for the sector, which is about to be developed.

²⁹Cf., e.g., Gillman and Kejak [2005b], Chang [2002], Marquis and Reffett [1995], Orphanides and Solow [1990].

This simple money supply rule is sufficient for our purposes because we are not concerned with either the replication of actual data of central bank behaviour or the design of optimal monetary policy.

All revenue from money creation is allocated to households in form of a lump-sum cash transfer, $T(\tau)$

$$M^s(\tau) = T(\tau) \tag{2.26}$$

The state does not levy taxes and there is no government spending apart from the transfer of seigniorage to households.

2.2.8 Assets and Households

There is a positive number of investment funds in the economy which finance research activities. Each fund is of sufficient size to diversify the risk associated with its investments, such that they only care about the expected profit of each investment.³⁰ Funds flowing to the investment funds at time τ are used to finance current research activities. At a given real interest rate $r(\tau)$, research investment must be equal to the desired change in non-monetary savings of households, whom we now proceed to describe.

There is a continuum of households with mass one distributed uniformly on the interval [0, 1]. The infinitely lived representative household is assumed to maximise the present value of utility from consumption of the final good $c(\tau)$ and real balances $m(\tau) = \frac{M(\tau)}{P(\tau)}$ over his lifetime, where future flows of instantaneous utility are discounted with the factor $\rho > 0.3^{11}$ Assuming the rate of population growth to be zero, a standard specification for households' utility is

$$U = \int_{s=0}^{\infty} e^{-\rho s} \frac{(c(s)^{1-\theta} m(s)^{\theta})^{1-\eta} - 1}{1-\eta} ds$$
 (2.27)

³⁰At the same time, funds are not big enough to internalise existing distortions.

³¹The household needs money for transaction purposes. Instead of modelling the transactions services of money explicitly or introducing a cash-in-advance-constraint we adopt the widespread shortcut-assumption that households derive utility from holding real balances $m = \frac{M}{P}$. Feenstra [1986] shows that our case of non-separable utility for consumption and real balances is equivalent to the explicit modelling of cash holdings' transaction cost reducing function, which is the standard justification for choosing this shortcut.
where we assume $\theta \in [0, 1)$ and $\eta \ge 1$, where the latter condition is sufficient for convergence of the interval.^{32,33} The representative household maximises (2.27) subject to his budget constraint. Combining the budget constraint with a no-arbitrage-condition for the asset market, we get:

$$\dot{v(\tau)} = \frac{w(\tau)}{P(\tau)}L + \frac{T(\tau)}{P(\tau)} + r(\tau)v(\tau) - c(\tau) - [\pi(\tau) + r(\tau)]m(\tau)$$
(2.28)

where v is the real value of the household's monetary and non-monetary wealth, $\frac{w}{P}L$ is the real wage income from inelastically supplying L units of labour, $\frac{T(\tau)}{P(\tau)}$ is the real value of the transfer received from the government and r is the real interest rate.³⁴

The first-order conditions resulting from the maximization of the household's utility (2.27) subject to (2.28) are:

$$\frac{\theta}{1-\theta}\frac{c(\tau)}{m(\tau)} = r(\tau) + \pi(\tau)$$
(2.29)

and

$$[\eta + \theta (1 - \eta)] \frac{c(\tau)}{c(\tau)} - \theta (1 - \eta) \frac{m(\tau)}{m(\tau)} = r(\tau) - \rho$$
 (2.30)

Since real interest rate r and inflation π are constant at steady state, equation (2.29) implies that $c(\tau)/m(\tau)$ is constant and thus, the growth rates of consumption and real balances must be equal. Using this in equation (2.30), we have the familiar Euler equation:

$$\frac{c(\tau)}{c(\tau)} = \frac{r-\rho}{\eta} \tag{2.31}$$

³²This assumption is not restrictive. Kimball [1995], for example argues that the value one for the intertemporal elasticity of substitution $1/\eta$ that is assumed in most of the RBC literature and many New Keynesian models is implausibly high given empirical estimates. ³³At the same time, the assumption ensures that the transversality condition $\lim_{t\to\infty} \xi_t v_t e^{-\rho t} = \lim_{t\to\infty} e^{-rt} v_t = 0$ holds where ξ is the shadow value of wealth. The condition holds since the household's real wealth v grows at rate γ at steady state and $r - \gamma = \rho + (\eta - 1)\gamma > 0$ for $\eta \geq 1$.

³⁴The household receives real interest payments of $r(\tau)$ on his non-monetary assets, $v(\tau) - m(\tau)$ while the value of real money holdings depreciates at rate $\pi(\tau)$, where $\pi(\tau)$ is the rate of inflation.

Equation (2.29) states that in equilibrium the ratio of consumed goods' marginal utilities equal their cost ratio with the opportunity cost of money given by the sum of real interest rate $r(\tau)$ and inflation rate $\pi(\tau)$, i.e. the nominal interest rate.³⁵

Equilibrium in the final goods market requires that the household's optimal choice of consumption equal production:

$$c(\tau) = Y(\tau) \tag{2.32}$$

2.3 General equilibrium

We will first use households' optimal behaviour and information from the public sector to determine equilibrium in the money market. We then introduce the compiled information into the production side equilibrium to analyse the model's general equilibrium.

2.3.1 Closing the model

2.3.1.1 Money market equilibrium

Money demand must equal supply, $M^s(\tau) = M^d(\tau)$ or, given the initial money stock owned by households M(0), the growth rate of real money supply, $\frac{m^s(\tau)}{m^s(\tau)} = \psi - \pi$,³⁶ must equal the growth rate of demand for real balances $\frac{m^d(\tau)}{m^d(\tau)}$. Using final goods market equilibrium and households' optimal behaviour, we have $\gamma = \frac{c(\tau)}{c(\tau)} = \frac{m^d(\tau)}{m^d(\tau)}$.³⁷

Equalizing the growth rates of real money demand and supply shows that money market equilibrium implies that the inflation rate at steady state is the difference between the money growth rate and the economy's output

³⁵As there is no aggregate uncertainty in the model, there is no need to distinguish between expected and actual inflation, such that $i = r + \pi$.

³⁶Remember that the nominal supply is expanded at the constant rate $\frac{M^s(\tau)}{M^s(\tau)} = \psi$.

 $^{^{37}}$ See the household's static optimality condition (2.29).

growth rate

$$\pi = \psi - \gamma \tag{2.33}$$

In equation (2.15) we see that the wage $w(\tau)$ grows at rate $\gamma + \pi$.³⁸ Using (2.33) we thus have that the growth rate of marginal cost equals the growth rate of money supply:

$$\psi = \gamma + \pi = \omega \tag{2.34}$$

2.3.1.2 Research intensity in General Equilibrium

Inserting the value for $Y(\tau)/Q(\tau)$ from the labour market equation (2.22), using equilibrium in the money market (2.33), the equality of wage growth and money growth (2.34) and $i = r + \pi$, and rearranging, the R&D zero profit condition (2.24) can be rewritten as:

$$\frac{\left(\frac{L}{\lambda}-\mu\frac{q^{\alpha-1}}{\alpha-1}\right)\left(\frac{p^{*}(\tau)}{P(\tau)Q(\tau)}\right)^{-\alpha}}{\left[\left(\frac{\mu+\beta}{\mu+\beta-\alpha\psi}\right)^{\frac{-1}{\alpha}}\right]^{-\alpha}} = \left[r+\mu-(2-\alpha)\gamma\right]\frac{r+\mu+\beta-(2-\alpha)\gamma-\alpha\psi}{r+\mu+\beta-(2-\alpha)\gamma}$$

$$(2.35)$$

The LHS of equation (2.35) shows the instantaneous profits for a firm entering the market with a new patent. The RHS represents the compound discount rate for this firm's future profit streams.

Further using the value of the current optimal relative price from equation (2.15) in Appendix (2.A.3), the Euler equation (2.31) and the equation relating economic growth to research intensity (2.21), we get an equation in μ and the model's parameters:

$$\frac{\left(\frac{L}{\lambda} - \mu \frac{q^{\alpha-1}}{\alpha-1}\right) \left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\psi}\right)^{-\alpha/(\alpha-1)}}{\left[\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\psi}\right)^{\frac{1}{\alpha-1}} \left(\frac{\mu+\beta}{\mu+\beta-\alpha\psi}\right)^{-\frac{1}{\alpha}}\right]^{-\alpha}} = (\rho+\widetilde{\eta}\mu) \frac{\rho+\beta+\widetilde{\eta}\mu-\alpha\psi}{\rho+\beta+\widetilde{\eta}\mu} \quad (2.36)$$

³⁸As the RHS of (2.15) is constant at steady state, so must be the left hand side of the equation. Thus, with $p^*(\tau) = \frac{\alpha}{\alpha-1} \frac{i+\mu+\beta-\alpha\pi-\gamma}{i+\mu+\beta-\alpha\pi-\gamma-\omega} w(\tau)$, we have $\frac{w(\tau)}{w(\tau)} = \frac{P(\tau)}{P(\tau)} + \frac{Q(\tau)}{Q(\tau)}$ which using that $\gamma_Q = \gamma$ at the steady state implies $\omega = \pi + \gamma$.



Figure 2.1: Steady state equilibrium research intensity μ

with $\tilde{\eta} = 1 + (\eta + \alpha - 2) \frac{q^{\alpha-1}-1}{\alpha-1}$. Both sides of equation (2.36) reflect the dependence of the optimal research intensity μ on the new firm's value. The equation is represented graphically in the solid lines in figure 2.1.

2.3.2 Existence and uniqueness of the steady state equilibrium

Assumptions We prove the existence and uniqueness of a steady state under the following two standard conditions:

$$\frac{L}{\lambda} > \rho \tag{2.37}$$

$$\psi \leq \alpha^{-1} \beta \frac{L/\lambda - \rho}{L/\lambda - \rho_{\rho+\beta}^{\beta}}$$
(2.38)

Condition (2.37) is the usual no-growth-trap-condition familiar from the underlying real growth model.³⁹ Condition (2.38) implies that price rigidity cannot be too strong or that, for any given β , there exists an upper bound on the growth rate of money supply ψ compatible with steady state equilibrium.⁴⁰ The existence of such an upper bound is necessary in all standard

³⁹See for example Barro and Sala-i-Martin [2003].

⁴⁰Note that since $L/\lambda > \rho > 0$ (condition 2.37), condition 2.38 implies $\mu + \beta > \alpha \psi$ and hence is sufficient for aggregate intermediate good demand L^X to be well-defined (see footnote 19). Together with the assumption $\eta \ge 1$ made in section 2.2.8, it also ensures that $\mu + \beta - \alpha \psi + (\tilde{\eta} - 1) \mu + \rho > 0$ which means that the conditions from footnote 9 hold

New-Keynesian business cycle models with price rigidity.⁴¹

Proposition 1 Under conditions (2.37)-(2.38), the economy has a unique steady state equilibrium with $\mu > 0$.

The proofs to this and the following propositions can be found in Appendix 2.A.5.

Intuition Consider equation (2.36). The RHS of the equation shows the effects of the future research intensity on the compound discount rate of the new firm's entering the market for intermediate goods.⁴² Given the assumption $\eta \geq 1$ from section 2.2.8, the discount rate rises in the research intensity μ as in the model without money. This is intuitive since an increase in μ means that the probability of being replaced increases, lowering the weight attached to potential future profits.⁴³

Now turn to the LHS of equation (2.36) which shows how the instantaneous profit associated with the production of the new intermediate good depends on the current and past values of μ . In the case without effective price rigidity,⁴⁴ the LHS simplifies to $L/\lambda - \mu \frac{q^{\alpha-1}}{\alpha-1}$ and linearly decreases in μ (see LHS| $_{\beta\to\infty}$ -curve in figure 2.1). This is because an increase in current research intensity μ spurs demand for research labour $\frac{q^{\alpha-1}}{\alpha-1}\mu$ ("investment"),

such that a firm's optimal price p^* is well-defined.

⁴¹The existence of an upper bound is commented on by, e.g. Ascari [2004] and King and Wolman [1996]. In the baseline example we introduce in section 2.5, the maximum money growth rate is $\psi = 0.180$, see footnote 62.

⁴²To help intuition, we stress the fact that the compound discount rate depends on the *future* value of μ while instantaneous profits depend on the variable's *current* and *past* values. Equation (2.36) shows that these values are of course identical at steady state equilibrium.

⁴³In addition to this direct effect of an increase in μ there are also several indirect effects of an increase in μ on the discount rate because μ is proportional to the growth rate γ : The real interest rate rises in γ while because of the decrease in inflation caused by a rising γ the nominal interest rate $i = \rho + \psi + (\eta - 1)\gamma$ may rise or fall in γ . Demand for the good and hence, the growth rate of profits falls in γ because although aggregate demand increases proportionately in γ , the decrease in inflation slows down the erosion of the good's relative price. For $\eta + \alpha - 2 > 0$, the positive effects of γ on the discount rate outweigh the negative effects.

⁴⁴That is, either in the absence of price rigidity ($\beta \to \infty$) or in case existing price rigidity is irrelevant because money supply is constant, $\psi = 0$.

which for a given size of the labour force reduces production of the final good ("consumption") and hence, demand for the new firm's good and its profits. Under price rigidity, this linear curve is shifted by the two already known terms depending on money growth ψ and price rigidity β that reflect the initial mark-up effect, $\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\psi}\right)^{-\alpha/(\alpha-1)}$, and the price dispersion effect $\left[\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\psi}\right)^{\frac{1}{\alpha-1}}\left(\frac{\mu+\beta}{\mu+\beta-\alpha\psi}\right)^{-\frac{1}{\alpha}}\right]^{-\alpha}$, respectively, which will be important for our discussion of comparative statics.

Both these effects depend on the total amount of price rigidity or flexibility, which depends not only on β but also on the innovation rate μ . We show in Appendix 2.A.5 that as depicted in figure 2.1, these influences of μ on price rigidity make the LHS a concave function for $\beta < \infty$.⁴⁵ Yet note that the frequency of innovation μ is small relative to the frequency with which incumbents can change prices β , so that the effect of μ on $\mu + \beta$ is small. Therefore, these effects are quantitatively small both in reality and in our calibrated examples.

Existence and uniqueness follow from the shapes of the two curves.

Note that given assumptions (2.37)-(2.38) and concavity of the LHScurve, the slope at the steady state equilibrium of the LHS-curve is smaller than that of the RHS-curve. This implies that in spite of the discussed positive effect of μ on instantaneous profits, the incentive to innovate at equilibrium decreases in μ , as in the model without money.

2.4 Comparative statics

The comparative static properties of the steady state equilibrium will be analysed using the zero profit condition (2.36) and figures 2.1 and 2.2.⁴⁶

⁴⁵Note that the slope of the LHS-curve is positive for small μ as in figure 2.1 if $0 < \lim_{\mu \to 0} \frac{\partial LHS}{\partial \mu} = -\frac{q^{\alpha-1}}{\alpha-1} \frac{\beta-\alpha\psi}{\beta} + \frac{L}{\lambda} \frac{\alpha\psi}{\beta^2}$. Whether this is the case is irrelevant for the existence and uniqueness of the equilibrium (concavity of the LHS-function is sufficient) and for the later discussion of comparative statics, where it is sufficient that at the equilibrium, $\partial LHS/\partial \mu < \partial RHS/\partial \mu$, which is assured by the additional assumption (2.38) and $\eta \geq 1$



Figure 2.2: Comparative statics for $\beta < \infty$: money growth rate

2.4.1 Economy without price rigidity: Superneutrality of money

Proposition 2 In the limiting case without rigidities $(\beta \to \infty)$, money is superneutral. In particular, $\lim_{\beta\to\infty} \frac{\partial\mu}{\partial\psi} = \lim_{\beta\to\infty} \frac{\partial\gamma}{\partial\psi} = 0$.

The proof can be found in Appendix 2.A.5.

Intuition Without price rigidity, relative prices are undistorted such that the size of the final good sector, the firm's relative price and the only discount rate relevant for the firm are independent of money growth. The zero profit condition reduces to the one from the model without money and all other comparative static properties are those of the model without money.

from the household's problem.

⁴⁶We concentrate on comparative statics with respect to ψ and β since the effects on growth of the other parameters are standard: An increase in the household's rate of time preference ρ raises the interest rate r and the discount rate for a new firm's profits, which reduces the R&D-incentive and growth. In figure 2.1, the increase in ρ causes an upward movement of the RHS curve, lowering μ . The size of the research-productivity-adjusted labour force L/λ has a scale effect of on the firm's profits and growth. In figure 2.1, an increase in L/λ causes an upward movement of the LHS curve, which leads to a higher research intensity μ .

2.4.2Economy with price rigidity: Negative growth effect of money growth

For $\beta < \infty$, the money growth rate ψ affects the zero profit equation (2.36) via three channels: The price dispersion effect and the initial price effect affect the innovating firm's instantaneous profits on the LHS of equation (2.36), while the price erosion effect changes the firm's compound discount rate on the RHS of the equation. All these effects on economic growth operate through money growth's influence on relative prices.

Negative price dispersion effect of non-zero money growth The price dispersion effect of money growth is captured in the term

 $\left[\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\psi} \right)^{\frac{1}{\alpha-1}} \left(\frac{\mu+\beta}{\mu+\beta-\alpha\psi} \right)^{-\frac{1}{\alpha}} \right]^{-\alpha} \text{ on the LHS of equation (2.36). The firm's}$ instantaneous profits decrease in price dispersion which in turn increases in the absolute value of the money growth rate ψ .⁴⁷ This is because an increase in the absolute value of ψ raises the absolute value of the growth rate of the optimal price which under price rigidity increases price dispersion regarding intermediate goods. This increases the disparity in the amounts of intermediate goods used in the final good sector, which in turn reduces production efficiency and thus lowers the intermediate good producer's instantaneous profits. Figure 2.2 depicts the case $\psi > 0$, where an the increase in ψ ceteris paribus causes a downward shift of the LHS-curve, reducing the research intensity μ .⁴⁸

To get a more detailed intuition, refer back to equation (2.12). Remember that the price dispersion in the intermediate goods sector caused by a nonzero growth rate ψ of marginal costs and optimal prices distorts demand towards intermediate goods with old (new) low prices when money growth is positive (negative).⁴⁹ Due to the concavity in intermediate goods of the

$${}_{47}\partial\left[\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\omega}\right)^{\frac{1}{\alpha-1}}\left(\frac{\mu+\beta}{\mu+\beta-\alpha\omega}\right)^{-\frac{1}{\alpha}}\right]^{-\alpha} \ge 0 \quad f_{\alpha\alpha\beta} < 0$$

 $\frac{47}{\frac{\partial \left[\left(\mu+\beta-(\alpha-1)\omega\right)\right]}{\partial \psi}} \stackrel{(\mu+\beta-\alpha\omega)}{\geq} 0 \text{ for } \psi \ge 0, \text{ see footnote 21.}$ $\frac{48}{48} \text{At } \psi < 0, \text{ the LHS-curve would be shifted upwards by an increase in } \psi.$

⁴⁹For $\psi = 0$, marginal cost and optimal prices are constant over time. Intermediate goods producers have no desire to readjust their initial prices so that there is no price dispersion despite the presence of nominal rigidity.

final good production function, this leads to inefficient production: Given current demand for research labour, and given inelastic labour supply L, the resulting labour employed in the intermediate goods sector indirectly produces the smaller an amount of final goods, the higher the deviation of ψ from zero.⁵⁰ As demand for the new intermediate good is proportional to the amount of output in efficiency units produced, the new producer's instantaneous profits and the incentive to innovate decrease (increase) in ψ for $\psi > 0$ ($\psi < 0$).

Negative initial relative price effect of an increase in money growth The term $\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\psi}\right)^{-\alpha/(\alpha-1)}$ on the LHS of equation (2.36) shows that the new intermediate good firm's instantaneous profits depend negatively on the firm's initial relative price because demand for the good falls in the good's relative price. An increase in ψ raises the initial relative price, reducing demand for the good and therefore profits. The increase in ψ thus ceteris paribus reduces the incentive to innovate, reflected in the downward shift of the LHS-curve in figure 2.2.

As explained in section (2.2.2.1), for $\psi \neq 0$ intermediate goods firms set a forward-looking initial mark-up differing from the static optimum one to offset the effect of the money growth rate, which is also the growth rate of marginal cost, on their relative price while their price is fixed. An increase in the money growth rate raises the extent of relative price erosion and hence, the optimal forward-looking initial mark-up. Also, the level of the real wage is influenced by an increase in ψ as discussed in section 2.2.3. Taking both effects into account, in general equilibrium with price rigidity an increase in ψ raises the relative price of the innovating firm at market entry which reduces demand for the firm's good and therefore, profits.⁵¹

Taking together the two effects, the LHS of equation (2.36) always decreases in ψ : At $\psi > 0$, both effects reduce the firm's instantaneous profits, at

⁵⁰Remember that since intermediate goods are produced one for one with labour, labour indirectly produces the final good.

⁵¹Remember from the discussion of equation (2.10) that the firm's optimal price was chosen so as to offset the effect of money growth on the firm's profit *per unit* so that the effect of ψ on the relative price only affect profits via the quantity demanded.

 $\psi < 0,$ the negative effect via the relative price dominates the price dispersion effect.

Positive relative price erosion effect of an increase in money growth The RHS of equation (2.36) is the new firm's compound discount rate that was first discussed in section 2.2.2.2. An increase in ψ lowers the compound discount rate because faster relative price erosion increases demand and the profit growth rate, ceteris paribus raising the incentive to innovate. In figure 2.2, the RHS-curve is shifted down by an increase in ψ , ceteris paribus increasing μ .

Going into more detail, remember that the compound discount rate consists of the discount rate of a firm under price flexibility corrected by a factor that contains the appropriate discount rates for a firm under price rigidity for periods where prices can be changed or are fixed, respectively. The discount rates for periods where prices can be changed (both under flexibility and under rigidity) are unaffected by money growth. Yet money growth increases profit growth in periods where prices are fixed, reducing the associated discount rate:⁵² While prices are fixed, the new good's relative price erodes at rate $-\pi$, leading to a growth rate $\alpha\pi$ of demand for the good. The rising demand translates into a higher growth rate of the new intermediate firm's profits.⁵³ An increase in ψ ceteris paribus raises inflation π and therefore reduces the discount rate, ceteris paribus raising the incentive to innovate and the research intensity μ .

Negative net effect of money growth on economic growth We have discussed that for $\psi > 0$, the negative effects of an increase in money growth ψ cause a downward shift of the LHS-curve while the positive relative price erosion effect entails a downward shift of the RHS-curve. Given the shapes of the curves discussed in section 2.3.2, it remains only to clarify which effect

 $^{^{52}\}mathrm{Remember}$ that the discount rate is the obsolescence-adjusted interest rate minus the profit growth rate.

 $^{^{53}}$ See equation (2.3) for the determinants of demand faced by the firm.

on the research intensity μ is stronger. The following proposition is proven in Appendix 2.A.5:

Proposition 3 An increase in the steady state money growth rate ψ decreases the research intensity μ and the real growth rate $\gamma: \frac{d\mu}{d\psi} < 0, \frac{d\gamma}{d\psi} < 0.$

Money growth and inflation thus reduce economic growth in our model, in line with the stylised fact presented in the introduction. To get an intuition why growth depends negatively on the money growth rate over the entire range of admissible values, note first that *holding constant* the level of the real wage, an increase in ψ lowers (raises) the incentive to innovate at $\psi > 0$ $(\psi < 0)$. This is due to the fact that the price dispersion effect increases in the absolute value of ψ and due to money growth's effect on the optimal mark-up (initial price effect given w/(PQ)) and effective mark-ups (relative price erosion effect): Holding constant the real wage, the firm cannot be made better of by imposing a restriction on the choice and adjustment of its markup, so profits are highest when the restriction does not bind, which is the case at $\psi = 0$ where due to constant marginal cost there is no incentive to change prices. Profits decrease as the restriction on individual price setting becomes more binding, i.e. as the absolute value of the growth rate of marginal cost ψ rises.⁵⁴

Since the level of the real wage also influences the initial relative price effect,⁵⁵ the fact that growth decreases in ψ at $\psi < 0$, too, must be due to this effect. From the discussion in section 2.2.3, we know that the wage increases in ψ at $\psi \leq 0$. While we have not shown this analytically, in calibrated examples it decreases in the money growth rate at large admissible values of ψ . Since an increase in the real wage raises the firm's initial relative price, lowering demand and profits, it remains only to clarify why the effect of an increase in ψ on the real wage level dominates at $\psi < 0$ but not at

⁵⁴The sign of $d\mu/d\psi$ for given w/(PQ) is again determined by the sign of $\frac{\partial LHS}{\partial \psi} - \frac{\partial LHS}{\partial \psi}$. Straightforward calculations yield $\left(\frac{\partial LHS}{\partial \psi} - \frac{\partial LHS}{\partial \psi}\right)\Big|_{\frac{W}{PQ}} =$

 $^{-\}psi \left[\frac{\alpha(\alpha-1)LHS}{(\rho+\beta+\tilde{\eta}\mu-\alpha\psi)[\rho+\beta-(\alpha-1)\psi+\tilde{\eta}\mu]} + \frac{\alpha LHS}{[\mu+\beta-(\alpha-1)\psi](\mu+\beta-\alpha\psi)} \right] \stackrel{\leq}{\leq} 0 \text{ for } \psi \stackrel{\geq}{\geq} 0.$ ⁵⁵As opposed to the mark-up erosion effect, where only the growth rate of the relative

⁵⁵As opposed to the mark-up erosion effect, where only the *growth rate* of the relative price and hence of wages matters.

positive values of ψ . To see this, note that since the value of the elasticity of substitution between intermediate goods α exceeds unity, the effect of any change in the initial relative price on demand is stronger when the relative price is smaller than one than when it exceeds unity- and given price rigidity, the initial price is smaller (bigger) than one when marginal cost shrinks (grows) over time, i.e. $\psi < 0$ ($\psi > 0$).⁵⁶

2.4.3 Economy with price rigidity: Negative growth effect of the exogenous level of rigidity under inflation

We now use equation (2.36) to discuss the comparative statics of the steady state for $\beta < \infty$ with regard to the level of exogenous rigidity, which is inversely related to β .

Proposition 4 An increase in the level of rigidity (i.e., decrease in β) decreases (increases) the research intensity μ and the real growth rate γ when money growth is positive (negative): $\frac{d\gamma}{d\beta} \gtrsim 0$ for $\psi \gtrsim 0$.

Intuition The effect of a higher β on γ is the result of three already familiar effects: Firstly, for $\psi \neq 0$ a higher β (less rigidity) reduces the *price* dispersion effect: A higher value of β means that prices can be changed more often and thus are less dispersed, so as reflected in the term

 $\left[\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\psi}\right)^{\frac{1}{\alpha-1}}\left(\frac{\mu+\beta}{\mu+\beta-\alpha\psi}\right)^{-\frac{1}{\alpha}}\right]^{-\alpha}$, demand is less distorted towards relatively cheap goods, raising production efficiency. Via the mechanism explained above, an increase in β therefore raises the incentive to innovate. Secondly, by shifting more weight to present as opposed to future profits and reducing price dispersion, an increase in β reduces the deviation of the firm's initial mark-up and relative price from their values under flexibility. An increase in β thus reduces the *initial relative price effect*. The decrease (increase) in the initial relative price raises (lowers) demand, profit and thus, ceteris paribus the incentive to innovate at $\psi > 0$ ($\psi < 0$). The effect on price dispersion

 $^{{}^{56}}$ See equation (2.15).

(the initial relative price) dominates at $\psi > 0$ ($\psi < 0$) so that an increase in β shifts the LHS-curve in figure 2.1 upward (downward), ceteris paribus increasing (decreasing) μ .

At the same time, the RHS of equation (2.35) increases in β (decreases in the level of rigidity), so that an increase in β causes an upward shift of the RHS-curve in figure 2.1, ceteris paribus reducing μ . This is because a higher degree of price flexibility β reduces the extent of *relative price erosion* which only takes place while prices are fixed. Thus the increase in β shifts more weight to the discount rate for periods when prices are flexible. As this rate is higher than the fix-price discount rate due to the relative price erosion effect of money growth, an increase in β raises the compound discount rate, ceteris paribus reducing μ .⁵⁷

The intuition for the net effect of β on μ is closely connected to the effects of money growth in that a decrease in β magnifies the effect of a deviation of ψ from zero: At $\beta \to \infty$, all discussed effects of money on growth disappear. Distortion of relative prices is the more pronounced, the higher rigidity. So when at $\psi > 0$ this deviation of money growth from zero is detrimental to growth (growth-enhancing at $\psi < 0$), an increase in β that mitigates the consequences of money growth raises (lowers) growth.

2.5 Calibrating the model: Comparative Statics

We calibrate the model to see if the magnitude of the growth-reducing net effect of money growth is consistent with the findings of empirical studies. We use standard parameter values from the literature and use steady state considerations to calibrate our model. Our calibration of the household's intertemporal rate of substitution η^{-1} and discount rate ρ and money's re-

⁵⁷The compound discount rate is given by the discount rate for a firm under total price flexibility times a factor that corrects for price rigidity's effect on the discount rate. As seen above, through the mark-up erosion effect rigidity leads to a lower discount rate in fix-price periods than when prices can be changed. An increase in β (less rigidity) means that this rate is relevant less often, so that the downward correction of the flex-price rate is reduced and the compound discount rate increases.

latively small weight θ in the household's utility function follow standards from the calibration of New Keynesian business cycle models. For the same reason and in line with empirical estimates, α is chosen such that the markup in a steady state with constant marginal cost $\alpha/(\alpha - 1)$ is between 10% and 20%.⁵⁸ The values for q imply that innovations bring about quality improvements of 1% to 50% relative to existing products. Our choice of β is based on empirical findings that prices are fixed for two to five quarters in the literature,⁵⁹ and entails an upper bound on (partly endogenous) price rigidity of one year when growth is at its minimum value zero.⁶⁰ Using data on average US M1 growth,⁶¹ we further use the baseline money growth rate $\psi = 0.055$ in all examples. Table 2.1 lists parameters' values in our baseline case and the ranges wherein parameters were varied:^{62,63}

In the baseline case, firms' optimal price implies a mark-up of 15.2% over marginal cost, while the average time during which prices are fixed is 0.54 years, i.e. 6.5 months. Both results are compatible with the discussed

 61 We calculated the average growth rate of the monetary aggregate M1 in the US between 1979 and 2004 based on data from www.federalreserve.gov/releases/h6/hist/. The narrow aggregate was chosen because money is modelled as yielding utility to the household. This can best be motivated with the transaction services of money as opposed to its services as a store of value. By definition of the monetary aggregates, the transaction function is best reflected by M1.

⁶²In all checked examples, we examined the growth rate effects of money growth rates ψ with $\psi^0 < \psi < \psi^{\max}$. Here, ψ^{\max} is the maximum money growth rate compatible with a given level of β in constraint (2.38) calculated in footnote 40: $\psi^{\max} = \frac{\beta(\frac{L}{\lambda} - \rho)}{\alpha(\frac{L}{\lambda} - \rho\frac{\beta}{\rho + \beta})}$. In the baseline example, $\psi^{\max} = 0.180$. As our analysis applies to developed countries, this is not a severe restriction. The lower bound on ψ is implied by the household's static optimality condition (2.29) where the nominal interest rate $i = \rho + (\eta - 1)\gamma + \psi$ must be positive for the existence of equilibrium. As $\gamma = \gamma(\psi)$, the implied value of ψ is endogenous. We therefore choose $\psi^0 = -\rho$ which is sufficient for i > 0 for any $\eta \ge 1$.

⁶³The range of values for L/λ is determined by the fact that that we use the degree of freedom in choosing this variable so as to always make sure that $\gamma = 0.02$ at $\psi = 0.055$ given our other parameter choices.

⁵⁸Cf. e.g. to Chari, Kehoe and McGrattan [2000]. Markups of this dimension are reported for the US by Basu and Fernald [1995, 1997] and for Germany by Linnemann [1999].

⁵⁹See section 1.5.1.3 of the introductory chapter.

⁶⁰Remember that due to innovating firms' market entry with new prices, the average interval of fixed prices is partly endogenous in our model and given by $(\beta + \mu)^{-1} = \left(\beta + \frac{\alpha - 1}{q^{\alpha - 1} - 1}\gamma\right)^{-1}$.

Parameter	Baseline	Range	Parameter	Baseline	Range
η	2	[1, 2]	β	1.8	[1,3]
α	10	[6, 11]	q	1.2	[1.01, 1.5]
ρ	0.015	[0.01, 0.03]	ψ	0.055	$\left(\psi^{0},\psi^{\max} ight)$
θ	0.01	[0.01, 0.5]	L/λ	0.29455	

Table 2.1: Parameter values used for calibration



Figure 2.3: Function $\gamma(\psi)$ in baseline calibration

empirical estimates. The output growth rate is 2% per annum. At the same time, the average frequency of innovation $(1/\mu)$ is 23.1 years. So indeed we are investigating the effects of recurrent *short*-term frictions on *long*-term innovation.

Varying the policy parameter ψ shows that the size of the effect of an increase in the money growth rate corresponds closely to the findings of the empirical literature: Increasing the annual money growth rate from 1% to 10% reduces economic growth by some 0.2 percentage points from 2.07% to 1.87% per annum.⁶⁴ Figure 2.3 shows the rate of economic growth as a function of the money growth rate ψ in our baseline example.

The negative effect of an increase in money growth from $\psi = 0.01$ to

⁶⁴As reported in the introduction, Motley [1998], Barro [1996] Fischer [1993] all report results of a similar magnitude. In particular, Barro [1996] finds that a 10 percentage point increase in M1 growth (inflation) reduces annual GDP growth by about 0.23 (0.24) percentage points.

 $\psi = 0.1$ on economic growth is the net effect of quantitatively larger individual effects: The countervailing relative price erosion effect and initial relative price effect are quite sizeable individually, each changing the growth rate by over one percentage point. In our baseline example where price rigidity is not very severe, the effect through price dispersion is quantitatively small (the growth rate changes by -0.007 percentage points for the described money growth variation) in comparison to the other effects. Yet the relative importance of the three individual effects varies considerably under different parameter constellations: The importance of the production inefficiency channel rises considerably as rigidity becomes more severe.

The net effect of price rigidity on growth at positive money growth rates is negative but small: Decreasing β from $\beta = 1.8$ to $\beta = 1.5$ at $\psi = 0.055$, the growth rate drops from 2.00% to 1.96%.⁶⁵ The effect of rigidity on growth is stronger for higher rates of money growth and vice versa: The drop in the growth rate resulting from an increase of ψ from 1% to 10% growth is .33 percentage points at $\beta = 1.5$, compared to .20 percentage points at $\beta = 1.8$.⁶⁶

2.6 Conclusion

The analysis of the Schumpeterian growth model with small nominal frictions has shown that indeed, the influence of price rigidity is not restricted to the short-run. At the unique steady state equilibrium, it allows money growth to have a level effect on production efficiency and on the relative prices charged by intermediate goods producers. Since an intermediate good producer's profits determine the incentive to engage in research activities for the development of improved intermediate goods, this implies a level effect on the total amount of resources devoted to R&D. Finally, since R&D activities leading to growth in the quality of intermediates are the engine of growth in the economy, the level effects of inflation induced by nominal price rigidity

 $^{^{65}\}mathrm{The}$ implied change in the average period where prices are fixed is from 6.5 to 7.8 months.

⁶⁶For $\psi = 0.1$, the growth rate drops from 1.87% to 1.74% when β is decreased from $\beta = 1.8$ to $\beta = 1.5$.

also influence the economy's long-run output growth rate.

The interaction of money growth's influences on intermediate good firms' relative prices (via optimal and effective mark-ups and the real wage level) and production efficiency (via the degree of dispersion in intermediate good prices) makes the economic growth rate a decreasing function of the money growth rate over the entire range of admissible money growth rates. Although the examined transmission mechanisms differ, this result is shared by most of the papers on endogenous growth and inflation presented in section 1.4.1.2 of the introductory chapter. While the quantitative results of our model concerning the growth effects of discrete large increases in inflation are in line with the results of the empirical literature, neither our model nor those discussed in section 1.4.1.2 are compatible with the pieces of evidence discussed in section 1.4.2 which suggest that the effect of small positive inflation rates on growth may be insignificant or even positive. In chapter 5, the influence of money growth on endogenous labour supply is at the origin of a non-monotone money-growth relationship, where economic growth increases in the money growth rate ψ at small values of ψ . Before we turn to this variation of our model, an issue more closely connected to this chapter's analysis deserves further investigation: Our analysis so far has focused exclusively on the steady state equilibrium of the Schumpeterian model with price rigidity. This is a valid simplification for the analysis of the model economy's long-run behaviour only in case the steady state is stable, such that the economy for any given initial conditions converges to the steady state in the long-run. Therefore, chapter 3 is dedicated to the analysis of the steady state's stability.

2.A Appendix to chapter 2

2.A.1 Derivation of a new firm's market value

The firm's market value is the discounted sum of profits from future periods s where the profits are weighted due to two independent sources of uncertainty: The first weight is given by the probability $e^{-\mu(s-\tau)}$ of not having been replaced by time s. The second source of uncertainty is given by the firm's price in period s: The price charged can be any past optimal price $p^*(\theta)$ with $\theta \in (\tau, s)$ depending on when the last reset signal for the price was received between τ and s. Thus, the price charged at time s can be represented as a weighted sum of the past optimal prices, where the weights are as follows: The flow probability that a signal to reset prices was received in period θ is β . With probability $e^{-\beta(s-\theta)}$, no signal was received between θ and s given that a signal was received in θ .⁶⁷ As these two events are independent, the probability of having last reset one's price due to a price reset signal at $\theta \in (\tau, s)$ is $\beta e^{-\beta(s-\theta)}$. Additionally, if no reset signal has been received up to period s, the firm's price will continue to be $p^*(\tau)$, which has probability $(1 - \int_{\tau}^{s} \beta e^{-\beta(s-\theta)} d\theta)$. Since the processes for innovations and reset signals are independent, the joint probability of the described events takes on a multiplicative form:

$$E\left(V_{k_{j}}|\tau\right) = A\left(\tau\right) \int_{\tau}^{\infty} e^{-\chi(s-\tau)} \left[\int_{\tau}^{s} \beta e^{-\beta(s-\theta)} p^{*}\left(\theta\right)^{1-\alpha} d\theta + \left(1 - \int_{\tau}^{s} \beta e^{-\beta(s-\theta)} d\theta\right) p^{*}\left(\tau\right)^{1-\alpha} \right] ds$$
$$-A\left(\tau\right) w\left(\tau\right) \int_{\tau}^{\infty} e^{-(\chi-\omega)(s-\tau)} \left[\int_{\tau}^{s} \beta e^{-\beta(s-\theta)} p^{*}\left(\theta\right)^{-\alpha} d\theta + \left(1 - \int_{\tau}^{s} \beta e^{-\beta(s-\theta)} d\theta\right) p^{*}\left(\tau\right)^{-\alpha} \right] ds$$

where we define $\chi = i + \mu - \alpha \pi - \gamma$ and $A(\tau) = q^{k_j(\alpha-1)}Y(\tau)P(\tau)^{\alpha}$ and where $E(V_{k_j}|\tau)$ is a shorthand form for $E(V_{k_j}(\tau)|t_{k_j}=\tau)$.

Making use of the fact that the optimal price $p^*(\theta)$ grows with the growth rate of marginal cost, ω and solving the integrals associated with the prob-

⁶⁷Here, we have been able to definitize the constant $\tilde{B} = 1$ since we know that the probability of receiving two or more signals at time θ is negligible (see footnote 7) for details.

ability of receiving a price resetting signal yields:

$$E(V_{k_j} | \tau) = A(\tau) p^*(\tau)^{1-\alpha} \frac{\beta \int_{\tau}^{\infty} e^{-[\chi - (1-\alpha)\omega](s-\tau)} \left[1 - e^{-[\beta - (1-\alpha)\omega](s-\tau)}\right] ds}{\beta - (\alpha - 1)\omega}$$
$$+ A(\tau) p^*(\tau)^{1-\alpha} \int_{\tau}^{\infty} e^{-(\chi + \beta)(s-\tau)} ds$$
$$- A(\tau) p^*(\tau)^{-\alpha} w(\tau) \frac{\beta}{\beta - \alpha\omega} \int_{\tau}^{\infty} e^{-[\chi - (1-\alpha)\omega](s-\tau)} \left(1 - e^{-(\beta - \alpha\omega)(s-\tau)}\right) ds$$
$$- A(\tau) p^*(\tau)^{-\alpha} w(\tau) \int_{\tau}^{\infty} e^{-(\chi + \beta - \omega)(s-\tau)} ds$$

Calculating the value of the integrals yields

$$E(V_{k_{j}}|\tau) = A(\tau) p^{*}(\tau)^{1-\alpha} \left\{ \frac{\beta}{\beta - (\alpha - 1)\omega} \left(\frac{1}{\chi - (1 - \alpha)\omega} - \frac{1}{\chi + \beta} \right) + \frac{1}{\chi + \beta} \right\}$$
$$-A(\tau) w(\tau) p^{*}(\tau)^{-\alpha} \left\{ \frac{\beta}{\beta - \alpha\omega} \left(\frac{1}{\chi - (1 - \alpha)\omega} - \frac{1}{\chi + \beta - \omega} \right) + \frac{1}{\chi + \beta - \omega} \right\}$$

Multiplying out the terms in curly brackets, we have

$$E\left(V_{k_{j}}\left(\tau\right)\left|t_{k_{j}}=\tau\right.\right)=A\left(\tau\right)p^{*}\left(\tau\right)^{-\alpha}\frac{\chi+\beta+(\alpha-1)\omega}{\chi-(1-\alpha)\omega}\left[\frac{p^{*}\left(\tau\right)}{\chi+\beta}-\frac{w\left(\tau\right)}{\chi+\beta-\omega}\right]$$

Finally using the equation for the optimal price (2.8) and reinserting $\chi = i + \mu - \alpha \pi - \gamma$ and $A(\tau) = q^{k_j(\alpha-1)}Y(\tau)P(\tau)^{\alpha}$ we have equation (2.10) in the main text.

2.A.2 Derivation of the intermediate goods sector's labour demand

Equation (2.11) can be rewritten as

$$L^{X}(\tau) = Y(\tau)P(\tau)^{\alpha}Q(\tau)^{\alpha-1}\sum_{k=1}^{k^{\max}} d_{k}(\tau) \left(\frac{q^{k}}{Q(\tau)}\right)^{\alpha-1}\sum_{\{j|k_{j}=k\}} (p_{k_{j}}(\tau))^{-\alpha}$$

where p_{k_j} is the price of sector j that is at quality rung k and $d_k(\tau)$ is the number of sectors at quality rung k at time τ .

Following Benhabib, Schmitt-Grohé and Uribe [2001a] and [2001b], Leith and Wren-Lewis [2000] and Wolman [1999], the average price effective at time τ can be expressed as a weighted average of past optimal prices, where the weights $f(s,\tau)$ refer to the probability that a price valid at time τ has not been changed since time s. As the timing of innovations is independent of a sector's position on the quality ladder q^{k_j} , the structure of prices for a given q^k is the same as the structure for all sectors. Thus we have

$$\sum_{k=1}^{k^{\max}} d_k(\tau) \left(\frac{q^k}{Q(\tau)}\right)^{\alpha-1} \sum_{\{j|k_j=k\}} p_{k_j}(\tau)^{-\alpha} = \sum_{k=1}^{k^{\max}} d_k(\tau) \left(\frac{q^k}{Q(\tau)}\right)^{\alpha-1} \sum_{j=1}^{N} p_j(\tau)^{-\alpha}$$
$$= \int_{-\infty}^{\tau} f(s,\tau) [p^*(s)]^{-\alpha} ds$$

with $\sum_{k=1}^{k^{\max}} d_k(\tau) \left(\frac{q^k}{Q(\tau)}\right)^{\alpha-1} = \frac{\sum_{k=1}^{k^{\max}} d_k(\tau)q^{k(\alpha-1)}}{Q(\tau)^{\alpha-1}} = 1$ by equation (2.5) and therefore

$$L^{X}(\tau) = P(\tau)^{\alpha} Q(\tau)^{\alpha} \frac{Y(\tau)}{Q(\tau)} \int_{-\infty}^{\tau} f(s,\tau) p^{*}(s)^{-\alpha} ds \qquad (2.39)$$

A price in effect at time τ dates from time *s* if there was either an innovation or a price reset signal at time *s* and if there was no innovation between times *s* and τ and if no price reset signal was received in the same period. As explained in Appendix 2.A.1, the probability of no innovation and no price reset signal between times *s* and τ is $e^{-(\mu+\beta)(\tau-s)}$. The flow probability of an innovation or a price reset signal occurring at time *s* is $\mu + \beta$. Thus, we have $f(s,\tau) = (\mu + \beta) e^{-(\mu+\beta)(\tau-s)}$.⁶⁸ Using this and steady growth of p^* at rate ω in equation (2.39), we have

$$L^{X}(\tau) = \left[\frac{p^{*}(\tau)}{P(\tau)Q(\tau)}\right]^{-\alpha} \frac{Y(\tau)}{Q(\tau)} \int_{-\infty}^{\tau} \left(\mu + \beta\right) e^{-(\mu+\beta)(\tau-s)} e^{(\alpha\omega)(\tau-s)} ds$$

⁶⁸Note that the two Poisson processes governing innovations and the occurrence of price adjustment sigals are stochastically independent so that that the joint probability is the product of the individual probabilities.

Solving the integral which converges for $\mu + \beta > \alpha \omega$ leads to (2.12) in the main text.

2.A.3 Derivation of equation (2.13)

Analogously to the procedure in Appendix 2.A.2, the final good price (2.4) at steady state can be rewritten as

$$P(\tau)^{1-\alpha} = Q(\tau)^{(\alpha-1)} \left(\mu + \beta\right) \int_{-\infty}^{\tau} e^{-(\mu+\beta)(\tau-s)} p^*(s)^{1-\alpha} ds$$

From (2.8), we have that the optimal price at steady state grows at rate ω . Convergence of the integral is then ensured by assumption (2.38). Using the growth rate of the optimal price ω , we can calculate the integral's value and rearrange terms yielding equation (2.13).

2.A.4 Derivation of the growth rate of the Quality index

The expected growth rate of the aggregate quality index is

$$E\left[\widehat{Q(\tau)}\right] = \frac{1}{\alpha - 1} E\left[\sum_{j=1}^{N} \widehat{q^{(\alpha-1)k_j(\tau)}}\right]$$
(2.40)

where a hat denotes the proportional growth rate of variable x, $\hat{x} = \frac{dx/dt}{x}$. The growth rate of the sum in equation (2.40) is equal to the sum of the individual growth rates weighted with the sector's share in the aggregate quality index⁶⁹:

$$E\left[\widehat{Q(\tau)}\right] = \frac{1}{\alpha - 1} \sum_{j=1}^{N} E\left[q^{(\alpha - 1)k_j(\tau)} \frac{q^{(\alpha - 1)k_j(\tau)}}{\sum_{j=1}^{N} q^{(\alpha - 1)k_j(\tau)}}\right]$$

Using that the expected value of the product of two random variables is given by E[XY] = E[X]E[Y] + Cov[XY], where Cov[XY] denotes the

⁶⁹For example, $\widehat{X+Y} = \widehat{X} \frac{X}{X+Y} + \widehat{Y} \frac{Y}{X+Y}$.

covariance between X and Y, gives

$$E\left[\widehat{Q(\tau)}\right] = \frac{1}{\alpha - 1} \sum_{j=1}^{N} E\left[\widehat{q^{(\alpha - 1)k_j(\tau)}}\right] E\left[\frac{q^{(\alpha - 1)k_j(\tau)}}{\sum_{j=1}^{N} q^{(\alpha - 1)k_j(\tau)}}\right] + \frac{1}{\alpha - 1} \sum_{j=1}^{N} Cov\left[\widehat{q^{(\alpha - 1)k_j(\tau)}} \frac{q^{(\alpha - 1)k_j(\tau)}}{\sum_{j=1}^{N} q^{(\alpha - 1)k_j(\tau)}}\right]$$

The expected proportional change at time τ of sector j's contribution to the index is

$$E(\Delta qual_{\tau}^{j}) = \mu\left(\frac{q^{(\alpha-1)(k_{j}(\tau)+1)} - q^{(\alpha-1)k_{j}(\tau)}}{q^{(\alpha-1)k_{j}(\tau)}}\right) = \mu\left(q^{\alpha-1} - 1\right)$$

Using this, we have

$$E\left[\widehat{Q(\tau)}\right] = \sum_{j=1}^{N} \mu \frac{q^{\alpha-1} - 1}{\alpha - 1} E\left[\frac{q^{(\alpha-1)k_j(\tau)}}{\sum_{j=1}^{N} q^{(\alpha-1)k_j(\tau)}}\right] + \frac{1}{\alpha - 1} \sum_{j=1}^{N} Cov\left[q^{\widehat{(\alpha-1)k_j(\tau)}} \frac{q^{(\alpha-1)k_j(\tau)}}{\sum_{j=1}^{N} q^{(\alpha-1)k_j(\tau)}}\right]$$

As was shown in section 2.2.4, the probability of an innovation being made and thus $q^{(\alpha-1)k_j(\tau)}$ is independent of sector *j*'s position relative to the other sectors, $\frac{q^{(\alpha-1)k_j(\tau)}}{\sum_{j=1}^{N} q^{(\alpha-1)k_j(\tau)}}$. Thus, $Cov\left[q^{(\alpha-1)k_j(\tau)} \frac{q^{(\alpha-1)k_j(\tau)}}{\sum_{j=1}^{N} q^{(\alpha-1)k_j(\tau)}}\right] = 0$. Consequently, $E\left[\widehat{Q(\tau)}\right] = \mu \frac{q^{\alpha-1}-1}{\alpha-1}$

The Law of large numbers implies that for a large number N of sectors, the actual growth rate of the quality index, $\gamma_Q(\tau)$, converges to the expected growth rate, $E\left[\widehat{Q(\tau)}\right]$. More precisely, a standard version of the law of large numbers for independent and identically distributed variables $\{Y_t\}$ with $E\left[Y_t\right] = \mu$ and variance $E\left[(Y_t - \mu)^2\right] = \sigma^2$ states that the sample mean $\overline{Y}_T = (1/T) \sum_{t=1}^T Y_t$ converges in probability to the population mean μ . \overline{Y}_T

has expectation μ and variance $E\left[\left(\overline{Y}_T - \mu\right)^2\right] = \sigma^2/T$.⁷⁰ This variance goes to zero as $T \to \infty$ implying $\overline{Y}_T \xrightarrow{p} \mu$. In the present context, the weighting factors $(q^{(\alpha-1)k_j(\tau)}/Q(\tau)^{(\alpha-1)})$ are not constant as in the standard case (1/T). Yet their being small and summing up to unity suffices for the result to carry over to our case. With this, we have equation (2.20) in the main text.

2.A.5 Proofs of propositions 1-4

Proof of proposition 1. Given that we assumed $\eta \geq 1$, the RHS of equation (2.36) increases in μ as depicted in fig. 2.1. Further, the LHS of equation (2.36) is concave in μ for $\psi > 0$, $\frac{\partial^2 LHS}{\partial \mu^2} = -\frac{2\alpha\psi}{(\mu+\beta)^2} \left[\frac{q^{\alpha-1}}{\alpha-1} + \left(\frac{L}{\lambda} - \mu\frac{q^{\alpha-1}}{\alpha-1}\right)\frac{1}{\mu+\beta}\right] < 0$, while it decreases in μ at non-positive money growth rates, $\frac{\partial LHS}{\partial \mu} = \left[-\frac{q^{\alpha-1}}{\alpha-1}(\mu+\beta) + \frac{\alpha\psi}{\mu+\beta-\alpha\psi}\left(\frac{L}{\lambda} - \mu\frac{q^{\alpha-1}}{\alpha-1}\right)\right]\frac{\mu+\beta-\alpha\psi}{(\mu+\beta)^2} < 0$ for $\psi \leq 0$. With conditions (2.37) and (2.38) we make sure that the value for $\mu \to 0$ of the LHS of equation (2.36) is larger than that of the RHS. Further, note that for $\mu \to \infty$, the value of the RHS exceeds that of the LHS: $\lim_{\mu\to\infty} \left(\frac{L}{\lambda} - \mu\frac{q^{\alpha-1}}{\alpha-1}\right) \left(1 - \frac{\alpha\psi}{\mu+\beta}\right) = -\infty < \infty = \lim_{\mu\to\infty} \left(\rho + \tilde{\eta}\mu\right) \left(1 - \frac{\alpha\psi}{\rho+\beta+\tilde{\eta}\mu}\right)$ where $\tilde{\eta} = \left(1 + (\eta + \alpha - 2)\frac{q^{\alpha-1}-1}{\alpha-1}\right)$. Thus the two functions have a unique intersection with $\mu > 0$.

For the sake of completeness, we give a short description of the two additional effects on instantaneous profits of an increase in μ under price rigidity $(\beta < \infty, \psi \neq 0)$: Firstly, an increase in the frequency of innovation reduces price dispersion because any innovative firm replacing the incumbent to produce a new variety of intermediate good j can set a new price. This reduces production inefficiency, raising the average productivity of intermediates in final good production and therefore, demand for intermediate goods and a new intermediate producer's profits. Secondly, at $\psi > 0$ ($\psi < 0$) the increase in μ reduces (increases) the firm's initial relative price since a higher probability of being replaced shifts more weight to present as opposed to future profits in the firm's mark-up. This increases (reduces) demand for the good and profits.

Proof of proposition 2. For $\beta \to \infty$, the zero profit condition (2.36)

 $^{^{70}\}mathrm{See}$ e.g. Hamilton [1994].

reduces to $\frac{L}{\lambda} - \mu \frac{q^{\alpha-1}}{\alpha-1} = \rho + \left(1 + (\eta + \alpha - 2) \frac{q^{\alpha-1}-1}{\alpha-1}\right) \mu$ so that the growth rate of money ψ has no influence on the equilibrium research intensity μ . Since by equation (2.21) $\gamma = \frac{q^{\alpha-1}-1}{\alpha-1}\mu$, the economy's real growth rate γ is independent of ψ , too.

Proof of proposition 3. Consider equation (2.36). To see that $\frac{d\mu}{d\psi} = -\frac{\partial LHS}{\partial \Phi} - \frac{\partial RHS}{\partial \mu} < 0$, first refer to figure 2.1 to see that given assumptions (2.37)-(2.38) and concavity of the LHS-curve, the latter's slope is always smaller than that of the RHS-curve at the equilibrium $(\frac{\partial LHS}{\partial \mu} - \frac{\partial RHS}{\partial \mu} < 0)$. Further, $\frac{\partial LHS}{\partial \psi} - \frac{\partial RHS}{\partial \psi} = -\frac{\alpha(\frac{L}{\lambda} - \mu \frac{q^{\alpha-1}}{\alpha-1})[\rho + (\tilde{\eta} - 1)\mu]}{(\mu+\beta)(\rho+\beta-\alpha\psi+\tilde{\eta}\mu)} < 0$ since $1 < \tilde{\eta} = (1 + (\eta + \alpha - 2) \frac{q^{\alpha-1} - 1}{\alpha-1})$ given $\eta \ge 1$, $\beta > \alpha\psi$ from assumption (2.38) and $L/\lambda - \mu q^{\alpha-1}/(\alpha - 1) > 0$ since final good production is positive. Thus the negative effects of an increase in money growth dominate for the very reasons (assumption (2.38)) that ensure the uniqueness of the steady state equilibrium. Further, since $\gamma = \frac{q^{\alpha-1} - 1}{\alpha-1}\mu$, $\frac{d\gamma}{d\psi} = \frac{q^{\alpha-1} - 1}{\alpha-1}\frac{d\mu}{d\psi} < 0$. **Proof of proposition 4.** $\frac{d\mu}{d\beta} = -\frac{\partial LHS}{\partial D} - \frac{\partial RHS}{\partial D} \ge 0$ for $\psi \ge 0$ since $\frac{\partial LHS}{\partial \beta} - \frac{\partial RHS}{\partial \beta} = \alpha\psi \frac{\rho + \tilde{\eta}\mu}{\rho + \tilde{\eta}\mu + \beta} \frac{[(\rho + \beta + \tilde{\eta}\mu) + (\mu + \beta - \alpha\psi)][\rho + (\tilde{\eta} - 1)\mu]}{(\mu + \beta)(\rho + \tilde{\eta}\mu + \beta)} \ge 0$ for $\psi \ge 0$ is positive as $\beta > \alpha\psi$ and $\tilde{\eta} > 1$ given assumption (2.38).

Chapter 3

Local Stability

3.1 Introduction

In the preceding chapters, the Schumpeterian model with nominal price rigidity was analysed with regard to the comparative static properties of its unique steady state equilibrium. This analysis delivers relevant predictions for the model economy's long run behaviour only if the economy converges to the steady state in the long run, i.e. if the steady state is stable. To underscore the relevance of our approach, we therefore examine the stability of the steady state in this chapter. Given that the dynamical system describing the economy's behaviour under price rigidity and endogenously determined growth off steady state contains a large number of variables, this is a very challenging task. Neither can its global behaviour be analysed nor can we derive analytical conditions for its stability in the neighbourhood of the steady state. We therefore proceed by reporting the local stability properties of a large number of calibrated examples. In order to derive the local stability properties of the model several steps are necessary: In section 3.2, the model's key equations are adjusted to take account of the off-steady state behaviour of the variables. In the process, variables which have positive growth rates in the model's original steady state are transformed by normalising them such that all relevant variables are constant at steady state. As we limit our analysis to the local dynamics, the equations are linearised

around the steady state using a first order Taylor approximation. As discussed in chapter 1, this is the standard approach used in the analysis of New Keynesian Business Cycle or NNS models. Note, however, that in contrast to the NNS approach, our interest here lies not in generating impulse responses to simulated shocks in order to characterise the model economy's behaviour off steady state quantitatively. Our exclusive aim is the qualitative characterisation of the steady state's local stability through the analysis of the dynamical system's eigenvalues. For this purpose, the eight equations describing the economy's behaviour in the neighbourhood of the steady state are reduced to a dynamical system in six variables in the Appendix. Section 3.3 presents the reduced dynamical system and the results of our stability analysis: The eigenvalues associated with the reduced system determine the steady state's local stability. We find that for all parameter constellations examined, the steady state is locally stable, so that our analysis of the steady state's comparative static properties is indeed relevant for the description of the model economy's long-run behaviour. Interestingly, for all parameter constellations the equilibrium is also indeterminate, i.e. for given initial conditions the convergent path to the steady state is not unique.

For the linearisation, we use the following notation: The steady state value of the variable $x(\tau)$ is denoted by dropping the time index, so that $[x(\tau) - x]$ denotes the deviation of the variable from steady state and $[x(\tau) - x]$ is the time derivative of this expression.

3.2 Key equations for the economy's off-steadystate behaviour

3.2.1 Firms

3.2.1.1 Intermediate goods sector: Optimal intermediate goods price

Relaxing the assumptions that the interest rate $i(\tau)$, the intensity of innovation $\mu(\tau)$ and the rates of growth of output, wages and prices, $\gamma(\tau)$, $\omega(\tau)$ and $\pi(\tau)$ are all constant at steady state, the optimal price chosen by an intermediate goods firm allowed to reset its price is given by¹

$$P^{*}(\tau) = \frac{\alpha}{\alpha - 1} \frac{\int\limits_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} [i(\theta) + \mu(\theta) + \beta] d\theta} P(s)^{\alpha} w(s) Y(s) ds}{\int\limits_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} [i(\theta) + \mu(\theta) + \beta] d\theta} P(s)^{\alpha} Y(s) ds}$$
(3.1)

We aim to linearise the model around a steady state where all relevant variables are constant. We therefore define the normalised variables $\widetilde{w}(\tau) = \frac{w(\tau)}{P(\tau)Q(\tau)}$ and $Y(\tau) = \frac{Y(\tau)}{Q(\tau)}$ which are constant at steady state. Using these and dividing both sides of equation (3.1) by $P(\tau) Q(\tau)$ yields

$$\widetilde{P^*}(\tau) = \frac{\alpha}{\alpha - 1} \frac{\int\limits_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} [i(\theta) + \mu(\theta) + \beta] d\theta} \left(\frac{P(s)}{P(\tau)}\right)^{\alpha} \widetilde{w}(s) \frac{P(s)Q(s)}{P(\tau)Q(\tau)} \frac{Y(s)}{Q(s)} \frac{Q(s)}{Q(\tau)} ds}{\int\limits_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} [i(\theta) + \mu(\theta) + \beta] d\theta} \left(\frac{P(s)}{P(\tau)}\right)^{\alpha} \frac{Y(s)}{Q(s)} \frac{Q(s)}{Q(\tau)} ds}$$

Using the fact that $\frac{X(s)}{X(\tau)} = \frac{X(0)e^{\int_{\theta=0}^{s}\gamma_X(\theta)d\theta}}{X(0)e^{\int_{\theta=0}^{\tau}\gamma_X(\theta)d\theta}} = e^{\int_{\theta=\tau}^{s}\gamma_X(\theta)d\theta}$, where $\gamma_X(\theta)$ is the growth rate of X at time θ , this can be rewritten as

$$\widetilde{P^*}(\tau) = \frac{\alpha}{\alpha - 1} \frac{\int\limits_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} \left[i(\theta) + \mu(\theta) - (\alpha + 1)\pi(\theta) + \beta - 2\gamma_Q(\theta)\right] d\theta} \widetilde{w}(s) \frac{Y(s)}{Q(s)} ds}{\int\limits_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} \left[i(\theta) + \mu(\theta) + \beta - \alpha\pi(\theta) - \gamma_Q(\theta)\right] d\theta} \frac{Y(s)}{Q(s)} ds}$$
(3.2)

¹Note that the relevant probability that the firm has not been replaced and has not received a pricing signal by time τ given that the last innovation was at time t_{k_j} , $\widetilde{B}e^{-\int_{\theta=t_{k_j}}^{\tau}(\mu(\theta)+\beta)d\theta}$, is unchanged, except for the fact that $\mu(\tau)$ need not be constant outside steady state. Note that the unchanged razor's edge condition from the research sector continues to hold, so that $\mu(\tau)$ is the same for all sectors j in equilibrium, which we use here.

The equation is linearised in Appendix $3.A.1^2$, yielding

$$\begin{bmatrix} \widetilde{P^*}(\tau) - \widetilde{P^*} \end{bmatrix} = + \left(\frac{\widetilde{P^*}^{1-\alpha}}{L_2} \frac{Y}{Q} - \alpha \psi \right) \begin{bmatrix} \widetilde{P^*}(\tau) - \widetilde{P^*} \end{bmatrix} \\ - \frac{\widetilde{P^*}}{L_2} \psi \begin{bmatrix} L_2(\tau) - L_2 \end{bmatrix} \\ + \widetilde{P^*} \frac{1}{\frac{Y}{Q}} \psi \begin{bmatrix} \frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q} \end{bmatrix} \\ - \frac{\widetilde{P^*}^{1-\alpha}}{L_2} \frac{Y}{Q} \frac{\alpha}{\alpha - 1} \begin{bmatrix} \widetilde{w}(\tau) - \widetilde{w} \end{bmatrix} \\ - \widetilde{P^*} \begin{bmatrix} \pi(\tau) - \pi \end{bmatrix} \\ - \widetilde{P^*} \frac{q^{\alpha-1} - 1}{\alpha - 1} \begin{bmatrix} \mu(\tau) - \mu \end{bmatrix}$$
(3.3)

where $L_2(\tau) = \widetilde{P^*}(\tau)^{1-\alpha} \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^s \left[i(\theta)+\mu(\theta)+\beta-\alpha\pi(\theta)-\gamma_Q(\tau)\right]d\theta} \frac{Y(s)}{Q(s)} ds$ and $\frac{L_2(\tau)}{\widetilde{P^*}(\tau)}$

reflects the development of the firm's revenues.³ For later reference, the linearised version of $L_2(\tau)$ is⁴

$$\begin{bmatrix} L_{2}(\tau) - L_{2} \end{bmatrix} = L_{2}(1-\alpha)\widetilde{P^{*}}^{-1} \left[\widetilde{P^{*}}(\tau) - \widetilde{P^{*}}\right] \\ + L_{2}\left[r(\tau) - r\right] \\ + L_{2}\left(1 - \frac{q^{\alpha-1} - 1}{\alpha - 1}\right)\left[\mu(\tau) - \mu\right] \\ - L_{2}(\alpha - 1)\left[\pi(\tau) - \pi\right] \\ + \left[\beta + h - (\alpha - 1)\psi\right]\left[L_{2}(\tau) - L_{2}\right] \\ - \widetilde{P^{*}}^{1-\alpha}\left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q}\right] \\ - (1-\alpha)\widetilde{P^{*}}^{-\alpha}\frac{Y}{Q}\left[\widetilde{P^{*}}(\tau) - \widetilde{P^{*}}\right] \end{bmatrix}$$

²Note that for this and the following linearisations, the fact that $\gamma_Q(\tau)$ is proportional to $\mu(\tau)$ by equation (3.13) is used to replace the former variable.

³See the denominators in the fractions in equations (3.1) and (3.2).

⁴See the Appendix for details on the linearisation.

3.2.1.2 An intermediate good producer's market value at market entrance

Normalising variables as before and using the shorthand notation $E(V_{\tau}) = E(V_{k_j}(\tau) | t_{k_j} = \tau)$, the market value given by the present value at market entrance in t_{k_j} of all future profits of the firm making an innovation at time τ is

$$\frac{E\left(V_{\tau}\right)}{P\left(\tau\right)Q\left(\tau\right)} = \left(\frac{q^{k_{j}+1}}{Q\left(\tau\right)}\right)^{\alpha-1} \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} \left(\frac{P_{k_{j}}\left(s\right)}{P(s)Q(s)}\right)^{1-\alpha} ds$$
$$-\left(\frac{q^{k_{j}+1}}{Q\left(\tau\right)}\right)^{\alpha-1} \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} \widetilde{w}(s) \left(\frac{P_{k_{j}}\left(s\right)}{P(s)Q(s)}\right)^{-\alpha} ds$$
(3.4)

where $h(\theta) = i(\theta) + \mu(\theta) - \pi(\theta) + (\alpha - 2)\gamma_Q(\theta)$.

Taking account of the future development of the firm's normalised price $P_{k_j}(s) / [P(s)Q(s)]$ by using the probability density for pricing signals, we can rewrite equation (3.4) as a function of the development of the normalised optimal price $\widetilde{P^*}(s)$:⁵

$$\frac{E(V_{\tau})}{A_{1}(\tau)} = \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} \left[\int_{\zeta=\tau}^{s} \beta e^{-\int_{t=\zeta}^{s} [\beta-(\alpha-1)\widetilde{\pi}(t)]dt} \widetilde{P^{*}}(\zeta)^{1-\alpha} d\zeta \right] ds$$

$$+ \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} \left[e^{-\int_{t=\tau}^{s} [\beta-(\alpha-1)\widetilde{\pi}(t)]dt} \widetilde{P^{*}}(\tau)^{1-\alpha} \right] ds$$

$$- \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} \widetilde{w}(s) \left[\int_{\zeta=\tau}^{s} \beta e^{-\int_{t=\zeta}^{s} [\beta-\alpha\widetilde{\pi}(t)]dt} \widetilde{P^{*}}(\zeta)^{-\alpha} d\zeta \right] ds$$

$$- \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} \widetilde{w}(s) \left[e^{-\int_{t=\tau}^{s} [\beta-\alpha\widetilde{\pi}(t)]dt} \widetilde{P^{*}}(\tau)^{-\alpha} \right] ds$$

$$(3.5)$$

⁵As this involves a number of tedious steps we present these calculations in Appendix 3.A.2.

where $A_1(\tau) = \left(\frac{q^{k_j+1}}{Q(\tau)}\right)^{\alpha-1} P(\tau) Q(\tau), \ \widetilde{\pi}(\tau) = \pi(\tau) + \gamma_Q(\tau)$ and h is the constant steady state value of $h(\tau)$.

3.2.1.3 Final good sector: Price level

Taking into account that $\mu(\tau)$ is not constant outside steady state, the price level (2.4) can analogously to the procedure in Appendix 2.A.3 be rewritten as

$$P(\tau)^{1-\alpha} = Q(\tau)^{(\alpha-1)} \int_{-\infty}^{\tau} \left[\mu(s) + \beta \right] e^{-\int_{\theta=s}^{\tau} [\mu(\theta) + \beta] d\theta} P^*(s)^{1-\alpha} ds$$

In this, we follow the literature⁶ in making the assumption that the price level can in and near steady state be rewritten as a weighted average of past optimal prices, which here implies that the structure of past prices for a given quality level q^k is approximately the same as the structure for all quality levels.⁷

We rewrite the equation in terms of the normalised optimal price $\widetilde{P^*}(s) = P^*(s) / [P(s)Q(s)]$ which is constant at steady state:

$$1 = \int_{-\infty}^{\tau} \left[\mu\left(s\right) + \beta \right] e^{-\int_{\theta=s}^{\tau} \left[\mu(\theta) + \beta \right] d\theta} \left[\frac{P^*(s)}{P(s)Q(s)} \right]^{1-\alpha} \left[\frac{P(s)Q(s)}{P(\tau)Q(\tau)} \right]^{1-\alpha} ds$$

which can be rewritten as

$$1 = \int_{-\infty}^{\tau} \left[\mu\left(s\right) + \beta \right] e^{-\int_{\theta=s}^{\tau} \left[\mu(\theta) + \beta - (\alpha - 1) \left(\pi(\theta) + \gamma_Q(\theta) \right) \right] d\theta} \widetilde{P^*}(s)^{1-\alpha} ds \qquad (3.6)$$

where $\gamma_Q(\tau)$ is the off-steady-state growth rate of the quality index Q at time τ .⁸

⁸Note that off the steady state, $\gamma(\tau) = \gamma_Q(\tau)$ need not hold.

⁶See Benhabib, Schmitt-Grohé and Uribe [2001a,2001b], Leith and Wren-Lewis [2000] and Wolman [1999]).

⁷While this is indeed a simplification for a general local stability analysis, it is a natural assumption for the analysis of local stability in the following situation: Think of an economy that has been in steady state – where the incentive to innovate and the past distribution of prices are indeed independent of a sector's quality rung – for a long time and is then moved into the neighbourhood of the steady state by a slight perturbation.

Differentiating equation (3.6) with respect to time results in

$$0 = \left[\mu\left(\tau\right) + \beta\right]\widetilde{P^{*}}(\tau)^{1-\alpha} - \left[\mu\left(\tau\right) + \beta - \left(\alpha - 1\right)\left(\pi\left(\tau\right) + \gamma_{Q}\left(\tau\right)\right)\right]$$

Linearising this equation around the steady state yields

$$\left[\widetilde{P^*}(\tau) - \widetilde{P^*}\right] = \frac{\widetilde{P^*}}{\mu + \beta - (\alpha - 1)\psi} \left[\pi(\tau) - \pi\right] + \frac{\widetilde{P^*}}{\mu + \beta - (\alpha - 1)\psi} \left(\frac{q^{\alpha - 1} - 1}{\alpha - 1} - \frac{\psi}{\mu + \beta}\right) \left[\mu(\tau) - \mu\right] \quad (3.7)$$

3.2.1.4 The R&D zero profit condition

Inserting $\phi(k_j(\tau))$ from equation (2.18) and a new intermediate firm's expected market value (3.5) into the zero profit equation in the research sector (2.17) and linearising, we have⁹

$$\begin{split} [\widetilde{w}(\tau) - \widetilde{w}] &= \left[h + (\alpha - 1)\frac{1}{\lambda}\frac{Y}{Q}\widetilde{P^{*}}^{-\alpha}\right][\widetilde{w}(\tau) - \widetilde{w}] \\ &+ (\alpha - 1)\frac{1}{\lambda}\frac{Y}{Q}\widetilde{P^{*}}^{-\alpha}\left[\alpha\widetilde{P^{*}}^{-1}\left(\widetilde{P^{*}} - \widetilde{w}\right) - 1\right]\left[\widetilde{P^{*}}(\tau) - \widetilde{P^{*}}\right] \\ &+ \widetilde{w}\left[1 + (\alpha - 2)\frac{q^{\alpha - 1} - 1}{\alpha - 1}\right][\mu(\tau) - \mu] \\ &- (\alpha - 1)\frac{1}{\lambda}\widetilde{P^{*}}^{-\alpha}\left(\widetilde{P^{*}} - \widetilde{w}\right)\left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q}\right] \\ &+ \widetilde{w}\left[r(\tau) - r\right] \end{split}$$
(3.8)

where $h = r + \mu + (\alpha - 2) \gamma$ is the firm's steady state flex-price discount rate.

3.2.2 Labour market equilibrium

Equilibrium in the labour market requires that the sum of the labour demands of the intermediate goods sector, $L^{X}(\tau)$, and of the research sector,

⁹For details on the linearisation see Appendix 3.A.3.

 $L^{R}(\tau)$, equal labour supply L. Demand for research labour is given by

$$L^{R}(\tau) = \lambda \frac{q^{\alpha-1}}{\alpha-1} \mu(\tau)$$
(3.9)

The intermediate sector's labour demand is

$$L^{X}(\tau) = \frac{Y(\tau)}{Q(\tau)} \int_{-\infty}^{\tau} \left[\mu(s) + \beta\right] e^{-\int_{\theta=s}^{\tau} \left[\mu(\theta) + \beta\right] d\theta} \widetilde{P^{*}}(s)^{-\alpha} \left[\frac{P(s)Q(s)}{P(\tau)Q(\tau)}\right]^{-\alpha} ds$$
(3.10)

Using the labour demands (3.9) and (3.10), the labour market equilibrium equation is

$$L-\mu(\tau)\lambda\frac{q^{\alpha-1}}{\alpha-1} = \frac{Y(\tau)}{Q(\tau)}\int_{-\infty}^{\tau} \left[\mu(s)+\beta\right]e^{-\int_{\theta=s}^{\tau}\left[\mu(\theta)-\alpha\pi(\theta)-\alpha\gamma_Q(\theta)+\beta\right]d\theta}\widetilde{P^*}(s)^{-\alpha}ds$$
(3.11)

It is shown in Appendix 3.A.3.1 that linearising this equation gives

$$[\mu(\tau) - \mu] = -(\lambda \widetilde{q})^{-1} \frac{\widehat{\beta}}{\widetilde{\mu}} \widetilde{P^{*}}^{-\alpha} \left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q} \right]$$

$$-\alpha (\lambda \widetilde{q})^{-1} \frac{\widehat{\beta}}{\widetilde{\mu}} \frac{Y}{Q} \widetilde{P^{*}}^{-\alpha} [\pi(\tau) - \pi]$$

$$- \left\{ \widetilde{\mu} + (\lambda \widetilde{q})^{-1} \alpha \frac{\widehat{\beta}}{\widetilde{\mu}} \frac{Y}{Q} \widetilde{P^{*}}^{-\alpha} \left(\widehat{q} - \widehat{\psi} \right) \right\} [\mu(\tau) - \mu]$$

$$- (\lambda \widetilde{q})^{-1} \widehat{\beta} \widetilde{P^{*}}^{-\alpha} \left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q} \right]$$

$$+ \alpha (\lambda \widetilde{q})^{-1} \frac{Y}{Q} \widehat{\beta} \widetilde{P^{*}}^{-\alpha-1} \left[\widetilde{P^{*}}(\tau) - \widetilde{P^{*}} \right]$$

$$(3.12)$$

where we introduce the shorthand notation $\hat{q} = \frac{q^{\alpha-1}-1}{\alpha-1}, \ \tilde{q} = \frac{q^{\alpha-1}}{\alpha-1}, \ \tilde{\mu} = (\beta + \mu - \alpha\psi), \ \hat{\psi} = \frac{\psi}{\mu+\beta}, \ \hat{\beta} = \mu + \beta.$

3.2.2.1 The aggregate quality index $Q(\tau)$

Taking into account that $\mu(\tau)$ is not constant outside steady state, the equation relating the growth rate of Q to the research intensity μ is:

$$\gamma_Q(\tau) = \frac{q^{\alpha-1} - 1}{\alpha - 1} \mu(\tau) \tag{3.13}$$

In deriving this, we make the simplifying assumption that the Law of Large Numbers¹⁰ holds approximately in the neighbourhood of the steady state, implying that for a large number N of sectors, the actual growth rate of the quality index, γ_Q , converges to the expected growth rate. We use equation (3.13) to replace γ_Q in the other equilibrium relationships.

3.2.3 Households

Both consumption $c(\tau)$ and real money holdings $m(\tau)$ grow at rate γ at steady state. We normalise by dividing through the quality index $Q(\tau)$, such that we have the new variables $\frac{c(\tau)}{Q(\tau)}$ and $\frac{m(\tau)}{Q(\tau)}$ which are constant at steady state. The household's off-steady-state static optimality condition then reads:

$$\frac{\theta}{1-\theta}\frac{\frac{c(\tau)}{Q(\tau)}}{\frac{m(\tau)}{Q(\tau)}} = r\left(\tau\right) + \pi\left(\tau\right)$$
(3.14)

Linearly approximating both sides yields

$$\frac{\theta}{1-\theta} \left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q} \right] = (r+\pi) \left[\frac{m(\tau)}{Q(\tau)} - \frac{m}{Q} \right] + \frac{m}{Q} [r(\tau) - r] + \frac{m}{Q} [\pi(\tau) - \pi]$$
(3.15)

where equilibrium in the output market was used to replace c/Q with equilibrium in the output market Y/Q. Noting that $\frac{c(\tau)}{c(\tau)} - \gamma_Q(\tau) = \left(\frac{c(\tau)}{Q(\tau)}\right) \left(\frac{c(\tau)}{Q(\tau)}\right)^{-1}$

 $^{^{10}}$ See Appendix 2.A.4 in chapter 2.

and $\frac{m(\tau)}{m(\tau)} - \gamma_Q(\tau) = \left(\frac{m(\tau)}{Q(\tau)}\right) \left(\frac{m(\tau)}{Q(\tau)}\right)^{-1}$, the household's dynamic efficiency condition (2.30) can be rewritten in terms of the normalised variables c/Q and m/Q. Linearising both sides, we have

$$\begin{bmatrix} \underline{Y(\tau)} & -\underline{Y} \\ \overline{Q(\tau)} & -\underline{Y} \\ \end{bmatrix} = \frac{\theta(1-\eta)}{\eta+\theta(1-\eta)} \frac{\underline{Y}}{\underline{Q}} \begin{bmatrix} \underline{m(\tau)} \\ \overline{Q(\tau)} & -\underline{m} \\ \end{bmatrix} + \frac{\underline{Y}}{\eta+\theta(1-\eta)} \begin{bmatrix} r(\tau) & -r \end{bmatrix} - \frac{\eta}{\eta+\theta(1-\eta)} \frac{Y}{Q} \widehat{q} \begin{bmatrix} \mu(\tau) & -\mu \end{bmatrix}$$
(3.16)

where Y/Q was again substituted for c/Q.

3.2.4 Money market equilibrium

Normalising money demand as $\frac{m(\tau)}{Q(\tau)}$, equilibrium in the money market is characterised by

$$\frac{\left(\frac{m(\tau)}{Q(\tau)}\right)}{\frac{m(\tau)}{Q(\tau)}} = \psi - \pi(\tau) - \gamma_Q(\tau)$$
(3.17)

Linearising this equation yields

$$\left[\frac{m\left(\tau\right)}{Q\left(\tau\right)} - \frac{m}{Q}\right] = -\frac{m}{Q}\left[\pi\left(\tau\right) - \pi\right] - \frac{m}{Q}\widehat{q}\left[\mu\left(\tau\right) - \mu\right]$$
(3.18)

3.3 The linearised dynamical system and local stability of the steady state equilibrium

The eight equations describing the linearised off-steady state behaviour of the optimal price (3.3), and the revenue variable L_2 (3.22), the price index (3.7), the research intensity (3.8), the labour market (3.12), the household's behaviour (3.16) and (3.15) and the money market (3.18), respectively, form a system in the eight variables $\frac{Y(\tau)}{Q(\tau)}, \frac{m(\tau)}{Q(\tau)}, L_2(\tau), \mu(\tau), \tilde{P}^*(\tau), \tilde{w}(\tau), r(\tau)$ and

 $\pi(\tau)$.¹¹ Given that the two equations (3.7) and (3.15) are static, however, we can use them to eliminate $\frac{m(\tau)}{Q(\tau)}$ and $\tilde{P}^*(\tau)$ from the system, resulting in a dynamical system in six variables. To achieve this and in order to write the system in the form $[\mathbf{x}-\overline{\mathbf{x}}] = f[\mathbf{x}-\overline{\mathbf{x}}]$, where \mathbf{x} is the vector of variables contained in the system and $\overline{\mathbf{x}}$ is the steady state value of \mathbf{x} , some cumbersome manipulations are necessary which can be found in Appendix 3.A.6. Resulting from these is the following system describing the dynamical behaviour of the model economy:

$$\begin{bmatrix} \underline{Y(\tau)} & -\underline{Y} \\ Q(\tau) & -\underline{Y} \\ r(\tau) & -r \\ \mu(\tau) & -\mu \\ \vdots \\ \pi(\tau) & -\pi \\ \vdots \\ L_2(\tau) & -L_2 \end{bmatrix} = \mathbf{M} \begin{bmatrix} \underline{Y(\tau)} & -\underline{Y} \\ r(\tau) & -\underline{Y} \\ r(\tau) & -\mu \\ \mu(\tau) & -\mu \\ \pi(\tau) & -\pi \\ \widetilde{w}(\tau) & -\widetilde{w} \\ L_2(\tau) & -L_2 \end{bmatrix}$$

where the elements of the **6x6** matrix **M** can be found in Appendix 3.A.7. The Hartman-Grobman theorem states that the local properties of the linearised dynamical system also hold for the non-linear system if the matrix **M** has no root with zero real part.¹² In this case, the steady state $\left(\frac{Y}{Q}, r, \mu, \pi, \tilde{w}, L_2\right)$ is called a non-degenerate fixed point. Due to the system's complexity, it is not possible to describe the linearised dynamical system's stability properties analytically. We therefore investigate local stability in a number of calibrated examples. In all examined examples, the steady state was non-degenerate so that the linearised system's local properties carry over to the non-linearised system.

Parameters for the examples were chosen from the ranges indicated in

¹¹Note that since the household's current wealth does not influence the equilibrium allocation, it is not necessary to keep track here of the evolution of the household's wealth (2.28) which can be determined separately. The transversality condition $\lim_{\tau\to\infty} v(\tau) e^{-\int_{\theta=0}^{\tau} r(\theta)d\theta} = 0$ derived from (2.28) holds for any stable steady state equilibrium since at steady state, $v(\tau)$ grows at rate γ and $r - \gamma = \rho + (\eta - 1)\gamma > 0$ given our assumption $\eta \geq 1$.

 $^{^{12}}$ See Gandolfo [1996], p. 362.

table 3.1:

Parameter	Range	Parameter	Range
α	6 - 11	η	1 - 2
β	1 - 2.5	ρ	0.01 - 0.05
<i>q</i>	1.01 - 1.5	θ	0.01 - 0.5
ψ	-0.015 - 0.15		

Table 3.1: Parameter values used in local stability analysis

The eigenvalues of the matrix \mathbf{M} were computed for a total of 113 parameter constellations.¹³ Of the model's variables, only one is predetermined: Real money supply $m(\tau)/Q(\tau)$. Therefore for the steady state to be locally stable, at least one of the eigenvalues needs to have a negative real part. In order for the system to display saddle-path-stable behaviour, i.e. for the convergent solution to be unique, there needs to be exactly one eigenvalue with a negative real part.¹⁴ In all examples where the eigenvalues were computed, there were two eigenvalues with negative real parts, implying that the steady state is locally stable but the equilibrium is indeterminate, i.e. the convergent solution is not unique.

The existence of many equilibrium paths for the model's variables is an interesting feature since multiple solutions can imply the existence of self-fulfilling expectations which allow for endogenous fluctuations.¹⁵

Unlike in the two-dimensional case it is not possible to classify the dynamic behaviour of the system according to the sign and real or complex

¹³Note that it can be shown that in the case without price rigidity $(\beta \to \infty)$, the dynamical system can be reduced to two dimensions. Since prices are flexible, the variables $m(\tau)/Q(\tau)$ and $Y(\tau)/Q(\tau)$ are both non-predetermined such that the steady state under flexibility is locally stable for any choice of parameters.

While we have not examined other parameter constellations, for logarithmic utility $(\eta = 1)$ it can be shown analytically that the steady state is determinate, i.e. saddle-path stable (for systems that contain only jump variables, the constellation with all eigenvalues being positive is the equivalent of saddle path stability, although the stable arm is of dimension zero. See Benhabib, Schmitt-Grohé and Uribe [2001]).

¹⁴Stability conditions for dynamic systems are presented in Gandolfo [1996]. See also Buiter [1984].

¹⁵For a careful survey on the causes and consequences of indeterminacy in monetary and more general macroeconomic models, see Benhabib and Farmer [1999].
character of the eigenvalues.¹⁶ Even so, it is interesting to note that the number of pairs of complex eigenvalues increases with the money growth rate.¹⁷ This might imply that the complexity of the model's off-steady state behaviour increases with the size of the distortions caused by price rigidity.¹⁸

By way of conclusion, irrespective of these considerations, the main result that the steady state is locally stable assures us of the relevance of our analysis' focus on the steady state properties of the Schumpeterian growth model with price rigidity.

Appendix to chapter 3 **3.A**

3.A.1 Linearising the optimal price

We define new variables $H(\tau) = \widetilde{P^*}(\tau) \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} \left[i(\theta)+\mu(\theta)+\beta-\alpha\pi(\theta)-\gamma_Q(\tau)\right] d\theta} \frac{Y(s)}{Q(s)} ds$

and

 $K(\tau) = \frac{\alpha}{\alpha - 1} \int_{\theta = \tau}^{\infty} e^{-\int_{\theta = \tau}^{s} \left[i(\theta) + \mu(\theta) - (\alpha + 1)\pi(\theta) + \beta - 2\gamma_Q(\tau)\right] d\theta} \widetilde{w}(s) \frac{Y(s)}{Q(s)} ds.$ Equation (3.2)

that can now be rewritten as

$$H\left(\tau\right) = K\left(\tau\right)$$

In order to derive the linearised version of this equation that depends only on the relevant variables time τ values, we first take the derivative with respect to time τ of both sides of the equation separately and then linearise the

¹⁶See Gandolfo [1996], pp. 346-358 for the two-dimensional case.

¹⁷In 21 out of 22 parameter constellations with $\psi = 0$ all the system's eigenvalues were real. The exception ocurred in a case when the intertemporal elasticity of substitution $1/\eta$ took on its maximum value one. Here, there was one pair of complex eigenvalues. In contrast, at the maximum money growth rate investigated for each of the parameter constellations, the negative real parts of eigenvalues were indeed in all cases associated with complex roots.

 $^{^{18}}$ This observation is limited to the effect of large *positive* money growth rates since the zero bound on nominal interest rate prevents us from analysing large negative money growth rates.

resulting equation in a second step.¹⁹

Taking the time derivative of $H(\tau)$ yields

$$\begin{split} H(\tau) &= \frac{1}{\widetilde{P^*}(\tau)} \widetilde{P^*}(\tau) \widetilde{P^*}(\tau) \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} \left[i(\theta) + \mu(\theta) + \beta - \alpha \pi(\theta) - \gamma_Q(\theta)\right] d\theta} \frac{Y(s)}{Q(s)} ds \\ &- \widetilde{P^*}(\tau) \frac{Y(\tau)}{Q(\tau)} \\ &+ \iota(\tau) \widetilde{P^*}(\tau) \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} \left[i(\theta) + \mu(\theta) + \beta - \alpha \pi(\theta) - \gamma_Q(\tau)\right] d\theta} \frac{Y(s)}{Q(s)} ds \end{split}$$

where $\iota(\tau) = [i(\tau) + \mu(\tau) + \beta - \alpha \pi(\tau) - \gamma_Q(\tau)]$. Taking account of the definition of $H(\tau)$, this reduces to

$$\stackrel{\cdot}{H(\tau)} = \left[\frac{1}{\widetilde{P^*}(\tau)} \stackrel{\cdot}{\widetilde{P^*}(\tau)} + \left[i(\tau) + \mu(\tau) + \beta - \alpha \pi(\tau) - \gamma_Q(\tau) \right] \right] H(\tau)$$
$$- \widetilde{P^*}(\tau) \frac{Y(\tau)}{Q(\tau)}$$

Similarly,

$$\begin{split} \dot{K}(\tau) &= -\frac{\alpha}{\alpha - 1} \widetilde{w}(\tau) \frac{Y(\tau)}{Q(\tau)} \\ &+ \left[i(\tau) + \mu(\tau) - (\alpha + 1) \pi(\tau) + \beta - 2\gamma_Q(\tau) \right] K(\tau) \end{split}$$

With $H(\tau) = K(\tau)$ we have

$$\begin{bmatrix} \frac{1}{\widetilde{P^*}(\tau)} \widetilde{P^*}(\tau) + \left[i(\tau) + \mu(\tau) + \beta - \alpha\pi(\tau) - \gamma_Q(\tau)\right] \end{bmatrix} H(\tau) - \widetilde{P^*}(\tau) \frac{Y(\tau)}{Q(\tau)}$$
$$= -\frac{\alpha}{\alpha - 1} \widetilde{w}(\tau) \frac{Y(\tau)}{Q(\tau)} + \left[i(\tau) + \mu(\tau) - (\alpha + 1)\pi(\tau) + \beta - 2\gamma_Q(\tau)\right] K(\tau)$$

¹⁹Note that the result is unchanged by the sequence of steps taken, in particular we could first linearise the equation and then take its time derivative.

Chapter 3. Local Stability

.

Using $H(\tau) = K(\tau)$ and replacing $\gamma_Q(\tau)$ by $\frac{q^{\alpha-1}-1}{\alpha-1}\mu(\tau)$, this reduces to

$$\frac{\widetilde{P^{*}}\left(\tau\right)}{\widetilde{P^{*}}\left(\tau\right)} = \frac{1}{H\left(\tau\right)} \left[\widetilde{P^{*}}\left(\tau\right) - \frac{\alpha}{\alpha - 1}\widetilde{w}\left(\tau\right)\right] \frac{Y(\tau)}{Q\left(\tau\right)} - \left[\pi\left(\tau\right) + \frac{q^{\alpha - 1} - 1}{\alpha - 1}\mu\left(\tau\right)\right]$$

We introduce an additional variable $L_2(\tau)$ with $L_2(\tau) \widetilde{P^*}(\tau)^{\alpha} = H(\tau)$ which allows us to eliminate $H(\tau)$:²⁰

$$\frac{\widetilde{P^{*}}(\tau)}{\widetilde{P^{*}}(\tau)} = \frac{\widetilde{P^{*}}(\tau)^{-\alpha}}{L_{2}(\tau)} \left[\widetilde{P^{*}}(\tau) - \frac{\alpha}{\alpha - 1} \widetilde{w}(\tau) \right] \frac{Y(\tau)}{Q(\tau)} - \left[\pi(\tau) + \frac{q^{\alpha - 1} - 1}{\alpha - 1} \mu(\tau) \right]$$
(3.19)

Linearising, we have

$$\begin{split} \left[\widetilde{P^*}\left(\tau\right) - \widetilde{P^*}\right] &= \left\{ \frac{\widetilde{P^*}^{1-\alpha}}{L_2} \frac{Y}{Q} - \alpha \frac{\widetilde{P^*}^{-\alpha}}{L_2} \frac{Y}{Q} \left(\widetilde{P^*} - \frac{\alpha}{\alpha - 1} \widetilde{w}\right) \right\} \left[\widetilde{P^*}\left(\tau\right) - \widetilde{P^*}\right] \\ &- \frac{\widetilde{P^*}^{1-\alpha}}{(L_2)^2} \frac{Y}{Q} \left(\widetilde{P^*} - \frac{\alpha}{\alpha - 1} \widetilde{w}\right) \left[L_2\left(\tau\right) - L_2\right] \\ &+ \frac{\widetilde{P^*}^{1-\alpha}}{L_2} \left(\widetilde{P^*} - \frac{\alpha}{\alpha - 1} \widetilde{w}\right) \left[\frac{Y(\tau)}{Q\left(\tau\right)} - \frac{Y}{Q}\right] \\ &- \frac{\widetilde{P^*}^{1-\alpha}}{L_2} \frac{Y}{Q} \frac{\alpha}{\alpha - 1} \left[\widetilde{w}\left(\tau\right) - \widetilde{w}\right] \\ &- \widetilde{P^*} \left[\pi\left(\tau\right) - \pi\right] \\ &- \widetilde{P^*} \frac{q^{\alpha-1} - 1}{\alpha - 1} \left[\mu\left(\tau\right) - \mu\right] \end{split}$$

Using the fact that at steady state, $0 = \frac{\widetilde{P^*}}{\widetilde{P^*}} = \frac{\widetilde{P^*}^{-\alpha}}{L_2} \frac{Y}{Q} \left(\widetilde{P^*} - \frac{\alpha}{\alpha-1} \widetilde{w} \right) - \psi$ to simplify the coefficients in the first three lines, we have equation (3.3) in the text.

²⁰For details on $L_2(\tau)$ refer to section 3.2.1.1.

Chapter 3. Local Stability

3.A.2 A new intermediate good firm's market value

Defining $h(\theta) = r(\theta) + \mu(\theta) + (\alpha - 2) \gamma_Q(\theta)$ and $A_1(\tau) = \left(\frac{q^{k_j}}{Q(\tau)}\right)^{\alpha - 1} P(\tau) Q(\tau)$, equation (3.4) from the text can be rewritten as

$$E(V_{\tau}) = A_{1}(\tau) \begin{cases} \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} \left(P(s)Q(s)\right)^{\alpha-1} P_{k_{j}}(s)^{1-\alpha} ds \\ \int_{\infty}^{s=\tau} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} \widetilde{w}(s) \left(P(s)Q(s)\right)^{\alpha} P_{k_{j}}(s)^{-\alpha} ds \end{cases}$$

Like at steady state, the probability of having last reset one's price due to a pricing signal at time $\zeta \in (\tau, s)$ is $\beta e^{-\beta(s-\zeta)}$. Thus, we have

$$E\left(V_{\tau}\right) = A_{1}\left(\tau\right) \begin{cases} \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{\frac{Y(s)}{Q(s)}}{(P(s)Q(s))^{1-\alpha}} \left[\int_{r=\tau}^{s} \beta e^{-\beta(s-\zeta)}P^{*}\left(\zeta\right)^{1-\alpha}d\zeta\right] ds \\ + \int_{\infty}^{s=\tau} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{\frac{Y(s)}{Q(s)}}{(P(s)Q(s))^{1-\alpha}} \left(1 - \int_{\zeta=\tau}^{s} \beta e^{-\beta(s-\zeta)}d\zeta\right)P^{*}\left(\tau\right)^{1-\alpha}ds \\ - \int_{s=\tau}^{s=\tau} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{\frac{Y(s)}{Q(s)}\widetilde{w}(s)}{(P(s)Q(s))^{-\alpha}} \left[\int_{\zeta=\tau}^{s} \beta e^{-\beta(s-\zeta)}P^{*}\left(\zeta\right)^{-\alpha}d\zeta\right] ds \\ - \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{\frac{Y(s)}{Q(s)}\widetilde{w}(s)}{(P(s)Q(s))^{-\alpha}} \left(1 - \int_{\zeta=\tau}^{s} \beta e^{-\beta(s-\zeta)}d\zeta\right)P^{*}\left(\tau\right)^{-\alpha}ds \end{cases}$$

Rewriting the equation in terms of the normalised optimal price

$$\begin{split} \widetilde{P^*}(\tau) &= P^*\left(\tau\right) / \left[P(\tau)Q\left(\tau\right)\right] \text{ we have} \\ & \left\{ \begin{array}{l} \int\limits_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} \left[\int\limits_{\zeta=\tau}^{s} \beta e^{-\beta(s-\zeta)} \widetilde{P^*}\left(\zeta\right)^{1-\alpha} \left(\frac{P(s)Q(s)}{P(\zeta)Q(\zeta)}\right)^{\alpha-1} d\zeta \right] ds \\ & +\int\limits_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} e^{-\beta(s-\tau)} \widetilde{P^*}\left(\tau\right)^{1-\alpha} \left(\frac{P(s)Q(s)}{P(\tau)Q(\tau)}\right)^{\alpha-1} ds \\ & -\int\limits_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} \widetilde{w}(s) \left[\int\limits_{\zeta=\tau}^{s} \beta e^{-\beta(s-\zeta)} \widetilde{P^*}\left(\zeta\right)^{-\alpha} \left(\frac{P(s)Q(s)}{P(\zeta)Q(\zeta)}\right)^{\alpha} d\zeta \right] ds \\ & -\int\limits_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} \widetilde{w}(s) \left[e^{-\beta(s-\tau)} \widetilde{P^*}\left(\tau\right)^{-\alpha} \left(\frac{P(s)Q(s)}{P(\tau)Q(\tau)}\right)^{\alpha}\right] ds \end{split}$$

which can be rearranged to yield equation (3.5) in the text. For further use in this appendix, we define new variables $L_1(\tau)$, $L_3(\tau)$ and $L_4(\tau)$ which refer to lines one, three and four of the term in curly brackets in $E(V_{\tau})$ in equation (3.5):

$$L_{1}(\tau) = \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} \left[\int_{\zeta=\tau}^{s} \beta e^{-\int_{t=\zeta}^{s} [\beta-(\alpha-1)\widetilde{\pi}(t)]dt} \widetilde{P^{*}}(\zeta)^{1-\alpha} d\zeta \right] ds,$$

$$L_{3}(\tau) = -\int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} \widetilde{w}(s) \left[\int_{\zeta=\tau}^{s} \beta e^{-\int_{t=\zeta}^{s} [\beta-\alpha\widetilde{\pi}(t)]dt} \widetilde{P^{*}}(\zeta)^{-\alpha} d\zeta \right] ds \text{ and}$$

$$L_{4}(\tau) = -\int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} \widetilde{w}(s) \left[e^{-\int_{t=\tau}^{s} [\beta-\alpha\widetilde{\pi}(t)]dt} \widetilde{P^{*}}(\tau)^{-\alpha} \right] ds.$$

3.A.3 Deriving the linearised zero-profit-condition

Inserting the market value (3.5) using the variables L_1 - L_4 defined in Appendix 3.A.2 and section 3.2.1.1 and $\phi(k_j(\tau))$ from equation (2.18) into the zero profit equation in the research sector (2.17) and rearranging yields

$$\left[(\alpha - 1) \frac{1}{\lambda} \right] \left[L_1(\tau) + L_2(\tau) + L_3(\tau) + L_4(\tau) \right] = \widetilde{w}(\tau)$$
 (3.20)

The derivative of $L_{1}(\tau)$ with respect to time τ is

$$L_{1}(\tau) = h(\tau) \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} h(\theta)d\theta} \frac{Y(s)}{Q(s)} \left[\int_{\zeta=\tau}^{s} \beta e^{-\int_{t=\zeta}^{s} [\beta-(\alpha-1)\widetilde{\pi}(t)]dt} \widetilde{P^{*}}(\zeta)^{1-\alpha} d\zeta \right] ds$$
$$-\widetilde{P^{*}}(\tau)^{1-\alpha} \int_{s=\tau}^{\infty} \frac{Y(s)}{Q(s)} \beta e^{-\int_{\theta=\tau}^{s} [\beta+h(\theta)-(\alpha-1)\widetilde{\pi}(\theta)]d\theta} ds$$

or, using the definitions of $L_{1}(\tau)$ and $L_{2}(\tau)$,

$$\dot{L}_{1}(\tau) = h(\tau) L_{1}(\tau) - \beta L_{2}(\tau)$$

Analogously, the derivative of \sum_{α}^{∞}

$$L_2(\tau) = \widetilde{P^*}(\tau)^{1-\alpha} \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^s \left[i(\theta) + \mu(\theta) + \beta - \alpha \pi(\theta) - \gamma_Q(\tau)\right] d\theta} \frac{Y(s)}{Q(s)} ds \text{ with respect to time is}$$

time is

$$L_{2}(\tau) = (1-\alpha) \frac{\widetilde{P^{*}}(\tau)}{\widetilde{P^{*}}(\tau)} \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} [\beta+h(\theta)-(\alpha-1)\widetilde{\pi}(\theta)]d\theta} \frac{Y(s)}{Q(s)} \widetilde{P^{*}}(\tau)^{1-\alpha} ds$$

+ $[\beta+h(\tau)-(\alpha-1)\widetilde{\pi}(\tau)] \int_{s=\tau}^{\infty} e^{-\int_{\theta=\tau}^{s} [\beta+h(\theta)-(\alpha-1)\widetilde{\pi}(\theta)]d\theta} \widetilde{P^{*}}(\tau)^{1-\alpha} \frac{Y(s)}{Q(s)} ds$
 $-\widetilde{P^{*}}(\tau)^{1-\alpha} \frac{Y(\tau)}{Q(\tau)}$

which can be simplified to

$$L_{2}(\tau) = \left\{ (1-\alpha) \frac{\widetilde{P^{*}}(\tau)}{\widetilde{P^{*}}(\tau)} + [\beta + h(\tau) - (\alpha - 1)\widetilde{\pi}(\tau)] \right\} L_{2}(\tau) - \widetilde{P^{*}}(\tau)^{1-\alpha} \frac{Y(\tau)}{Q(\tau)}$$

$$(3.21)$$

Linearising this equation yields

$$\begin{bmatrix} L_{2}(\tau) - L_{2} \end{bmatrix} = L_{2}(1-\alpha)\widetilde{P^{*}}^{-1} \left[\widetilde{P^{*}}(\tau) - \widetilde{P^{*}}\right] + L_{2}[r(\tau) - r] + L_{2}\left(1 - \frac{q^{\alpha-1} - 1}{\alpha - 1}\right)[\mu(\tau) - \mu] - L_{2}(\alpha - 1)[\pi(\tau) - \pi] + [\beta + h - (\alpha - 1)\psi][L_{2}(\tau) - L_{2}] - \widetilde{P^{*}}^{1-\alpha} \left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q}\right] - (1-\alpha)\widetilde{P^{*}}^{-\alpha}\frac{Y}{Q}\left[\widetilde{P^{*}}(\tau) - \widetilde{P^{*}}\right]$$
(3.22)

where $h = r + \mu + (\alpha - 2) \gamma$.

Using the definitions of $L_{3}(\tau)$ and $L_{4}(\tau)$, the time derivative of $L_{3}(\tau)$ is

$$L_{3}(\tau) = h(\tau) L_{3}(\tau) - \beta L_{4}(\tau)$$

Lastly,

$$L_{4}(\tau) = \frac{Y(\tau)}{Q(\tau)} \widetilde{w}(\tau) \widetilde{P^{*}}(\tau)^{-\alpha} + \left[(-\alpha) \frac{\widetilde{P^{*}}(\tau)}{\widetilde{P^{*}}(\tau)} + h(\tau) + \beta - \alpha \widetilde{\pi}(\tau) \right] L_{4}(\tau)$$

can be derived analogously.

Taking into account that $L_1(\tau) + L_2(\tau) + L_3(\tau) + L_4(\tau) = \left[(\alpha - 1) \frac{1}{\lambda} \right]^{-1} \widetilde{w}(\tau)$

we have

$$\left[(\alpha - 1) \frac{1}{\lambda} \right]^{-1} \widetilde{w} (\tau) = h(\tau) \left[L_1(\tau) + L_2(\tau) + L_3(\tau) + L_4(\tau) \right]$$
$$- (\alpha - 1) \frac{\widetilde{P^*}(\tau)}{\widetilde{P^*}(\tau)} \left[L_2(\tau) + \frac{\alpha}{\alpha - 1} L_4(\tau) \right]$$
$$- (\alpha - 1) \widetilde{\pi} (\tau) \left[L_2(\tau) + \frac{\alpha}{\alpha - 1} L_4(\tau) \right]$$
$$- \frac{Y(\tau)}{Q(\tau)} \widetilde{P^*} (\tau)^{-\alpha} \left[\widetilde{P^*}(\tau) - \widetilde{w}(\tau) \right]$$

Inserting the optimal price (3.2) into the definition of $L_2(\tau)$ shows that $L_2(\tau) + \frac{\alpha}{\alpha - 1}L_4(\tau) = 0$. Using this and the fact that $L_1(\tau) + L_2(\tau) + L_3(\tau) + L_4(\tau) = \left[(\alpha - 1) \frac{1}{\lambda} \right]^{-1} \widetilde{w}(\tau)$, we have

$$\widetilde{w}(\tau) = h(\tau)\widetilde{w}(\tau) - \left[(\alpha - 1)\frac{1}{\lambda} \right] \frac{Y(\tau)}{Q(\tau)} \widetilde{P^*}(\tau)^{-\alpha} \left[\widetilde{P^*}(\tau) - \widetilde{w}(\tau) \right]$$

Linearising this equation yields

$$\begin{bmatrix} \widetilde{w}(\tau) - \widetilde{w} \end{bmatrix} = \begin{bmatrix} h + (\alpha - 1) \frac{1}{\lambda} \frac{Y}{Q} \widetilde{P^*}^{-\alpha} \end{bmatrix} \begin{bmatrix} \widetilde{w}(\tau) - \widetilde{w} \end{bmatrix} \\ + \widetilde{w} \begin{bmatrix} h(\tau) - h \end{bmatrix} \\ - (\alpha - 1) \frac{1}{\lambda} \widetilde{P^*}^{-\alpha} \left(\widetilde{P^*} - \widetilde{w} \right) \begin{bmatrix} \frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q} \end{bmatrix} \\ + (\alpha - 1) \frac{1}{\lambda} \frac{Y}{Q} \widetilde{P^*}^{-\alpha} \left[\alpha \widetilde{P^*}^{-1} \left(\widetilde{P^*} - \widetilde{w} \right) - 1 \right] \begin{bmatrix} \widetilde{P^*}(\tau) - \widetilde{P^*} \end{bmatrix}$$

We get equation (3.8) in the text by using that $h\left(\tau\right) = i\left(\tau\right) + \mu\left(\tau\right) - \pi\left(\tau\right) + \left(\alpha - 2\right)\gamma_{Q}\left(\tau\right) = r\left(\tau\right) + \left[1 + \left(\alpha - 2\right)\frac{q^{\alpha-1}-1}{\alpha-1}\right]\mu\left(\tau\right).$

3.A.3.1 Linearising Labour Market equilibrium

Taking the derivative with respect to time τ of both sides of equation (3.11) leads to

$$-\mu (\tau) \lambda \widetilde{q} = \frac{Y(\tau)}{Q(\tau)} \int_{-\infty}^{\tau} \left[\mu(s) + \beta \right] e^{-\int_{\theta=s}^{\tau} \left[\mu(\theta) - \alpha \widetilde{\pi}(\theta) + \beta \right] d\theta} \widetilde{P^*}(s)^{-\alpha} ds$$
$$+ \frac{Y(\tau)}{Q(\tau)} \left[\mu(\tau) + \beta \right] \widetilde{P^*}(\tau)^{-\alpha}$$
$$- \left[\mu(\tau) - \alpha \widetilde{\pi}(\tau) + \beta \right] \frac{Y(\tau)}{Q(\tau)} \int_{-\infty}^{\tau} \left[\mu(s) + \beta \right] e^{-\int_{\theta=s}^{\tau} \left[\mu(\theta) - \alpha \widetilde{\pi}(\theta) + \beta \right] d\theta} \widetilde{P^*}(s)^{-\alpha} ds$$

Noting that $\tilde{\pi}(\theta) = \pi(\theta) + \gamma_Q(\theta)$ and using again equation (3.11), this can be rewritten as

$$-\mu \dot{(\tau)} \lambda \widetilde{q} = \left\{ \frac{1}{\frac{Y(\tau)}{Q(\tau)}} \frac{Y(\tau)}{Q(\tau)} - \left[\mu(\tau) - \alpha \pi(\tau) - \alpha \gamma_Q(\tau) + \beta \right] \right\} \left[L - \mu(\tau) \lambda \widetilde{q} \right] \\ + \frac{Y(\tau)}{Q(\tau)} \left[\mu(\tau) + \beta \right] \widetilde{P^*}(\tau)^{-\alpha}$$

Linearising the equation, we have

$$\begin{split} -\lambda \widetilde{q}[\mu\left(\tau\right) - \mu] &= \left\{ \begin{aligned} \frac{\left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q}\right]}{\frac{Y}{Q}} - \left[\left(1 - \alpha \widehat{q}\right)\left[\mu\left(\tau\right) - \mu\right] - \alpha\left[\pi\left(\tau\right) - \pi\right]\right] \right\} (L - \mu\lambda \widetilde{q}) \\ &+ \left[\left(1 - \alpha \widehat{q}\right)\mu - \alpha\pi + \beta\right]\lambda \widetilde{q}\left[\mu\left(\tau\right) - \mu\right] \\ &+ \frac{Y}{Q} \widetilde{P^*}^{-\alpha}\left[\mu\left(\tau\right) - \mu\right] \\ &+ \widehat{\beta} \widetilde{P^*}^{-\alpha}\left[\frac{Y(\tau)}{Q\left(\tau\right)} - \frac{Y}{Q}\right] \\ &- \alpha \frac{Y}{Q} \widehat{\beta} \widetilde{P^*}^{-\alpha-1}\left[\widetilde{P^*}(\tau) - \widetilde{P^*}\right] \end{aligned}$$

where we have again used that $\gamma_Q(\tau) = \widehat{q}\mu(\tau)$. Collecting terms, using that at steady state, the equation for labour market equilibrium (3.11) reduces to $L - \mu\lambda\widetilde{q} = \frac{\widehat{\beta}}{\mu - +\beta\alpha\psi} \frac{Y}{Q} \widetilde{P^*}^{-\alpha}$ and rearranging, we have equation (3.12)

in the text.

3.A.4 Eliminating $\widetilde{P}^{*}(\tau)$ from the system

Using equation (3.7), we can eliminate $\tilde{P}^{*}(\tau)$ from the system of equations.

Turning first to the linearised equation for the optimal price (3.3), elim-

inating $\left[\widetilde{P^{*}}(\tau) - \widetilde{P^{*}}\right]$ and $\left[\widetilde{P^{*}}(\tau) - \widetilde{P^{*}}\right]$ yields

$$\begin{aligned} \left(\widehat{q} - \widehat{\psi}\right) \left[\mu(\tau) - \mu\right] + \left[\pi\left(\tau\right) - \pi\right] &= \\ &+ \left[\left(\frac{\widetilde{P^*}^{1-\alpha}}{L_2} \frac{Y}{Q} - \alpha\psi\right) - \widehat{\mu}\right] \left[\pi\left(\tau\right) - \pi\right] \\ &+ \left[\left(\frac{\widetilde{P^*}^{1-\alpha}}{L_2} \frac{Y}{Q} - \alpha\psi\right) \left(\widehat{q} - \widehat{\psi}\right) - \widehat{q}\widehat{\mu}\right] \left[\mu\left(\tau\right) - \mu\right] \\ &- \widehat{\mu} \frac{\widetilde{P^*}^{-\alpha}}{L_2} \frac{Y}{Q} \frac{\alpha}{\alpha - 1} \left[\widetilde{w}(\tau) - \widetilde{w}\right] \\ &+ \widehat{\mu} \frac{\psi}{\frac{Y}{Q}} \left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q}\right] \\ &- \widehat{\mu} \frac{\psi}{L_2} \left[L_2\left(\tau\right) - L_2\right] \end{aligned}$$
(3.23)

Similarly, the linearised zero profit equation (3.8) now reads

$$\begin{bmatrix} \widetilde{w}(\tau) - \widetilde{w} \end{bmatrix} = \begin{bmatrix} h + (\alpha - 1) \frac{1}{\lambda} \frac{Y}{Q} \widetilde{P^*}^{-\alpha} \end{bmatrix} \begin{bmatrix} \widetilde{w}(\tau) - \widetilde{w} \end{bmatrix} + \widetilde{w} [r(\tau) - r] + a_{\widetilde{w}\mu} [\mu(\tau) - \mu] - (\alpha - 1) \frac{1}{\lambda} \frac{Y}{Q} \left[1 + \alpha \widetilde{P^*}^{-1} \left(\widetilde{w} - \widetilde{P^*} \right) \right] \frac{\widetilde{P^*}^{1-\alpha}}{\widehat{\mu}} [\pi(\tau) - \pi] + (\alpha - 1) \frac{1}{\lambda} \widetilde{P^*}^{-\alpha} \left(\widetilde{w} - \widetilde{P^*} \right) \left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q} \right]$$
(3.24)

where $a_{\widetilde{w}\mu} = \left\{ \widetilde{w} \left[1 + (\alpha - 2) \, \widehat{q} \right] - \left[1 + \frac{\alpha}{\widetilde{P^*}} \left(\widetilde{w} - \widetilde{P^*} \right) \right] \frac{(\alpha - 1) \frac{1}{\lambda} \frac{V}{Q} \widetilde{P^*}^{1 - \alpha}}{\widehat{\mu}} \left(\widehat{q} - \widehat{\psi} \right) \right\}.$

Eliminating $\left[\widetilde{P^*}(\tau) - \widetilde{P^*}\right]$ from the labour market equilibrium condition (3.12) gives:

$$[\mu(\tau) - \mu] = -(\lambda \widetilde{q})^{-1} \frac{\widehat{\beta}}{\widetilde{\mu}} \widetilde{P^{*}}^{-\alpha} \left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q} \right]$$

$$+ \alpha (\lambda \widetilde{q})^{-1} \frac{Y}{Q} \widetilde{P^{*}}^{-\alpha} \left(\frac{\widehat{\beta}}{\widehat{\mu}} - \frac{\widehat{\beta}}{\widetilde{\mu}} \right) [\pi(\tau) - \pi]$$

$$+ \left\{ \alpha (\lambda \widetilde{q})^{-1} \frac{Y}{Q} \left(\widehat{q} - \frac{\psi}{\widehat{\beta}} \right) \widetilde{P^{*}}^{-\alpha} \left(\frac{\widehat{\beta}}{\widehat{\mu}} - \frac{\widehat{\beta}}{\widetilde{\mu}} \right) - \widetilde{\mu} \right\} [\mu(\tau) - \mu]$$

$$- (\lambda \widetilde{q})^{-1} \widehat{\beta} \widetilde{P^{*}}^{-\alpha} \left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q} \right]$$

$$(3.25)$$

In the linearised equation describing revenues, (3.22), eliminating $\left[\widetilde{P^*}(\tau) - \widetilde{P^*}\right]$ and $\left[\widetilde{P^*}(\tau) - \widetilde{P^*}\right]$ leads to

$$[L_{2}(\tau) - L_{2}] = -\widetilde{P^{*}}^{1-\alpha} \left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q} \right]$$

$$+ \left[L_{2}(1-\widehat{q}) - \frac{Y}{Q} \frac{(1-\alpha)\widetilde{P^{*}}^{1-\alpha}}{\widehat{\mu}} \left(\widehat{q} - \widehat{\psi} \right) \right] [\mu(\tau) - \mu]$$

$$+ \frac{(1-\alpha)L_{2}}{\widehat{\mu}} \left\{ \left(\widehat{q} - \widehat{\psi} \right) \left(\mu(\tau) - \mu \right) + \left(\pi(\tau) - \pi \right) \right\}$$

$$+ L_{2}[r(\tau) - r]$$

$$+ (\alpha - 1) \left[\frac{Y}{Q} \frac{\widetilde{P^{*}}^{1-\alpha}}{\widehat{\mu}} - L_{2} \right] [\pi(\tau) - \pi]$$

$$+ [h - (\alpha - 1)\psi + \beta] [L_{2}(\tau) - L_{2}]$$
(3.26)

3.A.5 Eliminating $\frac{m(\tau)}{Q(\tau)}$ from the system

We first eliminate $\frac{m(\tau)}{Q(\tau)}$ from the household's dynamic efficiency equation (3.16) by plugging in the value of $\frac{m(\tau)}{Q(\tau)}$ given in the money market equation

(3.18). Rearranging, this yields

$$\left(\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q}\right) = -\frac{\theta(1-\eta)}{\hat{\eta}} \frac{Y}{Q} (\pi(\tau) - \pi)
-\hat{q} \frac{Y}{Q} (\mu(\tau) - \mu)
+ \frac{\frac{Y}{Q}}{\eta + \theta(1-\eta)} (r(\tau) - r)$$
(3.27)

where $\hat{\eta} = \eta + \theta (1 - \eta)$. We then use the household's static efficiency condition (3.15) to eliminate $\frac{m(\tau)}{Q(\tau)}$ from equation (3.18), which results in:

$$\frac{\theta}{1-\theta}\frac{1}{r+\pi}\left(\frac{Y(\tau)}{Q(\tau)}-\frac{Y}{Q}\right) = \frac{\frac{m}{Q}}{r+\pi}(r(\tau)-r) + \frac{\frac{m}{Q}}{r+\pi}(\pi(\tau)-\pi) - \frac{m}{Q}(\pi(\tau)-\pi) - \frac{m}{Q}(\pi(\tau)-\pi) - \frac{m}{Q}\hat{q}(\mu(\tau)-\mu)$$
(3.28)

3.A.6 Rewriting the system

The system of differential equations is still in the form $y(\tau) = f\left[z(\tau), y(\tau)\right]$, where $y(\tau)$ is the vector containing the deviations of all variables in the system from their steady state values and $z(\tau)$ is a subset of $y(\tau)$. We therefore need to rearrange the system to write it in the form $y(\tau) = f[y(\tau)]$.

We first solve equation (3.23) for the expression $\left(\widehat{q} - \widehat{\psi}\right) \left(\mu(\tau) - \mu\right) + \left(\pi(\tau) - \pi\right) \text{ and insert this into equation (3.26) which}$ gives us $[L_2(\tau) - L_2]$ as a function of current level deviations of variables only:

$$\begin{split} \left[L_{2}\left(\tau\right)-L_{2}\right] &= -\widetilde{P^{*}}^{1-\alpha}\left[\frac{Y(\tau)}{Q(\tau)}-\frac{Y}{Q}\right] \\ &+ \left[L_{2}\left(1-\widehat{q}\right)-\left(1-\alpha\right)\frac{Y}{Q}\frac{\widetilde{P^{*}}^{1-\alpha}}{\widehat{\mu}}\left(\widehat{q}-\widehat{\psi}\right)\right]\left[\mu\left(\tau\right)-\mu\right] \\ &+ \frac{\left(1-\alpha\right)L_{2}}{\widehat{\mu}}\left\{\left(\frac{\widetilde{P^{*}}^{1-\alpha}}{L_{2}}\frac{Y}{Q}-\alpha\psi\right)-\widehat{\mu}\right\}\left[\pi\left(\tau\right)-\pi\right] \\ &+ \frac{\left(1-\alpha\right)L_{2}}{\widehat{\mu}}\left\{\left(\frac{\widetilde{P^{*}}^{1-\alpha}}{L_{2}}\frac{Y}{Q}-\alpha\psi\right)\left(\widehat{q}-\widehat{\psi}\right)-\widehat{q}\widehat{\mu}\right\}\left[\mu\left(\tau\right)-\mu\right] \\ &- \frac{\left(1-\alpha\right)L_{2}}{\widehat{\mu}}\left\{\widehat{\mu}\frac{\widetilde{P^{*}}^{-\alpha}}{L_{2}}\frac{Y}{Q}\frac{\alpha}{\alpha-1}\left[\widetilde{w}(\tau)-\widetilde{w}\right]\right\} \\ &+ \frac{\left(1-\alpha\right)L_{2}}{\widehat{\mu}}\widehat{\mu}\frac{\psi}{Y}\left[\frac{Y(\tau)}{Q(\tau)}-\frac{Y}{Q}\right] \\ &- \frac{\left(1-\alpha\right)L_{2}}{\widehat{\mu}}\widehat{\mu}\frac{\psi}{L_{2}}\left[L_{2}\left(\tau\right)-L_{2}\right] \\ &+ L_{2}\left[r\left(\tau\right)-r\right] \\ &+ \left(\alpha-1\right)\left[\frac{Y}{Q}\frac{\widetilde{P^{*}}^{1-\alpha}}{\widehat{\mu}}-L_{2}\right]\left[\pi\left(\tau\right)-\pi\right] \\ &+ \left[h-\left(\alpha-1\right)\psi+\beta\right]\left[L_{2}\left(\tau\right)-L_{2}\right] \end{split}$$

Collecting terms, we have

$$[L_{2}(\tau) - L_{2}] = + (\alpha - 1) \alpha \psi \frac{L_{2}}{\widehat{\mu}} [\pi(\tau) - \pi]$$

$$+ L_{2} \left\{ \left(q^{\alpha - 1} - \widehat{q} \right) + \left(\widehat{q} - \widehat{\psi} \right) \frac{\alpha \psi (\alpha - 1)}{\widehat{\mu}} \right\} [\mu(\tau) - \mu]$$

$$+ \alpha \widetilde{P^{*}}^{-\alpha} \frac{Y}{Q} [\widetilde{w}(\tau) - \widetilde{w}]$$

$$+ \left[(1 - \alpha) L_{2} \frac{\psi}{\frac{Y}{Q}} - \widetilde{P^{*}}^{1 - \alpha} \right] \left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q} \right]$$

$$+ L_{2} [r(\tau) - r]$$

$$+ (h + \beta) [L_{2}(\tau) - L_{2}]$$
(3.29)

Similarly, in a second step we use equation (3.27) to substitute for $\left(\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q}\right)$ in equation (3.25), which gives

$$\begin{pmatrix} \mu(\tau) - \mu \end{pmatrix} = \begin{cases} -\widetilde{\mu} + (\widetilde{q}\lambda)^{-1} \widetilde{P^*}^{-\alpha} \frac{Y}{Q} \left[\frac{\widehat{\beta}}{\widetilde{\mu}} \widehat{q} + \alpha \left(\widehat{q} - \widehat{\psi} \right) \left(\frac{\widehat{\beta}}{\widehat{\mu}} - \frac{\widehat{\beta}}{\widetilde{\mu}} \right) \right] \end{cases} (\mu(\tau) - \mu) \\ - (\widetilde{q}\lambda)^{-1} \widehat{\beta} \widetilde{P^*}^{-\alpha} \left(\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q} \right) \\ + (\widetilde{q}\lambda)^{-1} \frac{Y}{Q} \widetilde{P^*}^{-\alpha} \left[\frac{\widehat{\beta}}{\widetilde{\mu}} \frac{\theta(1-\eta)}{\widehat{\eta}} + \alpha \left(\frac{\widehat{\beta}}{\widehat{\mu}} - \frac{\widehat{\beta}}{\widetilde{\mu}} \right) \right] (\pi(\tau) - \pi) \\ - (\widetilde{q}\lambda)^{-1} \widetilde{P^*}^{-\alpha} \frac{\widehat{\beta}}{\widetilde{\mu}} \frac{Y}{\widehat{\eta}} (r(\tau) - r)$$
(3.30)

We then use equation (3.30) to eliminate $[\mu(\tau) - \mu]$ from equation (3.23), which yields

$$\begin{bmatrix} \pi(\tau) - \pi \end{bmatrix} = \widetilde{a}_{\pi\pi} [\pi(\tau) - \pi] + \widetilde{a}_{\pi\mu} [\mu(\tau) - \mu] + \left\{ \widehat{\mu} \frac{\psi}{\frac{Y}{Q}} + \left(\widehat{q} - \widehat{\psi}\right) (\widetilde{q}\lambda)^{-1} \widehat{\beta} \widetilde{P^*}^{-\alpha} \right\} \begin{bmatrix} \underline{Y(\tau)} - \underline{Y} \\ Q(\tau) - \underline{Y} \end{bmatrix} + \left(\widehat{q} - \widehat{\psi}\right) (\widetilde{q}\lambda)^{-1} \widetilde{P^*}^{-\alpha} \frac{\widehat{\beta}}{\widetilde{\mu}} \frac{\underline{Y}}{\widehat{\eta}} [r(\tau) - r] - \widehat{\mu} \frac{\widetilde{P^*}^{-\alpha}}{L_2} \frac{Y}{Q} \frac{\alpha}{\alpha - 1} [\widetilde{w}(\tau) - \widetilde{w}] - \widehat{\mu} \frac{\psi}{L_2} [L_2(\tau) - L_2]$$
(3.31)

where
$$\widetilde{a}_{\pi\mu} = \left(\widehat{q} - \widehat{\psi}\right) \left(\frac{\widetilde{P^*}^{1-\alpha}}{L_2} \frac{Y}{Q} - \alpha\psi\right) - \widehat{q}\widehat{\mu} + \left(\widehat{q} - \widehat{\psi}\right) \left\{\widetilde{\mu} - (\widetilde{q}\lambda)^{-1} \frac{Y}{Q} \widetilde{P^*}^{-\alpha} \left[\alpha \left(\widehat{q} - \widehat{\psi}\right) \left(\frac{\widehat{\beta}}{\widehat{\mu}} - \frac{\widehat{\beta}}{\widehat{\mu}}\right) + \frac{\widehat{\beta}}{\widetilde{\mu}} \widehat{q}\right]\right\} \text{ and}$$

 $\widetilde{a}_{\pi\pi} = \left(\frac{\widetilde{P^*}^{1-\alpha}}{L_2} \frac{Y}{Q} - \alpha\psi\right) - \widehat{\mu} - \left(\widehat{q} - \widehat{\psi}\right) (\widetilde{q}\lambda)^{-1} \frac{Y}{Q} \widetilde{P^*}^{-\alpha} \left[\frac{\widehat{\beta}}{\widetilde{\mu}} \frac{\theta(1-\eta)}{\widehat{\eta}} + \alpha \left(\frac{\widehat{\beta}}{\widehat{\mu}} - \frac{\widehat{\beta}}{\widetilde{\mu}}\right)\right].$

In a last step, we use this equation (3.31) and the household's dynamic efficiency condition (3.27) to substitute for $[\pi(\tau) - \pi]$ and $\left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q}\right]$ in equation (3.28), respectively. To this end, we first rewrite equation (3.28) as

$$(r(\tau) - r) = -(\pi(\tau) - \pi)$$

$$+ \frac{\theta}{1 - \theta} \frac{1}{\frac{m}{Q}} \left(\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q} \right)$$

$$+ (r + \pi) (\pi(\tau) - \pi)$$

$$+ (r + \pi) \hat{q} (\mu(\tau) - \mu)$$

Plugging in (3.27) for $\left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q}\right]$ yields

$$\begin{split} (r\left(\tau\right)-r) &= -(\pi\left(\tau\right)-\pi) \\ &+ \left[(r+\pi) - \frac{\theta}{1-\theta} \frac{\frac{Y}{Q}}{\frac{m}{Q}} \frac{\theta(1-\eta)}{\widehat{\eta}} \right] (\pi\left(\tau\right)-\pi) \\ &+ \widehat{q} \left[r+\pi - \frac{\theta}{1-\theta} \frac{\frac{Y}{Q}}{\frac{m}{Q}} \right] (\mu\left(\tau\right)-\mu) \\ &+ \frac{\theta}{1-\theta} \frac{\frac{Y}{Q}}{\frac{m}{Q}} \frac{1}{\widehat{\eta}} \left(r\left(\tau\right)-r \right) \end{split}$$

where the coefficient on the deviation $(\mu(\tau) - \mu)$ is zero given equation (3.14).

Further using equation (3.31) to eliminate $(\pi (\tau) - \pi)$ from this equation

and collecting terms yields

$$(r(\tau) - r) = a_{r\mu} [\mu(\tau) - \mu]$$

$$- \left\{ \widehat{\mu} \frac{\psi}{\underline{Y}} + \left(\widehat{q} - \widehat{\psi}\right) (\widetilde{q}\lambda)^{-1} \widehat{\beta} \widetilde{P^*}^{-\alpha} \right\} \left[\frac{Y(\tau)}{Q(\tau)} - \frac{Y}{Q} \right]$$

$$+ a_{r\pi} [\pi(\tau) - \pi]$$

$$+ \frac{1}{\widehat{\eta}} \left[r + \pi - \left(\widehat{q} - \widehat{\psi}\right) (\widetilde{q}\lambda)^{-1} \widetilde{P^*}^{-\alpha} \frac{\widehat{\beta}}{\widetilde{\mu}} \frac{Y}{Q} \right] [r(\tau) - r]$$

$$+ \widehat{\mu} \frac{\widetilde{P^*}^{-\alpha}}{L_2} \frac{Y}{Q} \frac{\alpha}{\alpha - 1} [\widetilde{w}(\tau) - \widetilde{w}]$$

$$+ \widehat{\mu} \frac{\psi}{L_2} [L_2(\tau) - L_2]$$
(3.32)

with

$$a_{r\mu} = -\left(\widehat{q} - \widehat{\psi}\right) \left(\frac{\widetilde{P^{*}}^{1-\alpha}}{L_{2}} \frac{Y}{Q} + \widetilde{\mu} - \alpha\psi\right) + \widehat{q}\widehat{\mu} + \left(\widehat{q} - \widehat{\psi}\right) \frac{\frac{Y}{Q}}{\widetilde{q}\lambda} \widetilde{P^{*}}^{-\alpha} \left[\alpha \left(\widehat{q} - \widehat{\psi}\right) \left(\frac{\widetilde{\beta}}{\widehat{\mu}} - \frac{\widetilde{\beta}}{\widetilde{\mu}}\right) + \frac{\widetilde{\beta}}{\widetilde{\mu}} \widehat{q}\right]$$

and
$$a_{r\pi} = (r + \pi) \frac{\eta}{\widehat{\eta}} - \left(\frac{\widetilde{P^{*}}^{1-\alpha}}{L_{2}} \frac{Y}{Q} - \alpha\psi\right) + \widehat{\mu} + \left(\widehat{q} - \widehat{\psi}\right) (\widetilde{q}\lambda)^{-1} \frac{Y}{Q} \widetilde{P^{*}}^{-\alpha} \left[\frac{\widetilde{\beta}}{\widetilde{\mu}} \frac{\theta(1-\eta)}{\widehat{\eta}} + \alpha \left(\frac{\widetilde{\beta}}{\widetilde{\mu}} - \frac{\widetilde{\beta}}{\widetilde{\mu}}\right)\right]$$

3.A.7 Elements of the linearised system's matrix

$$m_{ij}$$
 refers to the element in the *i*th row and *j*th column of the matrix **M**.

$$\begin{split} m_{11} &= 0, \ m_{12} = \frac{Y}{Q} \frac{1}{\hat{\eta}}, \ m_{13} = -\frac{Y}{Q} \hat{q}, \ m_{14} = -\frac{Y}{Q} \frac{\theta(1-\eta)}{\hat{\eta}}, \ m_{15} = 0, \ m_{16} = 0, \\ m_{21} &= -\widetilde{P^*}^{-\alpha} \left\{ \widehat{\beta} \left(\widehat{q} - \widehat{\psi} \right) (\widetilde{q}\lambda)^{-1} + \frac{\widehat{\mu}}{L_2} \left(\widetilde{P^*} - \frac{\alpha}{\alpha-1} \widetilde{w} \right) \right\}, \ m_{22} = \frac{r+\pi}{\hat{\eta}} - \frac{\widehat{q} - \widehat{\psi}}{\widehat{q}\lambda \widetilde{P^*}^{\alpha}} \frac{\widehat{\beta}}{\widehat{\mu}} \frac{Y}{Q} \frac{1}{\hat{\eta}}, \\ m_{23} &= -\frac{\widehat{\mu} \left(\frac{1}{L_2} \widetilde{P^*}^{1-\alpha} \frac{Y}{Q} - \alpha \psi \right) (\widehat{q} - \widehat{\psi}) - \widehat{q} \widehat{\mu}^2}{\widehat{\mu}} + \left(\widehat{q} - \widehat{\psi} \right) \left\{ \frac{\frac{Y}{Q}}{\widehat{q}\lambda \widetilde{P^*}^{\alpha}} \left[\alpha \left(\frac{\widehat{\beta}}{\widehat{\mu}} - \frac{\widehat{\beta}}{\widehat{\mu}} \right) \left(\widehat{\psi} - \widehat{q} \right) + \frac{\widehat{\beta}}{\widehat{\mu}} \widehat{q} \right] - \widetilde{\mu} \right\}, \\ m_{24} &= (r+\pi) \frac{\eta}{\hat{\eta}} - \frac{\widehat{\mu} \left(\frac{1}{L_2} \widetilde{P^*}^{1-\alpha} \frac{Y}{Q} - \alpha \psi \right) - \widehat{\mu}^2}{\widehat{\mu}} + \widetilde{P^*}^{-\alpha} \left(\widehat{q} - \widehat{\psi} \right) (\widetilde{q}\lambda)^{-1} \frac{Y}{Q} \left[\frac{\widehat{\beta}}{\widehat{\mu}} \frac{\theta(1-\eta)}{\widehat{\eta}} + \alpha \left(\frac{\widehat{\beta}}{\widehat{\mu}} - \frac{\widehat{\beta}}{\widehat{\mu}} \right) \right], \\ m_{25} &= \frac{\widehat{\mu}}{L_2} \frac{\alpha}{\alpha-1} \frac{Y}{Q} \widetilde{P^*}^{-\alpha}, \ m_{26} &= \widehat{\mu} \psi \frac{1}{L_2}, \\ m_{31} &= - (\widetilde{q}\lambda)^{-1} \widehat{\beta} \widetilde{P^*}^{-\alpha}, \ m_{32} &= -\frac{\widetilde{P^{*-\alpha}}}{\widehat{q}\lambda} \frac{\widehat{\beta}}{\widehat{\mu}} \frac{Y}{Q} \frac{1}{\widehat{\eta}}, \\ m_{33} &= (\widetilde{q}\lambda)^{-1} \widetilde{P^*}^{-\alpha} \frac{Y}{Q} \left[\alpha \left(\frac{\widehat{\beta}}{\widehat{\mu}} - \frac{\widehat{\beta}}{\widehat{\mu}} \right) \left[\widehat{\psi} - \widehat{q} \right] + \frac{\widehat{\beta}}{\widehat{\mu}} \widehat{q} \right] - \widetilde{\mu}, \\ m_{34} &= (\widetilde{q}\lambda)^{-1} \widetilde{P^*}^{-\alpha} \frac{Y}{Q} \left\{ \frac{\widehat{\beta}}{\widehat{\mu}} \frac{\theta(1-\eta)}{\widehat{\eta}} + \alpha \left[\frac{\widehat{\beta}}{\widehat{\mu}} - \frac{\widehat{\beta}}{\widehat{\mu}} \right] \right\}, \ m_{35} &= 0, \ m_{36} &= 0, \\ m_{41} &= \widetilde{P^*}^{-\alpha} \left\{ \widehat{\beta} \frac{\widehat{q} - \widehat{\psi}}{\widehat{q}\lambda} + \frac{\widehat{\mu}}{L_2} \left(\widetilde{P^*} - \frac{\alpha}{\alpha-1} \widetilde{w} \right) \right\}, \ m_{42} &= \widetilde{P^*}^{-\alpha} \frac{\widehat{q} - \widehat{\psi}}{\widehat{q}\lambda} \frac{\widehat{\beta}}{\widehat{\mu}} \frac{Y}{\widehat{\eta}}, \end{aligned}$$

Chapter 3. Local Stability

$$\begin{split} m_{43} &= \frac{\hat{\mu} \left(\frac{1}{L_2} \widetilde{P^*}^{1-\alpha} \frac{Y}{Q} - \alpha \psi\right) (\hat{q} - \hat{\psi}) - \hat{\mu}^2 \hat{q}}{\hat{\mu}} - \left(\widehat{q} - \widehat{\psi}\right) \left\{ \frac{\frac{Y}{Q}}{\tilde{q}\lambda \widetilde{P^*}^{\alpha}} \left[\alpha \left(\frac{\hat{\beta}}{\tilde{\mu}} - \frac{\hat{\beta}}{\tilde{\mu}} \right) \left(\widehat{\psi} - \widehat{q} \right) + \frac{\hat{\beta}}{\tilde{\mu}} \widehat{q} \right] - \widetilde{\mu} \right\}, \\ m_{44} &= \frac{\hat{\mu} \left(\frac{1}{L_2} \widetilde{P^*}^{1-\alpha} \frac{Y}{Q} - \alpha \psi\right) - \hat{\mu}^2}{\hat{\mu}} - \widetilde{P^*}^{-\alpha} \left(\widehat{q} - \widehat{\psi} \right) (\widetilde{q}\lambda)^{-1} \frac{Y}{Q} \left\{ \frac{\hat{\beta}}{\tilde{\mu}} \frac{\theta(1-\eta)}{\hat{\eta}} + \alpha \left[\frac{\hat{\beta}}{\tilde{\mu}} - \frac{\hat{\beta}}{\tilde{\mu}} \right] \right\}, \\ m_{45} &= -\frac{\hat{\mu}}{L_2} \frac{\alpha}{\alpha-1} \frac{Y}{Q} \widetilde{P^*}^{-\alpha}, m_{46} = -\hat{\mu} \psi \frac{1}{L_2}, \\ m_{51} &= (\alpha - 1) \frac{1}{\lambda} \widetilde{P^*}^{-\alpha} \left(\widetilde{w} - \widetilde{P^*} \right), m_{52} = \widetilde{w}, \\ m_{53} &= \widetilde{w} \left[1 + (\alpha - 2) \, \widehat{q} \right] - \frac{\alpha-1}{\lambda} \frac{Y}{Q} \left[\frac{\alpha}{\widetilde{P^*}} \left(\widetilde{w} - \widetilde{P^*} \right) + 1 \right] \frac{\widetilde{P^*}^{1-\alpha}}{\hat{\mu}} \left(\widehat{q} - \widehat{\psi} \right), \\ m_{54} &= -(\alpha - 1) \frac{1}{\lambda} \frac{Y}{Q} \left[\alpha \left(\widetilde{w} - \widetilde{P^*} \right) \frac{1}{\widetilde{P^*}} + 1 \right] \frac{\widetilde{P^*}^{1-\alpha}}{\hat{\mu}}, \\ m_{55} &= \left[\widehat{r} - \beta \right] + (\alpha - 1) \frac{1}{\lambda} \frac{Y}{Q} \widetilde{P^*}^{-\alpha}, m_{56} = 0, \\ m_{61} &= \widetilde{P^*}^{-\alpha} \left[\left(1 - \alpha \right) \left(\widetilde{P^*} - \frac{\alpha}{\alpha-1} \widetilde{w} \right) - \widetilde{P^*} \right], m_{62} = L_2, \\ m_{63} &= L_2 \left[\frac{\alpha-1}{\hat{\mu}} \alpha \psi \left(\widehat{q} - \widehat{\psi} \right) + \left(q^{\alpha-1} - \widehat{q} \right) \right], \\ m_{64} &= \alpha \frac{\alpha-1}{\hat{\mu}} \psi L_2, m_{65} = \alpha \frac{Y}{Q} \widetilde{P^*}^{-\alpha}, m_{66} = \widehat{r} \end{split}$$

Chapter 4

Limit Pricing

4.1 Introduction

In the first chapters of this thesis, it was assumed that the incumbent intermediate firm's production is terminated when an innovator enters the market with an improved version of the good produced by the incumbent.¹ We now relax this assumption: Existing firms may leave the market whenever an innovator enters but are *not* required to do so. If the incumbent makes nonpositive profits in case the innovator sets the monopoly price, the latter will do so and we are back in the model of chapter 2. If in contrast exogenous parameters are such that the incumbent would make positive profits under monopoly pricing, the innovator will set a limit price to drive the incumbent out of the market. In this chapter, we examine the model's steady state equilibrium under limit pricing.

First, it is important to clarify under which conditions limit pricing will prevail. Remember that we interpret an innovation in sector j as a signal that allows both the innovator and the incumbent(s) in sector j to choose a new price. Equation (2.1) shows that in the production of the final good, used goods from sector j are weighted with their respective quality grades. Being one quality rung below the innovator, the incumbent can at most

¹Footnotes 5 and 8 explained that alternatively, it could be assumed that parameters are such that the incumbent makes non-positive profit when the innovator sets the monopoly price for his good and that the relevant conditions hold in all examined numerical examples.

charge a price that is a fraction q of the innovator's price. If charging 1/qtimes the monopoly price allows the incumbent to make a positive profit and thus to stay in the market, the innovator will not charge the monopoly price. Instead, he will choose the limit price $p^{\lim}(\tau)$ which makes the incumbent indifferent about exiting the market. In the model without price rigidity, the incumbent can make a positive profit under monopoly pricing whenever $(1/q) * \alpha/(\alpha - 1) > 1$. Thus under this condition limit pricing prevails, while the innovator sets the optimal monopolistic price whenever $(1/q) * \alpha/(\alpha - 1) \leq 1$. Now in the model with rigidity, the optimal monopolistic mark-up depends on endogenous variables such as the rate of inflation. Therefore, knowing the model's exogenous parameters is not in all cases sufficient to establish whether limit pricing or monopoly pricing will prevail. We will here present conditions that are sufficient for the applicability of monopoly and limit pricing, respectively. As stated in section 2.2.2.1, monopoly pricing will prevail when the sufficient conditions $\frac{1}{q}\frac{\alpha}{\alpha-1} \leq 1$ and $\frac{1}{q} \frac{\alpha}{\alpha - 1} \frac{\rho + \beta + (1 - \alpha)\psi}{\rho + \beta - \alpha\psi} \leq 1 \text{ hold.}^2 \text{ Conversely, when the two following conditions}$ both hold, firms will set limit prices:

$$\frac{1}{q}\frac{\alpha}{\alpha-1} > 1 \tag{4.1}$$

$$\frac{1}{q}\frac{\alpha}{\alpha-1}\frac{\rho+\beta+(1-\alpha)\psi}{\rho+\beta-\alpha\psi} > 1$$
(4.2)

If $\psi \geq 0$, the endogenous part of the firm's mark-up $\frac{\rho+\beta+\tilde{\eta}\mu+(1-\alpha)\psi}{\rho+\beta+\tilde{\eta}\mu-\alpha\psi}$ is equal to or bigger than one, such that the condition $\frac{1}{q}\frac{\alpha}{\alpha-1} > 1$ from the underlying real model is sufficient to ensure that limit prices will be charged. When $\psi < 0$, the endogenous part of the mark-up is smaller than one. Since it increases in μ , $\frac{1}{q} \frac{\alpha}{\alpha-1} \frac{\rho+\beta+(1-\alpha)\psi}{\rho+\beta-\alpha\psi} > 1$ is a sufficient condition for limit pricing in this case. We therefore assume in this chapter that both (4.1) and (4.2)hold.³

²For $\psi \leq 0$, the condition $\frac{1}{q} \frac{\alpha}{\alpha - 1} \leq 1$ is sufficient for monopoly pricing since in this case we have for the endogenous part of the mark-up $\frac{\rho+\beta+\tilde{\eta}\mu+(1-\alpha)\psi}{\rho+\beta+\tilde{\eta}\mu-\alpha\psi} \leq 1$. For $\psi > 0$, though, the endogenous part of the mark-up is bigger than one and decreases in μ . Therefore, $\frac{1}{q}\frac{\alpha}{\alpha-1}\frac{\rho+\beta+(1-\alpha)\psi}{\rho+\beta-\alpha\psi} \leq 1$ is a sufficient condition for monopoly pricing for $\psi > 0$. ³Note that this chapter makes no statement as to which pricing regime will hold in

The remainder of this chapter is organised as follows: In section 4.2, the general equilibrium of the model under limit pricing is derived, where only the elements that differ from the monopoly case will be discussed in detail. It is shown that, under some additional assumptions, the model with limit pricing has a unique steady state equilibrium. Section 4.4 then discusses changes in the comparative statics of the equilibrium and presents a calibrated example. Section 4.5 concludes.

4.2 The model with limit pricing

4.2.1 The limit price

To prevent the incumbent from staying in the market, the innovator chooses the limit price $p^{\text{lim}}(\tau)$ which given marginal cost $w(\tau)$ makes the incumbent

the case where none of the sets of conditions sufficient for either limit pricing or monopoly pricing holds. The following intuition implies that in these cases non-zero money growth might give rise to multiple equilibria: Take a situation where $\frac{1}{q} \frac{\alpha}{\alpha-1} < 1$ and $\psi > 0$. Since the optimal monopolistic mark-up $\frac{\rho+\beta+\left[(\alpha-2+\eta)\frac{q^{\alpha-1}-1}{\alpha-1}+1\right]\mu+(1-\alpha)\psi}{\rho+\beta+\left[(\alpha-2+\eta)\frac{q^{\alpha-1}-1}{\alpha-1}+1\right]\mu-\alpha\psi}$ decreases in μ , whether the incumbent can make a positive profit at the monopoly price (i.e., whether $\frac{1}{q} \frac{\alpha}{\alpha-1} \frac{\rho+\beta+\left[(\alpha-2+\eta)\frac{q^{\alpha-1}-1}{\alpha-1}+1\right]\mu+(1-\alpha)\psi}{\rho+\beta+\left[(\alpha-2+\eta)\frac{q^{\alpha-1}-1}{\alpha-1}+1\right]\mu-\alpha\psi} > 1$) depends on μ . Two scenarios are possible: First, all firms expect that μ will be high so that $\frac{1}{q} \frac{\alpha}{\alpha-1} \frac{\rho+\beta+\left[(\alpha-2+\eta)\frac{q^{\alpha-1}-1}{\alpha-1}+1\right]\mu+(1-\alpha)\psi}{\rho+\beta+\left[(\alpha-2+\eta)\frac{q^{\alpha-1}-1}{\alpha-1}+1\right]\mu-\alpha\psi} > 1$ and therefore set the optimal monopoly price. Second, firms expect that μ will be low so that $\frac{1}{q} \frac{\alpha}{\alpha-1} \frac{\rho+\beta+\left[(\alpha-2+\eta)\frac{q^{\alpha-1}-1}{\alpha-1}+1\right]\mu+(1-\alpha)\psi}{\rho+\beta+\left[(\alpha-2+\eta)\frac{q^{\alpha-1}-1}{\alpha-1}+1\right]\mu-\alpha\psi} > 1$ and therefore set the limit price. This chapter's analysis of steady state equilibrium under limit pricing will show that by taking away firms' opportunity to offset the mark-up erosion effect by choosing a correspondingly higher initial mark-up, limit pricing creates an additional channel for positive money growth to reduce profits and the incentive to innovate. Therefore, profits and the equilibrium value of μ will be lower (say, μ^{lim}) in the second case where all firms set limit prices relative to the first case (say, μ^{mon}) where firms set the optimal monopoly price. This implies that $\frac{1}{q} \frac{\alpha}{\alpha-1} \frac{\rho+\beta+\left[(\alpha-2+\eta)\frac{q^{\alpha-1}-1}{\alpha-1}+1\right]\mu^{lim}+(1-\alpha)\psi}{\rho+\beta+\left[(\alpha-2+\eta)\frac{q^{\alpha-1}-1}{\alpha-1}+1\right]\mu^{lim}-\alpha\psi} < 1 < \frac{1}{q} \frac{\alpha}{\alpha-1} \frac{\rho+\beta+\left[(\alpha-2+\eta)\frac{q^{\alpha-1}-1}{\alpha-1}+1\right]\mu^{lim}}$ (could hold, confirming firms' expectations

in both cases and making both limit pricing and monopoly pricing an equilibrium for the given money growth rate $\psi > 0$.

indifferent about exiting the market:⁴

$$p^{\lim}\left(\tau\right) = qw\left(\tau\right) \tag{4.3}$$

Analogously to the optimal monopoly price, the optimal limit price grows at the same rate as marginal cost. Yet unlike in the case with monopoly pricing, when quality increments q are small and limit pricing prevails, the innovator has no margin to choose a forward-looking mark-up.

4.2.2 A new firm's market value at entry

As in the monopoly case, the market value of a new firm at entry is

$$E\left(V_{k_{j}}\left(\tau\right)\left|t_{k_{j}}=\tau\right.\right)=A\left(\tau\right)P^{*}\left(\tau\right)^{-\alpha}\frac{\chi+\beta+\left(\alpha-1\right)\omega}{\chi-\left(1-\alpha\right)\omega}\left[\frac{p_{j}\left(\tau\right)}{\chi+\beta}-\frac{w\left(\tau\right)}{\chi+\beta-\omega}\right]$$

where $A(\tau) = q^{k_j(\alpha-1)}Y(\tau)P(\tau)^{\alpha}$, $\chi = i + \mu - \alpha\pi - \gamma$ and $p_j(\tau)$ is the price chosen by the innovator at the time of his market entry $\tau = t_{k_j}$. Inserting the limit price (4.3) gives the innovator's market value under limit pricing:

$$E\left(V_{k_{j}}\left(\tau\right)\left|t_{k_{j}}=\tau\right.\right) = \frac{A\left(\tau\right)w\left(\tau\right)}{P^{*}\left(\tau\right)^{\alpha}}\frac{\chi+\beta+\left(\alpha-1\right)\omega}{\chi-\left(1-\alpha\right)\omega}\left[\frac{q}{\chi+\beta}-\frac{1}{\chi+\beta-\omega}\right]$$
(4.4)

The term in square brackets can be interpreted as the present value of the firm's average profit per unit. Since the growth rate of marginal cost is still given by the money growth rate ψ , this term is influenced by ψ . In the monopoly case, firms choose a forward-looking mark-up that neutralises inflation's negative effect on the present value of average profits per unit (see equation (2.8) and the discussion of equation (2.10) in section 2.2.2.2). Under limit pricing, this is not possible. The average profit per piece will therefore decrease in ψ because while the price is fixed under price rigidity, unit cost grows at the rate ω which at steady state equals the money growth rate. In comparison to the monopoly case, this negative unit-profit effect creates an

⁴Firms producing qualities k_{j-n} with n > 1 could charge a price $P_{k_j-n} = \frac{P_{k_j}}{q^n} < w(\tau)$ such that they would exit the market.

additional channel for money to have real effects.

In equilibrium, the firm's market value must nevertheless be positive, i.e. $\frac{q}{\chi+\beta} > \frac{1}{\chi+\beta-\omega}$ is a necessary condition for equilibrium. Assumption (4.10) is a sufficient condition for this inequality to hold.

4.2.3 Labour demands and labour market equilibrium

The structure of the intermediate sector's labour demand remains unchanged: Since the limit price chosen by firms readjusting their prices grows at rate ω as does the optimal monopoly price, the average price at time τ charged by intermediate producers is still given by $\left(\frac{\mu+\beta}{\mu+\beta-\alpha\omega}\right)^{-\frac{1}{\alpha}}$ times the price chosen at time τ . Thus, the only change is that the time τ limit price $p^{\lim}(\tau)$ replaces the optimal monopoly price $p^*(\tau)$ in formula (2.11).

The demand for research labour is still given by (2.19). Inserting the intermediate sector's labour, this gives us labour market equilibrium:

$$\frac{Y(\tau)}{Q(\tau)} = \frac{L - \mu \lambda \frac{q^{\alpha-1}}{\alpha-1}}{\left[\frac{p^{\lim(\tau)}}{P(\tau)Q(\tau)} \left(\frac{\mu+\beta}{\mu+\beta-\alpha\omega}\right)^{-\frac{1}{\alpha}}\right]^{-\alpha}}$$

As already explained in sections 2.4 and 2.2.3, the *level* of the mark-up charged by intermediate goods firms is irrelevant for price dispersion. Therefore, using that analogously to Appendix 2.A.3 we can rewrite the current relative limit price in efficiency units as $\frac{p^{\lim}(\tau)}{P(\tau)Q(\tau)} = \left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\omega}\right)^{\frac{1}{\alpha-1}}$, the price dispersion term is unchanged relative to the model under monopoly pricing:

$$\frac{Y(\tau)}{Q(\tau)} = \frac{L - \mu \lambda \frac{q^{\alpha-1}}{\alpha-1}}{\left[\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\omega} \right)^{\frac{1}{\alpha-1}} \left(\frac{\mu+\beta}{\mu+\beta-\alpha\omega} \right)^{-\frac{1}{\alpha}} \right]^{-\alpha}}$$
(4.5)

4.2.4 The zero-profit-condition in general equilibrium

Inserting the unchanged razor's edge condition (2.18), the equations for the firms's market value (4.4) and the size of the final good sector (4.5) into the unchanged zero profit condition (2.17) from the monopoly pricing case, we

Chapter 4. Limit Pricing

have

$$\frac{\left(\alpha-1\right)\left(\frac{L}{\lambda}-\mu\frac{q^{\alpha-1}}{\alpha-1}\right)\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\omega}\right)^{\frac{-\alpha}{\alpha-1}}}{\left[\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\omega}\right)^{\frac{-1}{\alpha}}\left(\frac{\mu+\beta}{\mu+\beta-\alpha\omega}\right)^{-\frac{1}{\alpha}}\right]^{-\alpha}}\frac{\chi+\beta+(\alpha-1)\omega}{\chi-(1-\alpha)\omega}\left(\frac{q}{\chi+\beta}-\frac{1}{\chi+\beta-\omega}\right)=1$$

$$(4.6)$$

4.3 Existence and uniqueness of steady state equilibrium under limit pricing

First, using $\chi = i + \mu - \alpha \pi - \gamma$, $i = r + \pi$, $\pi = \psi - \gamma$ and the unchanged Euler equation from the household's problem (2.31) and rearranging, we rewrite the zero profit condition (4.6) as

$$\frac{\left(\alpha-1\right)\left(\frac{L}{\lambda}-\mu\frac{q^{\alpha-1}}{\alpha-1}\right)\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\omega}\right)^{\frac{-\alpha}{\alpha-1}}}{\left[\left(\frac{\mu+\beta}{\mu+\beta-(\alpha-1)\omega}\right)^{\frac{1}{\alpha-1}}\left(\frac{\mu+\beta}{\mu+\beta-\alpha\omega}\right)^{-\frac{1}{\alpha}}\right]^{-\alpha}}\left(\frac{q\left(\rho+\beta+\tilde{\eta}\mu-\alpha\psi\right)}{\rho+\beta+\tilde{\eta}\mu-(\alpha-1)\psi}-1\right)\right) = \left(\rho+\tilde{\eta}\mu\right)\frac{\rho+\beta+\tilde{\eta}\mu-\alpha\psi}{\rho+\beta+\tilde{\eta}\mu}$$
(4.7)

where $\tilde{\eta} = \left[1 + (\eta - 2 + \alpha) \frac{q^{\alpha-1} - 1}{\alpha - 1}\right]$. We show that equation (4.7) has a unique positive solution for μ under the following set of assumptions:

$$\frac{L}{\lambda} > \frac{\rho}{(\alpha - 1)(q - 1)} \tag{4.8}$$

$$\psi < \frac{\rho + \beta}{\alpha + (q-1)^{-1}} \tag{4.9}$$

$$\psi < \frac{\beta}{\alpha} \frac{\frac{q^{\alpha-1}}{\alpha-1}}{\frac{q^{\alpha-1}}{\alpha-1} + \frac{L}{\lambda}\frac{1}{\beta}}$$
(4.10)

$$\psi < \frac{1}{2}a - \left\{\frac{1}{4}a^2 - \frac{\left(\rho + \beta\right)\beta\left[\left(\alpha - 1\right)\frac{L}{\lambda}\left(q - 1\right) - \rho\right]}{\left(\alpha - 1\right)\alpha\left[\frac{L}{\lambda}\left[\left(q - 1\right)\left(\alpha - 1\right) + q\right] - \frac{\rho\beta}{\rho + \beta}\right]}\right\}^{1/2}$$
(4.11)

where a in Condition (4.11) is given by $a = \frac{\frac{\alpha}{\alpha-1}\beta\left[\frac{L}{\lambda}(\alpha-1)(q-1)-\rho\right] + (\rho+\beta)\left[\frac{L}{\lambda}(\alpha-1)(q-1)-\frac{\rho\beta}{\rho+\beta}\right] + \frac{L}{\lambda}[(q-1)(\rho+\beta)+\beta]}{\alpha\left[\frac{L}{\lambda}[(\alpha-1)(q-1)+q]-\frac{\rho\beta}{\rho+\beta}\right]}.$ Condition (4.8) replaces the no-growth-trap condition (2.37) that is required to hold under monopoly pricing. Similarly, the upper bound on money growth under monopoly pricing (2.38) is replaced by conditions (4.9)-(4.11). Condition (4.9) is sufficient for the market value of the firm, and thus, the return to R&D, to be positive in spite of money's influence on the present value of the average profit per unit. Condition (4.10) is sufficient for the LHS of equation (4.7) to be concave in μ . Conditions (4.8) and (4.11) are jointly sufficient for the value of the LHS of equation (4.7) to exceed that of the RHS for $\mu \to 0.^{6}$ In numerical examples, the restrictions on ψ turn out to be somewhat more restrictive than their equivalent in the monopoly pricing case.⁷

Since the form of the LHS-curve and the RHS-curve from equation (4.7) is qualitatively equivalent to the monopoly case, we use figure (2.2) from the monopoly pricing case in the ensuing discussion.

Proposition 5 Under conditions (4.8)-(4.11), the economy under limit pricing has a unique steady state equilibrium with $\mu > 0$.

The proof can be found in the Appendix to this chapter.

4.4 Comparative statics and a calibrated example

For the discussion of comparative statics, refer back to equation (4.7). We first discuss comparative statics with respect to the money growth rate and then turn to the effect of an increase in the degree of price rigidity before presenting a calibrated example.

⁵Condition (4.11) defines a positive upper bound on ψ whenever the term in curly brackets is non-negative. When it is negative, the restriction is non-binding.

⁶Besides the solution given in (4.11), this inequality also holds for $\psi > \psi_1$, where ψ_1 is given by the term of the RHS of inequality (4.11), with a plus in front of the term in brackets instead of a minus. Yet these solutions are negligible since they violate condition (4.10).

⁷See section 4.4.3.

4.4.1 Unit profit effect reinforces negative net effect of money growth on real growth

On the RHS of equation (4.7), we have the firm's compound discount rate which is unchanged relative to the case with monopoly pricing. An increase in money growth reduces the compound discount rate via the already discussed mark-up erosion effect, raising the incentive to innovate and causing an upward shift of the LHS-curve in figure 2.2. The LHS of the equation reflects the firm's instantaneous profits. The price dispersion effect and the initial relative price effect of money growth which were discussed for the monopoly case in section (2.4) persist in unchanged form. Additionally, the new term $\left(q \frac{\rho + \beta + \tilde{\eta} \mu - \alpha \psi}{\rho + \beta + \tilde{\eta} \mu - (\alpha - 1)\psi} - 1\right)$ can be interpreted as the present value of the firm's average profit per unit divided by the present value of its average unit cost. An increase in money growth has an additional negative effect on the firm's instantaneous profit via this measure of profit per unit:⁸ Since limit pricing prevents firms from choosing a mark-up that would offset the effect of money growth ψ , the profit per unit term is reduced by an increase in ψ because while costs per unit grow at rate ω where $\omega = \psi$ at steady state, revenue per unit is fixed under price rigidity.⁹ Thus via this reduction of instantaneous profits, an increase in ψ ceteris paribus lowers the incentive to innovate, which ceteris paribus translates into a downward shift of the LHScurve. This reinforces the downward shift of the LHS-curve that an increase in ψ causes via the price dispersion effect and the initial relative price effect.¹⁰ Since the net effect of an increase in money growth ψ on economic growth γ was already negative under monopoly pricing and limit pricing causes an

⁸Note that we are slightly simplifying in stating the term represents the present value of average profits per unit since the current real wage, to which unit profit is proportional, has already cancelled out against the cost of research in the zero profit equation. Yet our interpretation is correct insofar as the negative effect is indeed caused by the difference in growth rates between unit revenues and unit costs. Note also that the unit profit term is a *weighted* average of unit profits where the discount factor and the growth rate of demand for the good determine the weighting factor.

 $^{{}^{9}\}frac{\partial(q\frac{\rho+\beta+\tilde{\eta}\mu-\alpha\psi}{\rho+\beta+\tilde{\eta}\mu-(\alpha-1)\psi}-1)}{\partial\psi} = -q\frac{\rho+\beta+\tilde{\eta}\mu}{[\rho+\beta+\tilde{\eta}\mu-(\alpha-1)\psi]^{2}} < 0.$ Note that at $\psi < 0$, profit per unit increases over time since wages grow at rate ψ while the price is fixed. An increase in ψ here *decreases* this effect thus lowering the present value of average unit profit.

 $^{^{10}}$ See section 2.4.2 of chapter 2.

additional negative effect, it is intuitive that the total effect of an increase in money growth ψ on the research intensity μ and economic growth γ is negative under limit pricing:

Proposition 6 For $\beta < \infty$, an increase in the money growth rate ψ lowers the research intensity μ and real growth rate γ .

The proof can be found in the Appendix.

Note also that for $\beta \to \infty$, all effects of money growth vanish, such that money is superneutral in the model with flexible prices regardless of whether monopoly pricing or limit pricing prevails.

4.4.2 Effect of price rigidity on growth depends on money growth rate

Unsurprisingly, the sign of the effect of price rigidity on growth is unchanged:

Proposition 7 For $\psi > 0$ ($\psi < 0$), a decrease in β , i.e. an increase in the level of rigidity, decreases (increases) research intensity μ and the real growth rate γ .

The proof can be found in the Appendix. A reduction in the level of rigidity that acts like a reduction in the absolute value of the money growth rate ψ is conducive (detrimental) to growth when an increase in the absolute value of ψ decreases (increases) growth at $\psi > 0$ ($\psi < 0$).

4.4.3 Calibrated example

We use the baseline example from the calibration given in table (2.1) of the monopoly pricing case in section (2.5) but lower the parameter q indicating the size of improvements brought about by innovations to q = 1.1. With this change, we have that in the baseline case with $\psi = 0.055$, the mark-up is 14.9%. At all examined money growth rates, conditions (4.1) and (4.2) hold such that limit pricing prevails. We find that the restrictions on the money growth rate imposed by conditions (4.9)-(4.11) are rather more restrictive than in the case with monopoly pricing: We can only investigate money growth rates lower than 7.4 percent in our baseline example.¹¹

This reflects that the negative effect of money growth ψ on economic growth γ is strongly reinforced by the effect on the firm's profit per unit: While under monopoly pricing, increasing ψ from 0% to 10% reduced growth from 2.09% to 1.87% (0.22 percentage point reduction), increasing ψ from 0% to 5% already entails a reduction of the real growth rate from 3.22% to 2.15%, i.e. growth is reduced by 1.07 percentage points. At the maximum money growth rate of 7.4% implied by condition (4.10), the growth rate is 1.24%. Clearly, the effect of money growth is unrealistically large given empirical estimates.

Accordingly, the effect of an increase in price rigidity is more sizeable than under monopoly pricing: Decreasing β from 1.8 to 1.6 (i.e., increasing the average period where prices are fixed in a sector from 6.2 to 7.0 months) at $\psi = 0.055$ reduces growth by 0.2 percentage point from 2% to 1.80%, while the corresponding decline was 0.03 percentage point under monopoly pricing.

4.5 Conclusion

It has been shown that under additional restrictions on ψ , a unique steady state equilibrium under limit pricing exists. The additional restriction of the money growth rate is necessary because under limit pricing, the innovating firm cannot choose a forward-looking mark-up to correct the erosion of its profit per unit through money growth. This additional negative effect of money growth on the new firm's profits and the incentive to innovate leads to a stronger reduction of economic growth through a given increase in ψ than in the case of monopoly pricing. Thus high inflation is especially damaging to growth when the innovation size is small given the intensity of competition as measured in the elasticity of substitution between intermediate goods.

¹¹The maximum money growth rates implied by conditions (4.9) and (4.11) are nearly identical at 9.1% and 8.9%, respectively. Condition (4.10) is the most restrictive and implies a maximum money growth rate of 7.4% in the baseline example.

4.A Appendix: Proofs of the propositions

Proof of Proposition 5. Consider equation (4.7). Condition (4.10) ensures that the LHS of the equation is concave in μ for $\psi > 0$ $\left(\frac{\partial^2 LHS}{\partial \mu^2}\right) = -\frac{2\alpha\psi}{(\mu+\beta)^2} \left[q^{\alpha-1} + \frac{(\alpha-1)\frac{L}{\lambda} - \mu q^{\alpha-1}}{\mu+\beta}\right] < 0$ for $\psi > 0$), while the RHS of the equation increases in μ at an increasing rate for $\psi > 0$ $\left(\frac{\partial RHS}{\partial \mu}\right) = \tilde{\eta} \frac{\rho+\beta+\tilde{\eta}\mu-\alpha\psi}{\rho+\beta+\tilde{\eta}\mu} + (\rho+\tilde{\eta}\mu)\tilde{\eta}\frac{\alpha\psi}{(\rho+\beta+\tilde{\eta}\mu)^2} = \frac{\tilde{\eta}\left[(\rho+\beta+\tilde{\eta}\mu)^2 - \alpha\psi\beta\right]}{(\rho+\beta+\tilde{\eta}\mu)^2} > 0$ for all admissible ψ , $\frac{\partial^2 RHS}{\partial \mu^2} = \psi \frac{2\tilde{\eta}^2\alpha\beta}{(\rho+\beta+\tilde{\eta}\mu)^3} > 0$ for $\psi > 0$).¹² At $\psi \le 0$, the LHS of equation (4.7) decreases in μ $\left(\frac{\partial LHS}{\partial \mu} = LHS \left\{ \frac{-\frac{q^{\alpha-1}}{(L}}{(\frac{L}{\lambda} - \mu \frac{q^{\alpha-1}}{\alpha-1})} + \frac{\alpha\psi}{(\mu+\beta)(\mu+\beta-\alpha\psi)} + \frac{\tilde{\eta}q \frac{\psi}{(\rho+\beta+\tilde{\eta}\mu-(\alpha-1)\psi)^2}}{(q\frac{\rho+\beta+\tilde{\eta}\mu-\alpha}{\rho+\beta+\tilde{\eta}\mu-(\alpha-1)\psi}-1)} < 0$ for $\psi \le 0 \right\}$) while the RHS increases in μ . Furthermore, it is trivial to show that $\lim_{\mu\to\infty} LHS = -\infty < \infty = \lim_{\mu\to\infty} RHS$ and conditions (4.8) and (4.11) ensure that the value for $\lim_{\mu\to0} LHS > \lim_{\mu\to0} RHS$ where LHS and RHS still refer to equation (4.7). Thus the two functions always have a unique

intersection with $\mu > 0$ under conditions (4.8)-(4.11).

Proof of Proposition 6. Consider equation (4.7). To see that $\frac{d\mu}{d\psi} = -\frac{\partial LHS}{\partial \psi} - \frac{\partial RHS}{\partial \psi} < 0$, first refer to figure 2.2 to see that given assumptions (4.8) and (4.11) and concavity of the LHS-curve, the latter's slope is always smaller than that of the RHS-curve at the equilibrium $\left(\frac{\partial LHS}{\partial \mu} - \frac{\partial RHS}{\partial \mu} < 0\right)$. Further, $\frac{\partial LHS}{\partial \psi} - \frac{\partial RHS}{\partial \psi} = -\frac{(\rho + \tilde{\eta}\mu)(\rho + \beta + \tilde{\eta}\mu - \alpha\psi)}{\rho + \beta + \tilde{\eta}\mu} \left\{ \frac{\frac{q(\rho + \beta + \tilde{\eta}\mu)}{(\rho + \beta + \tilde{\eta}\mu - (\alpha - 1)\psi)^2}}{q + \frac{\rho + \beta + \tilde{\eta}\mu - \alpha\psi}{\rho + \beta + \tilde{\eta}\mu - (\alpha - 1)\psi} - 1} + \frac{\alpha [\rho + (\tilde{\eta} - 1)\mu]}{(\rho + \beta + \tilde{\eta}\mu - \alpha\psi)(\mu + \beta - \alpha\psi)} \right\} < 0$: The fraction in front of the square brackets is the positive compound discount rate, further by assumption (4.9) $q \frac{\rho + \beta + \tilde{\eta}\mu - \alpha\psi}{\rho + \beta + \tilde{\eta}\mu - (\alpha - 1)\psi} - 1 > 0$, while $1 < \tilde{\eta} = \left(1 + (\eta + \alpha - 2) \frac{q^{\alpha - 1} - 1}{\alpha - 1}\right)$ given $\alpha > 1$, $\eta \ge 1$ and $\beta > \alpha\psi$ given assumption (4.10) ensure that the remaining expressions are positive. Further, since $\gamma = \frac{q^{\alpha - 1} - 1}{\alpha - 1}\mu$, $\frac{d\gamma}{d\psi} = \frac{q^{\alpha - 1} - 1}{\alpha - 1}\frac{d\mu}{d\psi} < 0$.

 $\overline{\frac{12 \frac{\partial^2 LHS}{\partial \mu^2}}{\partial \mu^2}} = -\frac{2\alpha\psi}{(\mu+\beta)^2} \left[q^{\alpha-1} + \frac{(\alpha-1)\frac{L}{\lambda} - \mu q^{\alpha-1}}{\mu+\beta} \right] < 0 \text{ for } \psi > 0. \text{ For } \psi \le 0, \ \frac{\partial LHS}{\partial \mu} < 0 \text{ and} \\ \frac{\partial^2 LHS}{\partial \mu^2} \ge 0 \text{ which together with the condition (4.11) is sufficient for the existence of a unique steady state in the case <math>\psi \le 0$, too.

$$-\frac{\frac{\partial LHS}{\partial \beta} - \frac{\partial RHS}{\partial \beta}}{\frac{\partial LHS}{\partial \mu} - \frac{\partial RHS}{\partial \mu}} \gtrsim 0 \text{ for } \psi \gtrsim 0 \text{ since}$$

$$\frac{\partial LHS}{\partial \beta} - \frac{\partial RHS}{\partial \beta} = \psi \frac{(\alpha - 1)\left(\frac{L}{\lambda} - \mu \frac{q^{\alpha - 1}}{\alpha - 1}\right)}{\frac{\mu + \beta}{\mu + \beta - \alpha \psi}} \frac{\frac{q^{\alpha - 1}}{q \frac{\rho + \beta + \tilde{\eta} \mu - (\alpha - 1)\psi}{\rho + \beta + \tilde{\eta} \mu - (\alpha - 1)\psi} - 1} \\
+ \psi \frac{(\alpha - 1)\left(\frac{L}{\lambda} - \mu \frac{q^{\alpha - 1}}{\alpha - 1}\right)}{\frac{\mu + \beta}{\mu + \beta - \alpha \psi}} \frac{\alpha \left[(\rho + (\tilde{\eta} - 1)\mu)\left(\rho + \beta + \tilde{\eta}\mu + \beta + \mu - \alpha \psi\right)\right]}{(\mu + \beta)\left(\mu + \beta - \alpha \psi\right)\left(\rho + \beta + \tilde{\eta}\mu\right)\left(\rho + \beta + \tilde{\eta}\mu - \alpha \psi\right)} \\$$
(4.12)

depends on the sign of ψ as positive final good production $(\frac{L}{\lambda} - \mu \frac{q^{\alpha-1}}{\alpha-1} > 0)$, $\beta > \alpha \psi$ by condition (4.10), $q \frac{\rho+\beta+\tilde{\eta}\mu-\alpha\psi}{\rho+\beta+\tilde{\eta}\mu-(\alpha-1)\psi} > 1$ by condition (4.9) and $\tilde{\eta} > 1$ ensure that all terms except ψ on the right hand side of equation 4.12 are positive. Further, it was discussed above that $\frac{\partial LHS}{\partial \mu} - \frac{\partial RHS}{\partial \mu} < 0$.

Chapter 5

Money growth, employment and output growth

5.1 Introduction¹

In the preceding chapters, we have abstracted from the effects of money growth on employment. Yet these effects are of interest for two reasons: First, the level of employment is itself one of the most important performance figures of the economy and sustaining a high level of employment is a major goal of economic policy. Secondly, if the level of employment has a scale effect on an economy's output growth rate, the inflation-employment relationship is of additional interest for our analysis of the long-run effects of money growth on output growth.

As reported in the introductory chapter of this thesis, the mainstream view is that although inflation does raise employment in the short-run, the Phillips curve is vertical in the long run. However, the survey of the literature on endogenous growth and inflation in chapter 1 showed that changes in employment originating from households' substitution between goods and activities are regarded as an important transmission channel for the influence of inflation on long-run growth. Further, the literature survey showed that a small literature has recently emerged in the context of the New Keynesian

¹Parts of this chapter are based on Funk and Kromen [2006].

business cycle model which analyses how nominal price rigidity might lead to a non-vertical Phillips curve in the long run.

In the same spirit, we now endogenise labour supply in our Schumpeterian growth model with price rigidity in order to study the effect of money growth on the level of employment and the latter's interaction with growth. In order not to complicate the analysis unduly, labour supply is endogenised in a very simple and straightforward way: We assume the existence of a labour supply function that increases in the wage. The positive correlation between wages and employment implied by this shortcut can be justified e.g. as the outcome of the optimisation of a central labour union which sets the wage and whose wage claims are less ambitious (or less effective) when unemployment is high than when it is low. In addition to keeping the model tractable, endogenising labour supply in this way – rather than introducing leisure in the household's utility function as is usually done in the New Keynesian business cycle model – has the additional benefit of allowing us to discuss unemployment that is involuntary for the individual worker rather than being his optimal choice.

To facilitate the analysis of the model with endogenous labour supply, we slightly modify the structure of the model: In the preceding chapters, labour was used as an input both in the production of intermediate goods and in the R&D sector. We now assume that labour is only used in the final goods sector where it is combined with intermediate goods in the production of output, implying that there is a single aggregate wage. Labour being used in the representative perfectly competitive final good firm is more readily compatible with our underlying story of one central union than labour being employed in the intermediate goods or R&D sectors where – due to price rigidity and the different positions of goods and the corresponding firms on the quality ladder – the economic conditions faced by individual firms differ. Therefore, the final good is now used as an input in the intermediate goods sector and the R&D sector.

In our analysis of the interaction of money growth, the employment level and the output growth rate at steady state, we find that the endogeneity of labour matters for the inflation-growth relationship whereas the inflationemployment relationship is approximately invariant to changes in the output growth rate. Higher employment raises growth because the production technology is such that the marginal productivity of intermediate goods in final good production increases in the amount of labour used in production. The higher productivity of intermediate goods raises profits to be made with their sales and thus raises the incentive to invent improved ones. Since the growing quality of intermediate goods resulting from R&D activities is the engine of growth, employment thus raises growth in the model.

Conversely, we find that employment is approximately unchanged by variations in the economy's growth rate or, equivalently, research intensity: The level of employment is influenced by the aggregate degree of rigidity or flexibility in prices and the degree of price flexibility is in turn influenced by the research intensity since every firm entering the market with an innovative good gets to set a new price. However, innovations are rare events compared to the frequency of price adjustment implied by the Poisson parameter β , so that the total degree of price flexibility is approximately unchanged by variations in the research intensity - and hence, so is the level of employment.

Turning to the effect of money growth on employment, employment and the level of output in efficiency units are hump-shaped functions of the money growth rate which peak at a money growth rate associated with a positive inflation rate. Labour's marginal productivity in final good production increases in the total amount of intermediates used in production and on the efficiency with which they are used. The nonlinearity of the employmentmoney growth relationship is therefore due to the interaction of the money growth rate's effects on the efficiency and on the total amount of intermediate goods usage in production. We now give a short intuition for these effects.

Any increase in the absolute value of the money growth rate that raises the absolute value of inflation causes a reduction in production efficiency. This is because an increase in the absolute value of inflation, which is the growth rate of marginal cost and hence, of the optimal price, will under rigidity lead to an increase in *relative price dispersion* concerning intermediate goods. Price dispersion in turn distorts demand for intermediate goods towards those with low prices. As first shown in chapter 2, given that intermediate goods are imperfect substitutes, an increase in the distortion of quantities causes a decrease in production efficiency. This in turn reduces labour's productivity and hence, the wage and employment. Via this effect, an increase in the money growth rate raises (lowers) employment if it raises (lowers) the absolute value of the inflation rate.

At the same time, an increase in the money growth rate also has an impact on employment via its effect on the average mark-up charged by intermediate goods producers, which in turn determines the total amount of intermediates used in final good production and hence, the marginal productivity of labour. Money growth influences the *average mark-up*, i.e. the average monopolistic distortion present in the economy, via two channels: First, an increase in money growth and inflation raises marginal cost while prices are fixed under rigidity, eroding effective mark-ups. Second, in anticipation of this effect, the initial mark-ups set by firms increase in inflation, which tends to increase the average mark-up.

As a result of the interaction of the relative price dispersion effect and the average mark-up effect of an increase in money growth, employment increases in money growth at negative and small positive inflation rates and decreases in the money growth rate at high inflation – that is, there is a nonlinear long-run Phillips curve. In calibrated examples, the range of positive inflation rates where the average mark-up effect is positive and dominates, i.e. where employment increases in the money growth rate, is very small. At most values, an increase in inflation raises unemployment. Therefore, the model's main messages for monetary policy are that high inflation rates are bad for employment and that very moderate inflation is better than price stability from an employment perspective.

The relationship between the growth rates of money and output given employment is determined by money growth's two countervailing effects on the mark-up charged by an intermediate good firm: Given infrequent price adjustment, a firm's optimal *initial mark-up* increases in money growth and inflation. This is in anticipation of the fact that inflation (deflation) later leads to *mark-up erosion* (appreciation) while the firm's price is fixed. Given that the firm's mark-up thus generally differs from its optimum under flexible prices for any non-zero growth rate of marginal cost (i.e., inflation rate), demand for the good, the profits accruing to an innovator and hence, the incentive to innovate are lowered by non-zero inflation: The incentive to innovate given employment is a hump-shaped function of money growth that reaches its maximum at zero inflation.

Note that this implies that the main message from the previous chapters – that is, given employment, positive inflation is detrimental to growth – is unchanged by our change in the model's structure. The models do, however, differ with respect to the growth consequences of deflation in that in the present context, growth is (given employment) maximised at zero inflation whereas in the model of the preceding chapters growth decreases monotonically in the money growth rate over the entire range of admissible values. A short section is dedicated to the explanation of this difference.

The result that zero inflation maximises growth is no longer valid once it is taken into account that inflation's level effect on employment indirectly has an effect on the growth rate, too: Since the incentive to innovate increases in the level of employment and the latter is maximised at a positive inflation rate, the incentive to innovate and the output growth rate reach their maximum under positive inflation, too. The model is thus compatible with empirical evidence indicating a negative growth effect of medium and high inflation as well as the fact that this effect may only be present once inflation has reached a small positive threshold value. In a realistically calibrated numerical example, the effect of money growth on economic growth is quantitatively in line with the results of the empirical literature.

Summing up, short-term price rigidity allows inflation to affect the longrun levels of employment and output in a way that is consistent with a nonlinear long-run Phillips curve. Taking account of the relationship between money growth and output growth which is influenced by inflation's level effect on employment, any monetary authority interested in fostering economic growth or employment should avoid high inflation rates by choosing a money growth rate that leads to very moderate but positive inflation.

The remainder of this chapter is organised as follows: Section 5.2 presents the model, while section 5.3 discusses the general equilibrium. Comparative statics and a calibrated example are presented in section 5.5 and section 5.6 Chapter 5. Money growth, employment and growth

concludes.

5.2 The model

5.2.1 Final good sector

With our modification of the sectoral structure of the mode, in the perfectly competitive final goods sector, the economy's final good Y is now produced using labour L and N varieties of differentiated intermediate goods. Intermediate goods continue to be combined according to the constant-elasticity-of-substitution aggregator of Dixit and Stiglitz [1977]:

$$Y(\tau) = AL(\tau)^{1/\alpha} \sum_{j=1}^{N} \left(q^{k_j(\tau)} x_j(\tau) \right)^{(\alpha-1)/\alpha}$$
(5.1)

where L is labour, x_j is the amount of sector j intermediate good used, q^{k_j} is this type's productivity and we assume $\alpha > 1$. We again assume that only the highest quality k_j available of intermediate good j is produced in equilibrium.²

The representative firm's profits are given by

$$\Pi^{Y}(\tau) = P(\tau)Y(\tau) - \sum_{j=1}^{N} p_{j}(\tau)x_{j}(\tau) - w(\tau)L(\tau)$$
(5.2)

where $P(\tau)$ is the final good price, $P_j(\tau)$ is the price charged for one unit of sector j intermediate good and $w(\tau)$ is the nominal wage. The firm's optimal

²We make sure that only the latest quality is available in each sector by assuming that parameters are such that the innovator's monopolistic mark-up makes production unprofitable for the incumbent. Given the steady state mark-up from (5.22), $q > \frac{\alpha}{\alpha-1} \frac{\rho+\beta-(\alpha-1)\psi}{\rho+\beta-\alpha\psi}$ is a sufficient condition. This condition is satisfied at the examined money growth rates in our calibrated examples. An analysis of the equilibrium under limit pricing would again worsen the impact of inflation on profits by removing firms' opportunity to set a forward-looking price but promises no further insights.

demand for labour and for intermediate good j, respectively, are given by

$$\frac{1}{\alpha} \frac{Y(\tau)}{L^{d}(\tau)} = \frac{w(\tau)}{P(\tau)}$$
(5.3)

and

$$x_j(\tau) = \left(\frac{p_j(\tau)}{P(\tau)}\right)^{-\alpha} \left(\frac{\alpha - 1}{\alpha}A\right)^{\alpha} L(\tau) q^{(\alpha - 1)k_j}$$
(5.4)

Optimal demand for the type j intermediate good depends negatively on the type's relative price and positively on its productivity $q^{k_j(\alpha-1)}$ and on employment $L(\tau)$.

5.2.2 Intermediate goods sector

It is assumed that the firm that bought the patent for intermediate good j from the research firm which developed the innovation produces the intermediate good one for one with output:

$$x_j\left(\tau\right) = h_j\left(\tau\right) \tag{5.5}$$

where h_j is the quantity of output used for production. Given the linear production function, the development of marginal cost is given by the development of the economy's output price level $P(\tau)$.

An intermediate good producer's pricing problem The firms producing the N intermediate goods act in an environment of monopolistic competition. That is, they maximise the present value of profits by choosing a price while taking as given the final good sector's demand function (5.4) and Calvo [1983]-type price rigidity. Whenever they have the opportunity to readjust prices, firms choose a price to maximise the expected present value of nominal profits obtained while their price is fixed, which is given by

$$E\left[V\left(p_{j},\tau\right)\right] = \int_{\tau}^{\infty} e^{-\int_{\tau}^{s} \left[i(\theta) + \mu_{k_{j}}(\theta) + \beta\right] d\theta} \left[p_{j} - P(s)\right] x_{j}\left(s\right) ds \qquad (5.6)$$
where $[p_j - P(s)] x_j(s)$ is the firm's profit at time *s* and *i* is the nominal interest rate.³ Remember that the term $e^{-\int_{\tau}^{s} [i(\theta) + \mu_{k_j}(\theta) + \beta] d\theta}$ is the discount factor which is adjusted for the probability of obsolescence facing the firm in two different ways: Firstly, $e^{-\int_{\tau}^{s} \beta d\theta}$ represents the probability of not receiving a price setting signal before time *s* in the future. Secondly, the research intensity μ_{k_j} in research firm *j* determines the intermediate firm's probability $e^{-\int_{\tau}^{s} \mu_{k_j}(\theta) d\theta}$ of not having being replaced by a successful innovator by time *s*. Since profits accruing after either of these two events occurs are irrelevant for the firm's pricing decision at time τ , discounting of future profits is the stronger, the higher β and μ_{k_j} .

5.2.3 Patents and the R&D sector

We continue to assume free entry to the research sector. Due to our modification of the sectoral structure, for each small research firm j, the value $\mu_{k_j}(\tau)$ of the Poisson parameter governing the probability of making an innovation now depends linearly on the amount of *final good* used, $z_j(\tau)$, for a given quality rung k_j (i.e., current position of sector j):

$$\mu_{k_j}(\tau) = \phi(k_j(\tau))z_j(\tau) \tag{5.7}$$

Sector j research firm's expected profit at time τ is given by the expected revenue

 $\mu_{k_j}(\tau) E\left[V_{k_j+1}(\tau) | t_{k_j} = \tau\right]$, where $E\left[V_{k_j+1}(\tau) | t_{k_j} = \tau\right]$ is the expected present value at market entry of all future profits accruing to a potential producer of the new intermediate good, as given in equation (5.24), minus the input cost $P(\tau)z_j(\tau)$.

Due to the free entry assumption, firm j's expected profit is zero at every instant which using (5.7) implies that

$$\phi(k_j(\tau))E\left[V_{k_j+1}\left(\tau\right)\left|t_{k_j}=\tau\right]-P(\tau)=0$$
(5.8)

³A constant from the integration of the probability distribution of the price reset signal has already been eliminated.

holds for all active research firms.

The standard knife-edge specification for $\phi(k_j(\tau))$ makes sure that the optimal research intensity μ can be constant and independent of a sector's position and which implies the existence of spillovers in research. Specifically, the lower the sector's quality level, the easier is making an innovation:

$$\phi(k_j(\tau)) = \frac{1}{\lambda} q^{-(\alpha - 1)(k_j + 1)}$$
(5.9)

where $1/\lambda$ is the productivity of labour in research.

5.2.4 Public Sector

The specification of the public sector is unchanged: The state does not levy taxes or issue bonds. Its only policy instrument is the money supply, $M^s(\tau)$ which is perfectly controlled by an independent central bank by setting the constant exogenous money growth rate ψ :

$$\frac{M^s(\tau)}{M(\tau)} = \psi \tag{5.10}$$

All revenue from money creation is allocated to households in form of a lump-sum cash transfer, $T(\tau)$

.

$$M^s(\tau) = T(\tau) \tag{5.11}$$

There is no government spending apart from the transfer of seigniorage to households.

5.2.5 Consumption and money demand

Since we do not introduce a consumption-leisure choice at the household level, households' consumption and money demand decisions are unchanged. Remember that the representative infinitely lived household maximises the discounted present value of his lifetime utility flows, where $\rho > 0$ is the discount factor. Households derive utility both from consumption $c(\tau)$ of the economy's final good and from holding real balances $m(\tau) = \frac{M(\tau)}{P(\tau)}$. Specifically, the household maximises

$$U = \int_{s=0}^{\infty} e^{-\rho s} \frac{(c(s)^{1-\theta} m(s)^{\theta})^{1-\eta} - 1}{1-\eta} ds$$
(5.12)

with $\eta > 1^4$, $\theta \in [0, 1)$ subject to his budget constraint given labour:

$$\dot{v(\tau)} = \frac{w(\tau)}{P(\tau)} L(\tau) + \frac{T(\tau)}{P(\tau)} + r(\tau)v(\tau) - c(\tau) - [\pi(\tau) + r(\tau)]m(\tau) \quad (5.13)$$

where v is the real value of the household's monetary and non-monetary wealth, $\frac{w}{P}L$ is the household's real wage income from being employed $L \leq \overline{L}$ hours, $\frac{T}{P}$ is the real value of the transfer received from the government, and r is the real interest rate which is paid on the firms real holdings of shares in investment funds that finance R&D firms' activities.

The unchanged first-order conditions of the household's maximisation problem are:

$$\frac{\theta}{1-\theta}\frac{c(\tau)}{m(\tau)} = r(\tau) + \pi(\tau)$$
(5.14)

$$\left[\eta + \theta \left(1 - \eta\right)\right] \frac{\dot{c(\tau)}}{c(\tau)} - \theta (1 - \eta) \frac{\dot{m(\tau)}}{m(\tau)} = r\left(\tau\right) - \rho \tag{5.15}$$

Equation (5.14) is a static efficiency condition requiring that the ratio of marginal utilities from money holdings and consumption equal their cost ratio, where the opportunity cost of holding cash is the nominal interest rate $i(\tau) = r(\tau) + \pi(\tau)$. Equation (5.15) governs the utility-maximising allocation of the household's resources over time and will in steady state equilibrium reduce to the familiar Ramsey rule.

5.2.6 Labour supply

Labour supply is introduced in the simplest possible way as an *exogenously* given function $L^{s}(\widetilde{w})$ of real wages per efficiency unit $\widetilde{w}(\tau) = \frac{w(\tau)}{P(\tau)Q(\tau)}$,

⁴This is again sufficient for the transversality condition $\lim_{t\to\infty} \xi_t v_t e^{-\rho t} = \lim_{t\to\infty} e^{-rt} v_t = 0$ to hold.

where $L^{S}(\widetilde{w})$ is strictly increasing in \widetilde{w} from $L^{s}(0) = L^{min} > 0$ to $\overline{L} = \lim_{\widetilde{w}\to\infty} L^{s}(\widetilde{w}).^{5}$

For the sake of concreteness we assume that

$$L^{s}(\widetilde{w}) = \overline{L}\left(1 - \frac{e^{-\sigma\widetilde{w}}}{2}\right)$$
(5.16)

where $\overline{L} > 0$ is the maximal employment (full employment), $L^{min} = \overline{L}/2$ and $\sigma > 0$ is a parameter reflecting the reactiveness of employment with respect to the wage per efficiency unit \widetilde{w} where at time τ we have $\widetilde{w}(\tau) = w(\tau)/(P(\tau)Q(\tau))$. L^s will be constant in steady state equilibrium where $\widetilde{w}(\tau)$ is constant. The strength of labour supply's reaction with respect to the wage depends on the parameter σ .

5.3 Steady state equilibrium

We now analyse the model's general equilibrium restricting our attention to Rational Expectations steady state equilibria with constant output growth. In doing so, we pay special attention to the partial equilibrium analysis of the labour market equilibrium given the research intensity μ and of the research

Note that only the second interpretation allows us to discuss unemployment that is involuntary for the individual worker.

⁵One way to think about $L^s(\tilde{w})$ is to assume that it results from the utility maximization of households with extremely separable preferences: The household's "worker" maximises a function $v(\tilde{w}_{\tau}l_{\tau}, l_{\tau})$ facing a trade-off between the disutility of too much work and bringing home high labour income $\tilde{w}_{\tau}l_{\tau}$. The household's "shopper" receives $(\tilde{w}_{\tau}l_{\tau})_{\tau\geq 0}$ maximises (5.12) given $\{l_{\tau}\}_{\tau\geq 0}$ since he does not interfere with the "worker's" decision.

Assuming $v_1 > 0$, $v_{11} < 0$, $v_2 \ge 0$ for $L \le L^{min}$ and $v_2 \to -\infty$ for $L \to \overline{L}$, the worker's choice of $L^s(\widetilde{w})$ has the desired form.

Another way to think about the inverse of $L^s(\tilde{w})$ is to assume that wages $w(\tau)$ are set by a centralised labour union. The union's real wage claims per efficiency unit are moderated by a high level of unemployment – leading to a positive relation between wages and employment. This may either reflect the union's genuine interest in low unemployment together with its belief that a moderation of wage claims reduces unemployment or it may directly reflect the waning of the union's power to implement high wages when unemployment rises. Note that in the present setting control over nominal wages $w(\tau)$ in fact allows to control real wages per efficiency unit \tilde{w} and also that the union's belief of a negative short-run relation between $\tilde{w}(\tau)$ (and $w(\tau)$) and employment is warranted.

market equilibrium given employment L. Understanding the corresponding partial equilibrium effects facilitates our ensuing discussion of the joint determination of L and μ under price rigidity and non-zero inflation in general equilibrium.

5.3.1 Households

From the household's static optimality condition (5.14), we have that the growth rates of consumption and real money holdings are equal at steady state equilibrium. Using this in the household's dynamic optimality condition and rearranging yields the familiar Ramsey rule:

$$\frac{c(\tau)}{c(\tau)} = \frac{r-\rho}{\eta} \tag{5.17}$$

5.3.2 Money market equilibrium

Using $\psi - \pi = \frac{m^{s}(\tau)}{m^{s}(\tau)} = \frac{m^{d}(\tau)}{m^{d}(\tau)} = \frac{c(\tau)}{c(\tau)} = \gamma$ at steady state equilibrium,⁶ money market equilibrium is characterised by equality of the inflation rate and the output-growth adjusted money growth rate

$$\pi = \psi - \gamma \tag{5.18}$$

5.3.3 Behaviour of the aggregate quality index $Q(\tau)$ and the growth rate

The change in the final good sector's production technology leads to a slightly modified definition of the economy's aggregate technology index, $Q(\tau)$, as the weighted sum of the productivities $q^{k_j(\tau)}$ associated with each sector's intermediate good

$$Q(\tau) = \sum_{j=1}^{N} q^{(\alpha-1)k_j(\tau)}$$
(5.19)

 $^{^{6}}$ See the household's static optimality condition (5.14) and equation (5.26) in section 5.3.5.

The expected growth rate of the quality index Q at time τ , $E\left[\gamma_Q(\tau)\right]$, can be found by aggregating over j the changes in sector j's quality brought about by an innovation, weighted with the flow probability that an innovation will occur in sector j in the infinitesimal time interval beginning at τ . In steady state equilibrium, this probability will be constant and the same for all sectors, so we set $\mu_{k_j}(\tau) = \mu(\tau) = \mu$. Using the law of large numbers, the expected and actual growth rates of the quality index coincide. Following these steps gives us

$$\gamma_Q = \left(q^{\alpha - 1} - 1\right)\mu\tag{5.20}$$

Since at steady state, the growth rate of output γ again equals the growth rate of the aggregate quality index,⁷ we have

$$\gamma = \left(q^{\alpha - 1} - 1\right)\mu\tag{5.21}$$

5.3.4 Equilibrium in the market for intermediate goods

We now derive the optimal mark-up chosen by profit-maximising firms in the equilibrium under price rigidity. Together with the final sector's demand function (5.4), this allows us to derive the market value of an intermediate goods firm at market entry, which will determine the equilibrium patent price charged by successful R&D firms. We further use the optimal initial markup and (5.4) to find the quantity of intermediates produced in steady state equilibrium.

5.3.4.1 Optimal price at steady state equilibrium

We find the optimal price for an intermediate goods firm that first sets or readjusts its price by maximising the expected value of profits given in (5.6) with respect to the price p_j subject to the final good producing firms' demand function (5.4). Using that at steady state, the price level $P(\tau)$ grows at rate π , and the research intensity μ is equal for all sectors and constant leads to

⁷See equation (5.27) in section 5.3.5.

the following expression for the optimal price at time τ :⁸

$$p^{*}(\tau) = \frac{\alpha}{\alpha - 1} \frac{r + \mu + \beta - (\alpha - 1)\pi}{r + \mu + \beta - \alpha\pi} P(\tau)$$
(5.22)

where r is the real interest rate.

The optimal price is again a mark-up over marginal cost, with marginal cost now given by the output price $P(\tau)$. When prices can be constantly readjusted ($\beta \rightarrow \infty$), the optimal mark-up reduces to it flex-price value $\alpha/(\alpha-1)$ from static profit maximisation. Under price rigidity, the markup is higher (lower) than the optimal flex-price mark-up when the growth rate of marginal cost, the inflation rate π , is positive (negative). This higher (lower) mark-up is chosen by the firm in anticipation of the fact that while its price is fixed, the firm's revenue per unit will be constant while unit cost grows at rate π - i.e., inflation (deflation) will lead to erosion (appreciation) of the firm's mark-up. The mark-up is chosen as to offset this effect of inflation on the expected present value of profit per unit. Further, under inflation (deflation) the optimal mark-up ceteris paribus decreases (increases) in the real interest rate r, the research intensity associated with the probability of being replaced by a successful innovator μ and the flow probability of receiving a price resetting signal β . This is because an increase in any of these variables reduces the weight given to future profits relative to current ones, drawing the mark-up closer to the static optimum.

Given that at steady state the inflation rate π ceteris paribus increases in the growth rate of money supply, ψ ,⁹ we therefore have that the initial mark-up increases ceteris paribus with money growth, allowing it to influence real activity.

⁸The maximisation problem has a well-defined solution for $r + \mu + \beta - (\alpha - 1)\pi > 0$. Assumption (5.37) guarantees that this inequality holds in equilibrium.

 $^{^{9}}$ See section 5.3.2.

5.3.4.2 An intermediate good producer's market value at market entry

The market value $E(V_{k_j}(\tau) | t_{k_j} = \tau)$ at the time of market entry t_{k_j} of a new intermediate goods firm j determines the value of the patent for the good developed in the R&D-sector. This market value is the expected present value at time τ of of all future profits of the firm, given that $t_{k_j} = \tau$:

$$E\left(V_{k_{j}}\left(\tau\right)\left|t_{k_{j}}=\tau\right.\right) = \widetilde{A}L\int_{\tau}^{\infty} e^{-(i+\mu)(s-\tau)}\left[p_{j}\left(s\right)-P(s)\right]\left(\frac{p_{j}\left(s\right)}{P(s)}\right)^{-\alpha} ds \quad (5.23)$$

with $\widetilde{A} = \left(\frac{\alpha-1}{\alpha}A\right)^{\alpha} q^{(\alpha-1)k_j}.^{10}$

In the absence of price rigidity when firms can constantly readjust their prices (i.e., $\beta \to \infty$), $p_j(s) = \alpha/(\alpha - 1) P(s)$ so that the innovating firm's market value at market entry is given by

$$E\left(V_{k_{j}}\left(\tau\right)\left|t_{k_{j}}=\tau\right)_{\beta\to\infty}=\frac{\left(\frac{\alpha-1}{\alpha}A\right)^{\alpha}q^{(\alpha-1)k_{j}}\frac{1}{\alpha-1}P\left(\tau\right)\left(\frac{p_{flex}^{*}\left(\tau\right)}{P\left(\tau\right)}\right)^{-\alpha}L\left(\tau\right)}{r+\mu}$$

The real market value $E(V_{k_j}(\tau) | t_{k_j} = \tau)_{\beta \to \infty} / P(\tau)$ can again be interpreted as the properly discounted present value of an infinite stream of profits growing at a constant rate: The numerator of this term corresponds to the firm's instantaneous profit, while the obsolescence-adjusted discount rate is given in the denominator. Since $p_{flex}^*(\tau) / P(\tau)$ and employment L are constant at steady state, the firm's profit growth rate is zero, implying that the discount factor is $r + \mu - 0$.¹¹ Note that instead of increasing in total final good production Y as in chapter 2, the firm's value is proportional to the amount of *labour* L employed in final good production since intermediate goods' productivity increases in L. The firm's value is also proportional to $P(\tau)$ since the price level determines both the firm's revenues and costs.

In the presence of Calvo-type price rigidity, deriving the firm's expected

¹⁰Note that the wage adjusts freely to clear the labour market such that in equilibrium, employment in the final good sector equals the constant labour supply L at all times.

¹¹Remember that the appropriate discount rate for an infinite stream of profits that grows at constant rate x is r - x.

market value at market entry again requires going through a number of steps, resulting in the following equation:¹²

$$E\left(V_{k_j}\left(\tau\right)\left|t_{k_j}=\tau\right.\right) = \frac{\left(\frac{\alpha-1}{\alpha}A\right)^{\alpha}q^{(\alpha-1)k_j}\frac{1}{\alpha-1}P(\tau)\left(\frac{p^*(\tau)}{P(\tau)}\right)^{-\alpha}}{\left(r+\mu\right)\frac{r+\mu+\beta-\alpha\pi}{r+\mu+\beta}}L\left(\tau\right)$$
(5.24)

Equation (5.24) differs from the flex-price market value in two respects: First, as seen in equation (5.22) in the previous section, with positive inflation the initial mark-up $p^{*}(\tau)/P(\tau)$ chosen by the firm under price rigidity is higher than the optimal mark-up under flexibility, $p_{flex}^{*}(\tau)/P(\tau) =$ $\alpha/(\alpha-1)$. This reduces demand for the good (see equation (5.4)) and therefore, the firm's instantaneous profits. Secondly, the discount rate under flexibility $r + \mu$ is replaced by a compound discount rate where the *flex-price* discount rate is corrected with the factor $(r + \mu + \beta - \alpha \pi) / (r + \mu + \beta)$ that consists of the appropriate discount rates for a firm under price *rigidity* for periods where prices can be changed or are fixed, respectively. The discount rate for periods where prices are fixed decreases in inflation. This is because while prices are fixed, the new good's mark-up and relative price erode at rate $-\pi$, which by equation (5.4) leads to a growth rate $\alpha\pi$ of demand for the good. Given positive profits per unit, the rising demand translates into a higher growth rate of the new intermediate firm's profits. Since the discount rate is the obsolescence-adjusted interest minus the profit growth rate, an increase in inflation thus reduces the discount rate for periods where prices are fixed and the compound discount rate.

An increase in the frequency of price adjustment, $\beta + \mu$ reduces the weight given to periods where prices cannot be changed and therefore reduces the necessary correction.¹³

¹²Derivation of the market value is described in more detail in Appendix 5.A.1.

¹³Note that at $\pi < 0$, the flex-price discount rate has to be corrected *upwards* for the negative growth rate of profits in periods where the mark-up appreciates. An increase in β here means that the correction term *rises* to reduce the extent of correction.

5.3.4.3 Intermediate goods production in steady state equilibrium

The final good sector's demand for intermediate goods can now be found by using the final good sector's demand function (5.4) for good j and aggregating over all intermediate goods:¹⁴

$$X(\tau) = \left(\frac{\alpha - 1}{\alpha}A\right)^{\alpha} LQ(\tau) \sum_{j=1}^{N} \left(\frac{p_j(\tau)}{P(\tau)}\right)^{-\alpha} \frac{q^{(\alpha - 1)k_j}}{Q(\tau)}$$

Aggregate demand for intermediate goods grows with the technology aggregate $Q(\tau)$ and depends negatively on the average relative price of intermediate goods. Using that the average price effective at time τ can again be expressed as a weighted average of past optimal prices set by firms at the last (stochastic) point in time *s* where they could change their prices and going through a number of steps, we can rewrite this equation as¹⁵

$$\left(\frac{X(\tau)}{Q(\tau)}\right)^* = \left(\frac{\alpha - 1}{\alpha}A\right)^{\alpha} L\left[\frac{p^*(\tau)}{P(\tau)}\left(\frac{\beta + \mu}{\beta + \mu - \alpha\pi}\right)^{-\frac{1}{\alpha}}\right]^{-\alpha}$$
(5.25)

where the term $\frac{p^*(\tau)}{P(\tau)} \left(\frac{\beta+\mu}{\beta+\mu-\alpha\pi}\right)^{-1/\alpha}$ is the average relative price, or equivalently, the *average mark-up*, which under flexible prices reduces to, $\alpha/(\alpha-1)$. Since both components of the term depend on the inflation rate π , the average mark-up and hence, total demand for intermediate goods can be influenced by monetary policy.¹⁶ The average mark-up increases in the optimal initial mark-up $p^*(\tau)/P(\tau)$ whose determinants were discussed in section 5.3.4.1. At the same time, the influence of past mark-ups on the average mark-up, which as explained above is a weighted average of the current and past values of the optimal mark-up, is captured in the term $\left[(\beta + \mu)/(\beta + \mu - \alpha\pi)\right]^{-1/\alpha}$: It implies that the average mark-up is lower (higher) than the current value

¹⁴Note that due to the linear production function (5.5), the total production of intermediate goods equals both the final goods sector's demand for intermediate goods and the intermediate goods sector's demand for the final good as an input.

¹⁵For details on the derivation of equation (5.25), see Appendix 5.A.2.

 $^{^{16}}$ Details will be discussed in section 5.3.6.1.

under inflation (deflation) because past optimal mark-ups are lower (higher). The weight of past mark-ups decreases in the frequency of price adjustments $\beta + \mu$: The higher the price setting signal or the frequency of market entry with new prices, the closer the average mark-up to its current value.

Since $p^*(\tau)/P(\tau)$ is constant at steady state equilibrium, $X(\tau)$ grows at the same rate as $Q(\tau)$. Bearing in mind that intermediate goods are produced one to one with output, $X(\tau)$ is also the intermediate sector's total demand for output.

5.3.5 Equilibrium in the final good market

Market equilibrium For the final good market to be in equilibrium, households' consumption must equal the difference between total final good production $Y(\tau)$ and the sum of the demands for final good by the intermediate goods and research sectors, which are $X(\tau)$ and $Z(\tau)$, respectively. In efficiency units, this is

$$\left(\frac{c\left(\tau\right)}{Q\left(\tau\right)}\right)^{*} = \left(\frac{Y\left(\tau\right)}{Q\left(\tau\right)}\right)^{*} - \left(\frac{X\left(\tau\right)}{Q\left(\tau\right)}\right)^{*} - \left(\frac{Z\left(\tau\right)}{Q\left(\tau\right)}\right)^{*}$$
(5.26)

Having already determined $(X(\tau)/Q(\tau))^*$ in the last section, we now turn to the steady state value of final good production in efficiency units, $(Y(\tau)/Q(\tau))^*$.¹⁷

Final good production in steady state equilibrium Now that we know both the final good sector's demand function for intermediate goods (5.4) and the optimal price chosen by intermediate goods producers (5.22), we can insert those equations into the final good production function to find that total production is

$$Y(\tau) = AL(\tau)^{\frac{1}{\alpha}} \sum_{j=1}^{N} \left(q^{\alpha k_j(\tau)} \left(\frac{p_j(\tau)}{P(\tau)} \right)^{-\alpha} \left(\frac{\alpha - 1}{\alpha} A \right)^{\alpha} L(\tau) \right)^{\frac{\alpha - 1}{\alpha}}$$

¹⁷The value $(Z(\tau)/Q(\tau))^*$ will be determined in equation (5.31) of section 5.3.7.

Going through similar steps as in the derivation of total intermediate good production (5.25) and some additional steps, this can be rewritten as¹⁸

$$\left(\frac{Y(\tau)}{Q(\tau)}\right)^* = AL(\tau)^{\frac{1}{\alpha}} \left(\frac{X(\tau)}{Q(\tau)}\right)^{\frac{\alpha-1}{\alpha}} \frac{\frac{\beta+\mu}{\beta+\mu-(\alpha-1)\pi}}{\left(\frac{\beta+\mu}{\beta+\mu-\alpha\pi}\right)^{\frac{\alpha-1}{\alpha}}}$$
(5.27)

where the total amount of intermediate goods produced $X(\tau)/Q(\tau)$ is given in equation (5.25). Note that since $X(\tau)/Q(\tau)$ is constant at steady state equilibrium, $Y(\tau)$ grows at the same rate as $Q(\tau)$. Output production in equation (5.27) is the product of two terms: The term $AL(\tau)^{\frac{1}{\alpha}} [X(\tau)/Q(\tau)]^{\frac{\alpha-1}{\alpha}}$ shows production when a total of $X(\tau)/Q(\tau)$ quality-weighted intermediate goods is employed efficiently. In contrast, the term

 $\left(\frac{\beta+\mu}{\beta+\mu-(\alpha-1)\pi}\right) / \left(\frac{\beta+\mu}{\beta+\mu-\alpha\pi}\right)^{\frac{\alpha-1}{\alpha}} \leq 1 \text{ represents the production inefficiency due}$ to price dispersion under price rigidity: When inflation, i.e. the growth rate of marginal cost, is zero, all intermediate goods prices are equal in spite of price rigidity because the optimal price does not change over time. Given equation (5.4), the goods are then demanded in (quality-weighted) equal amounts, which given the constant elasticity of substitution between individual quality-weighted intermediates in the Dixit-Stiglitz final good production function means production is efficient.¹⁹ Any non-zero inflation rate in contrast implies that the optimal price changes over time so that there is dispersion in prices and demanded quantities of intermediates. The production inefficiency term $\left(\frac{\beta+\mu}{\beta+\mu-(\alpha-1)\pi}\right) / \left(\frac{\beta+\mu}{\beta+\mu-\alpha\pi}\right)^{\frac{\alpha-1}{\alpha}}$ consists of the ratio of output actually produced with a given total amount of intermediate goods given current relative prices and output that could be produced with this input spread efficiently over the intermediate goods types.²⁰ Price dispersion and production inefficiency are the more pronounced, the higher the absolute value of the growth rate of optimal prices π , and the higher price rigidity, i.e. the lower $\beta + \mu$.

¹⁸See Appendix 5.A.3.

¹⁹The term $\left(\frac{\beta+\mu}{\beta+\mu-(\alpha-1)\pi}\right) / \left(\frac{\beta+\mu}{\beta+\mu-\alpha\pi}\right)^{\frac{\alpha-1}{\alpha}}$ reaches its maximum value, unity, for $\pi = 0$. ²⁰For details see Appendix 5.A.3.

5.3.6 Labour market equilibrium given the innovation rate μ

By introducing the equilibrium amount of final goods produced (5.27) into the wage equation (5.3), we get the equilibrium real wage in efficiency units $\tilde{w}(\tau) = w(\tau) / [P(\tau)Q(\tau)]$:

$$\widetilde{w}\left(\tau\right) = \frac{1}{\alpha} AL\left(\tau\right)^{\frac{1-\alpha}{\alpha}} \left(\frac{X\left(\tau\right)}{Q\left(\tau\right)}\right)^{\frac{\alpha-1}{\alpha}} \frac{\frac{\beta+\mu}{\beta+\mu-(\alpha-1)\pi}}{\left(\frac{\beta+\mu}{\beta+\mu-\alpha\pi}\right)^{\frac{\alpha-1}{\alpha}}}$$

which inserting the equilibrium amount of intermediate goods produced X/Qfrom equation (5.25), inserting the optimal initial mark-up $p^*(\tau)/P(\tau) = \frac{\alpha}{\alpha-1} \frac{r+\mu+\beta-(\alpha-1)\pi}{r+\mu+\beta-\alpha\pi}$ from equation (5.22), using the Euler equation (5.17), and using equation (5.18) from money market equilibrium and $\gamma = \bar{q}\mu$ can be rewritten as

$$\widetilde{w} = \overline{A} \left[\frac{\rho + \beta + \mu \left(1 + \alpha \overline{q} \right) - \left(\alpha - 1 \right) \psi}{\rho + \beta + \mu \left[1 + \left(1 + \alpha \right) \overline{q} \right] - \alpha \psi} \left(\frac{\beta + \mu}{\beta + \mu - \alpha \left(\psi - \overline{q} \mu \right)} \right)^{-\frac{1}{\alpha}} \right]^{-\left(\alpha - 1\right)} \frac{\frac{\beta + \mu}{\beta + \mu - \left(\alpha - 1\right)\left(\psi - \overline{q} \mu\right)}}{\left(\frac{\beta + \mu}{\beta + \mu - \alpha\left(\psi - \overline{q} \mu\right)} \right)^{\frac{\alpha - 1}{\alpha}}}$$
(5.28)
with
$$\overline{A} = \frac{\left(\frac{\alpha}{\alpha - 1} \right)^{-2\left(\alpha - 1\right)}}{\alpha} A^{\alpha}.$$

5.3.6.1 Properties of the real wage function

From equation (5.28), the steady state real wage in efficiency units is a function of the research intensity μ and of exogenous parameters, in particular of the money growth rate ψ and of the price rigidity parameter β :

$$\widetilde{w}\left(\mu,\psi,\beta\right) \tag{5.29}$$

We now discuss the properties of this function in some detail because equilibrium employment has qualitatively the same properties.²¹ By equation (5.3) the real wage is determined by output per unit of labour $Y(\tau) / [Q(\tau) L(\tau)]$. Thus any influence of parameters on the total input of intermediate goods

 $^{^{21}}$ See section 5.3.6.2.

and on the efficiency with which this amount is used affects the wage and employment. First note that since $Y(\tau) / [Q(\tau) L(\tau)]$ is independent of total employment,²² so is \tilde{w} . This facilitates our analysis considerably.

Wage is a hump-shaped function of money growth ψ The two influences of money growth on the wage via the *average mark-up* and on *price dispersion* make $\tilde{w}(\mu, \psi, \beta)$ a hump-shaped function of money growth. We discuss both influences in turn.

Price dispersion effect As explained in section 5.3.5, any increase in the money growth rate that increases inflation (decreases deflation) raises (lowers) the absolute value of the growth rate of optimal prices and therefore raises (lowers) price dispersion and production inefficiency

 $\frac{\beta+\mu}{\beta+\mu-(\alpha-1)(\psi-\bar{q}\mu)} \left(\frac{\beta+\mu}{\beta+\mu-\alpha(\psi-\bar{q}\mu)}\right)^{-\frac{\alpha-1}{\alpha}}$. Since the productivity of labour increases in the average productivity of intermediate goods, this lowers (raises) the wage.²³

Average mark-up effect As explained in the last part of section 5.3.4.3, total demand for intermediate goods according to equation (5.25) depends negatively on the average mark-up charged by intermediate goods firms, $\frac{p^*(\tau)}{P(\tau)} \left(\frac{\beta+\mu}{\beta+\mu-(\alpha-1)\pi}\right)^{-\frac{1}{\alpha-1}}$ which is altered by an increase in money growth in two ways: Firstly, as discussed in section (5.3.4), an intermediate good firm's optimal *initial mark-up* $p^*(\tau)/P(\tau)$ increases in ψ since the growth rate of marginal cost π ceteris paribus rises in ψ , accelerating (slowing down) the future mark-up erosion (appreciation) under inflation (deflation). At the same time, the weight of past mark-ups in the average mark-up,

 $\{(\beta + \mu) / [\beta + \mu - (\alpha - 1)\pi]\}^{-\frac{1}{\alpha-1}}$, decreases in ψ since the higher inflation (the smaller deflation), the lower are past mark-ups relative to the current one and thus the smaller the average mark-up relative to the current one. A

$$23 \frac{\partial \left[\left(\frac{\beta+\mu}{\beta+\mu-(\alpha-1)\pi}\right) / \left(\frac{\beta+\mu}{\beta+\mu-\alpha\pi}\right)^{\frac{\alpha}{\alpha}} \right]}{\partial \psi} = \frac{-(\alpha-1)\pi \frac{\beta+\mu}{\beta+\mu-(\alpha-1)\pi}}{\left(\frac{\beta+\mu}{\beta+\mu-\alpha\pi}\right)^{\frac{\alpha-1}{\alpha}} [\beta+\mu-(\alpha-1)\pi](\beta+\mu-\alpha\pi)} \lessapprox 0 \text{ as } \pi \gtrless 0$$

²²This can be seen by inserting the value for X/Q from equation (5.25) into equation (5.27).

sufficient condition for the average mark-up to decrease in ψ is $\pi \leq 0.^{24}$ In calibrated examples, the average mark-up is minimised at a small positive inflation rate.

This explains why the wage is maximised at a money growth rate associated with a positive inflation rate despite the negative price dispersion effect of positive inflation:

Net effect of money growth on employment:

Lemma 8 The wage $w(\mu, \psi, \beta)$ is a hump-shaped function of the money growth rate ψ with a maximum at $\psi_1 > 0$ where ψ_1 is given in Appendix 5.A.4. At this unique maximum, the inflation rate $\pi(\psi_1)$ is strictly positive.

The proof to the lemma can be found in Appendix 5.A.4.

Thus holding constant the research intensity, the wage is a hump-shaped function of the money growth rate with its peak at a money growth rate associated with a positive inflation rate.

An increase in price rigidity can increase the wage An increase in β , i.e. a decrease in price rigidity, affects employment via the same channels as does money growth. Through all these channels, the effect of an increase in flexibility is very similar to the effect of a decrease in the *absolute value* of the money growth rate: The fact that prices can be changed more often reduces *price dispersion* and thus increases the productivity of labour and hence, employment. At the same time, the average mark-up depends on β in two different ways: First, the initial mark-up decreases (increases) in β under inflation (deflation) since an increased probability that prices may be readjusted soon draws the mark-up closer to the static optimum. Second, the deviation of the average mark-up from the current optimal mark-up is smaller when β increases since effective prices were on average set more recently. Therefore, an increase in β raises (lowers) the average mark-up under inflation

$${}^{24}\frac{\partial MU^{avg}}{\partial\psi} = \frac{\left(\frac{\beta+\mu}{\beta+\mu-\alpha\pi}\right)^{-\frac{1}{\alpha}}\left[\rho+\beta+\mu(1+\overline{q})\right]}{\{\rho+\beta+\mu[1+(1+\alpha)\overline{q}]-\alpha\psi\}^2} \left\{-\frac{\rho+\beta+\mu(1+\overline{q})-(\alpha-1)\pi}{\rho+\beta+\mu(1+\overline{q})}\frac{\rho+\beta+\mu(1+\overline{q})-\alpha\pi}{\beta+\mu-\alpha\pi}+1\right\} < 0 \text{ for } \\ \pi \le 0 \text{ where } MU^{avg} = \frac{\rho+\beta+\mu(1+\alpha\overline{q})-(\alpha-1)\psi}{\rho+\beta+\mu[1+(1+\alpha)\overline{q}]-\alpha\psi} \left(\frac{\beta+\mu}{\beta+\mu-\alpha\pi}\right)^{-\frac{1}{\alpha}}.$$

(deflation). Thus, the fact that prices can be readjusted more frequently reduces the strength of both price dispersion and the average mark-up effect.

Concerning the net effect of an increase in β , we have the following lemma:

Lemma 9 The wage and employment increase in the price flexibility parameter β under deflation. Under small positive inflation rates, wage and employment decrease in β .

Intuitively, an increase in price rigidity promotes employment under moderate inflation since under these circumstances, an increase in (the absolute value of) money growth and inflation raises employment. Since an increase in price flexibility mitigates the effects of money growth, it reduces employment.²⁵

Wage is approximately unchanged by the research intensity μ The effects of an increase in μ on the wage are qualitatively nearly identical to the effects of an increase in β since an increase in the innovation frequency reduces price rigidity, as does an increase in β .²⁶ Yet since the frequency of innovation μ is small compared to the frequency of Calvo-price adjustments β , its contribution to the degree of price flexibility $\beta + \mu$ is small. Therefore, the elasticities of price dispersion, the initial mark-up and the deviation of the average mark-up from the initial mark-up with respect to μ are very small. In fact, all the aforementioned elasticities with respect to μ go to zero for $\mu/(\mu + \beta) \rightarrow 0$ which holds approximately for all reasonable calibrations so that the effects of an increase in μ on the wage are quantitatively negligible.²⁷

 $^{^{25}}$ It is intuitive that for the same reasons under high inflation rates where an increase in the money growth rate reduces employment, an increase in the price flexibility parameter β raises employment. While we cannot prove this analytically, it is confirmed by all our numerical examples.

²⁶The effects of increases in μ and β are perfectly identical regarding price dispersion and the deviation of the average mark-up from the initial mark-up. In contrast, the initial mark-up decreases in μ not only due to the latter's influence on the degree of price flexibility $\beta + \mu$ but also via its indirect influence via the growth rate γ that increases the real interest rate r and lowers the inflation rate π . See section 5.3.4.1 for a description of the effects of rand π on the initial mark-up. Yet these indirect influences are not important numerically since the elasticity of the initial mark-up with respect to μ vanishes for $\mu/(\mu + \beta) \rightarrow 0$.

²⁷E. g., in the baseline case of our leading example, $\mu/(\mu + \beta) = 0.007$.

Chapter 5. Money growth, employment and growth

5.3.6.2 Equilibrium employment $L(\mu, \beta, \psi)$

Given that labour supply from equation (5.16) increases monotonically in \tilde{w} , the employment function (5.30) preserves the above-discussed properties of the wage function (5.29).

$$L(\mu,\beta,\psi) = L^{s}[\widetilde{w}(\mu,\psi,\beta)]$$
(5.30)

In particular, employment given the innovation rate μ is a hump-shaped function of money growth peaking at a value of ψ associated with a positive inflation rate, may be increased by an increase in rigidity under small positive inflation rates and is approximately invariant to the innovation rate μ .

5.3.7 Research market equilibrium given employment L

5.3.7.1 Equilibrium in the market for patents

The prospect of positive profits in intermediate goods production leads to buyers' competition in the market for patents in the course of which the price is bidden up to the market value of the new firm using the patent, (5.24). Given that research firms charge exactly this price, all new patents will be bought and the market for patents clears.

5.3.7.2 The R&D sector's demand for the final good at steady state equilibrium

The research sector's demand for the final good is found by rearranging (5.7), inserting $\phi(k_j(\tau))$ as defined in equation (5.9) and aggregating over all research firms. In efficiency units, this yields

$$\left(\frac{Z\left(\tau\right)}{Q\left(\tau\right)}\right)^{*} = \lambda \mu q^{\alpha - 1} \tag{5.31}$$

The constant steady state equilibrium demand $\left(\frac{Z(\tau)}{Q(\tau)}\right)^*$ depends on the value of the equilibrium research intensity μ that we determine next.

5.3.7.3 Equilibrium research intensity

Using a new firm's expected market value $E(V_{k_j+1}(\tau) | t_{k_j} = \tau)$ (5.24) and $\phi(k_j(\tau))$ from equation (5.9) in the zero profit condition (5.8) gives us an equation determining the equilibrium research intensity μ which makes current research firms indifferent with regard to the amount of research input used.

$$\frac{L}{\lambda} \frac{\left(\frac{\alpha-1}{\alpha}A\right)^{\alpha} \frac{1}{\alpha-1} P(\tau) \left(\frac{p^{*}(\tau)}{P(\tau)}\right)^{-\alpha}}{(r+\mu) \frac{r+\mu+\beta-\alpha\pi}{r+\mu+\beta}} = P(\tau)$$
(5.32)

Note that consistent with the assumption first made in section 5.3.3, the resulting steady state research intensity μ is the same for all research firms regardless of their sector's current position on the quality ladder.

Further using the optimal initial mark-up $\frac{\alpha}{\alpha-1} \frac{r+\mu+\beta-(\alpha-1)\pi}{r+\mu+\beta-\alpha\pi}$ from equation (5.22), the Euler equation (5.17), the equation relating economic growth to research intensity (5.21), using that equilibrium in the money market implies $\pi = \psi - \gamma$ and rearranging, we get an equation in μ , employment L and the model's parameters:

$$\frac{L}{\lambda} \left(\frac{\alpha}{\alpha-1}\right)^{-\alpha} A^{\alpha} \frac{1}{\alpha-1} \left(\frac{\alpha}{\alpha-1} \frac{\rho+\beta-(\alpha-1)\psi+(\overline{\eta}-\overline{q})\mu}{\rho+\beta-\alpha\psi+\overline{\eta}\mu}\right)^{-\alpha} = \left[\rho+(\eta\overline{q}+1)\mu\right] \frac{\rho+\beta-\alpha\psi+\overline{\eta}\mu}{\rho+\beta+(\eta\overline{q}+1)\mu}$$
(5.33)

where $\overline{q} = (q^{\alpha-1} - 1) > 0, \ \overline{\eta} = [(\eta + \alpha) \overline{q} + 1] \ \text{and} \ \overline{\eta} > \overline{q}.$

Both sides of the equation show the dependence of the optimal research intensity μ on the new firm's value: The LHS of equation (5.33) shows the instantaneous profits for a firm entering the market with a new patent as a function of the research intensity while the RHS represents the compound discount rate for this firm's future profit streams as a function of μ . Figure 5.1 depicts the LHS and the RHS of this equation.

The solution to equation (5.33) is a function of employment, money growth and rigidity:

$$\mu\left(L,\psi,\beta\right)\tag{5.34}$$

Lemma 10 Under conditions (5.35)-(5.37), there is a unique steady state



Figure 5.1: Partial equilibrium research intensity μ given employment L

equilibrium research intensity $\mu(L, \psi, \beta)$ for any $L \in \left(\frac{1}{2}\overline{L}, \overline{L}\right)$.²⁸

$$\frac{1}{2}\frac{\overline{L}}{\lambda} > \frac{\rho\frac{\rho+\beta-\alpha\psi}{\rho+\beta}}{\left(\frac{\alpha}{\alpha-1}\right)^{-2\alpha}A^{\alpha}\frac{1}{\alpha-1}} \left(\frac{\rho+\beta-(\alpha-1)\psi}{\rho+\beta-\alpha\psi}\right)^{\alpha} (5.35)$$

$$-\frac{\overline{q}}{\eta\overline{q}+1}\left(\rho+\beta\right) < \psi < \frac{1}{\alpha}\frac{\left(\rho+\beta\right)\left[2\left(\eta\overline{q}+1\right)+\left(\alpha-1\right)\overline{q}\right]}{\left(2+\frac{\alpha-1}{\alpha}\right)\left(\eta\overline{q}+1\right)+2\left(\alpha-1\right)\overline{q}}$$
(5.36)

$$\psi < \frac{1}{\alpha}\beta \tag{5.37}$$

where $\bar{q} = q^{\alpha-1}-1$. Condition (5.35) ensures that $\lim_{\mu\to 0} LHS > \lim_{\mu\to 0} RHS$ in equation (5.33). It implies that the efficiency weighted labour force cannot be too small. For $\beta < \infty$, conditions (5.36) and (5.37) are jointly sufficient for the LHS of equation (5.33) to be concave in μ , while condition (5.37) and the first inequality in condition (5.36) are sufficient to ensure that the RHS of the equation is convex in μ as depicted in fig. 5.1. Condition (5.36) can always be satisfied since the term to the very left is negative while the expression on the right hand side is positive. Conditions (5.36) and (5.37) imply that for any given β , there exist a lower and an upper bound on the growth rate of money supply ψ compatible with steady state equilibrium.

All conditions are easily satisfied in all our numerical examples.²⁹

 $^{^{28}\}mathrm{All}$ proofs can be found in Appendix 5.A.4.

²⁹In the leading example we introduce in section 5.5, condition 5.35 implies $\frac{\overline{L}}{\lambda} > 2.28$ while we choose $\frac{\overline{L}}{\lambda} = 4.725$. Condition 5.37 is less restrictive than condition 5.36 which

Intuition For intuition concerning the form of the LHS-curve, first note that in the case without price rigidity $(\beta \to \infty)$, the LHS of equation (5.33), which represents the instantaneous real profit associated with the production of the new good, simplifies to the constant $\frac{L}{\lambda} \left(\frac{\alpha}{\alpha-1}\right)^{-2\alpha} A^{\alpha} \frac{1}{\alpha-1}$. For $\beta < \infty$, the curve has a positive slope in μ since as discussed in section 5.3.4.1 the forward-looking initial mark-up chosen by firms under price rigidity decreases in μ .³⁰ Since demand for the new firm's good is inversely related to its mark-up and relative price, an increase in μ increases the instantaneous profits associated with its invention, hence the positive slope of the LHS-curve.

The RHS of equation (5.33) represents the compound discount rate applicable to the new firm's profits. For $\beta \to \infty$, the discount rate reduces to $r + \mu$ which increases linearly in μ since the probability of being replaced increases. Under price rigidity ($\beta < \infty$), this effect of an increase in μ is reinforced through an increase in the correction factor ($\rho + \beta - \alpha \psi + \overline{\eta} \mu$) / [$\rho + \beta + (\eta \overline{q} + 1) \mu$].³¹

Note that given assumptions (5.35)-(5.37) and concavity of the LHScurve, the slope at the steady state equilibrium of the LHS-curve is smaller than that of the RHS-curve. Intuitively, the increase in the discount rate caused by an increase in μ is bigger than the associated increase in instantaneous profits implying that expected profit from an innovation decreases in μ , as in the model without money.

At $\pi < 0$, the rigidity-caused part of the initial mark-up is *smaller* than unity because the mark-up will appreciate under deflation. An increase in μ further decreases the initial mark-up due to the indirect effect that $\pi = \psi - \gamma$ becomes even more negative, such that the future growth rates of revenues and costs diverge even further.

implies $-1.12 < \psi < 0.14$. The upper bound, which corresponds to an inflation rate of $\pi = 12.5\%$, does not restrict our analysis of innovation-driven growth unduly.

³⁰Note that in addition to the direct effect on the mark-up of an increase in the probability of being replaced by an innovator, an increase in μ has several indirect effects on the mark-up through its proportionality to the output growth rate γ and through the latter's effect on the interest rate $r = \rho + \eta \gamma$ and on the inflation rate $\pi = \psi - \gamma$ (see section 5.3.4.1 for an analysis of the influence of r and π on the mark-up). For $\pi > 0$, the net indirect effect is negative and thus reinforces the direct effect of an increase in μ .

³¹This implies that the extent of correction decreases (increases) at $\pi > 0$ ($\pi < 0$) where the correction factor is smaller (bigger) than unity: The main reason is that through its proportionality to γ , an increase in μ lowers inflation (increases deflation) $\pi = \psi - \gamma$, thereby lowering (increasing) the positive (negative) profit growth rate in periods where the erosion (appreciation) of the firm's mark-up through inflation (deflation) leads to an increase (decrease) in demand for the good. Thus the deviation of the profit growth rate from its flex-price value that that requires correction decreases (increases) in μ .

We now discuss the properties of the research intensity function (5.34) with the help of figure 5.1.

5.3.7.4 Standard scale effect of employment L on the innovation rate μ

Lemma 11 The innovation rate $\mu(L, \psi, \beta)$ increases monotonically in L.

An increase in L raises instantaneous profits, shifting up the LHS-curve in figure 5.1. Given that the RHS-curve is unaffected by the change in Land given the curves' shapes, the increase in \overline{L} results in a higher partial equilibrium innovation rate. The positive scale effect on growth of an increase in employment is a is a well-known feature of the underlying real growth model. In general equilibrium, this will allow for additional influences of exogenous parameters on the growth rate via their influence on employment.

5.3.7.5 Innovation rate μ depends negatively on absolute value of inflation $\pi = \psi - \gamma$ under price rigidity

Using equation (5.33), we note first that it is the presence of price rigidity that allows money to have an impact on $\mu(L, \psi, \beta)$:

Lemma 12 In the limiting case without rigidities, money is superneutral: $\lim_{\beta \to \infty} \frac{\partial \mu(L,\psi,\beta)}{\partial \psi} = 0.$

Intuitively, when prices are perfectly flexible, relative prices and markups are independent of inflation, so that demand and hence, a research firm's profits are unaffected by money growth.

In contrast for $\beta < \infty$, the money growth rate ψ has two clear-cut countervailing effects on the innovation rate $\mu(L, \psi, \beta)$ which operate through money growth's influence on the firm's mark-up and relative price:

Negative initial mark-up effect of money growth under price rigidity As explained in section 5.3.4 an increase in ψ that raises the growth rate of marginal cost π raises the initial mark-up and relative price chosen by an



Figure 5.2: Effects on the partial equilibrium research intensity given employment $\mu(\psi, \overline{L}, \beta)$ of an increase in ψ when inflation is positive.

intermediate good firm under price rigidity. The increase in the relative price lowers demand for the firm's good and hence, its instantaneous profit which in turn determines the incentive to innovate.^{32,33} In figure 5.2, the increase in ψ causes a downward shift of the LHS-curve which ceteris paribus reduces the innovation rate $\mu(L, \psi, \beta)$.

Positive mark-up erosion effect of money growth under price rigid-

ity The RHS of equation (5.33) is the new firm's compound discount rate. As first discussed in section 5.3.4.2, the compound discount rate decreases in the inflation rate which determines the rate of demand- and profit-raising mark-up erosion.³⁴ Since an increase in the money growth rate ψ ceteris paribus raises inflation, it therefore ceteris paribus raises the incentive to innovate and the innovation rate $\mu(L, \psi, \beta)$ via a decrease in the compound discount rate.

Graphically, the increase in ψ causes a downward shift in the RHS-curve

 $\overline{\frac{^{32}\partial LHS}{\partial \psi} = -\frac{\alpha LHS[(\eta \overline{q}+1)\mu + (\rho+\beta)]}{(\rho+\beta-\alpha\psi+\overline{\eta}\mu)[\rho+\beta-(\alpha-1)\psi + (\overline{\eta}-\overline{q})\mu]}} < 0 \text{ given condition (5.37) and } \overline{\eta} - \overline{q} = [\eta + (\alpha-1)] \left(q^{\alpha-1} - 1\right) + 1 > 0.$

³³Note that inflation only has an effect on profits through its influence on demand since the initial mark-up under price rigidity is optimally chosen by the firm to offset the direct effect of the changing mark-up on the firm's profit *per unit*. See Appendix 5.A.1.

³⁴Remember that the discount rate is the obsolescence-adjusted interest minus the profit growth rate and that the present value of unit profit is unaffected by inflation since the initial mark-up is chosen optimally to offset this effect.

in figure 5.2, which ceteris paribus raises μ .

Net effect of money growth on economic growth depends on whether inflation is positive or negative. The negative price dispersion effect of an increase in money growth ψ shifts the LHS-curve of equation (5.33) downward while the positive mark-up erosion effect shifts the RHS-curve downward. Which effect is stronger, i.e. the sign of the net effect of money growth on $\mu(L, \psi, \beta)$ depends on whether inflation is positive or negative:

Lemma 13 An increase in the money growth rate ψ decreases (increases) the innovation rate $\mu(L, \psi, \beta)$ when inflation is positive (negative).

This is intuitive since both the discussed effects describe the impact on a firm's effective mark-up of a restriction on its price setting. This restriction which leads to suboptimal mark-ups cannot make the firm better off. Now while at $\pi = 0$, rigidity is ineffective since marginal cost is constant over time so that firms have no desire to readjust prices, for any departure from price stability, price rigidity becomes binding. At $\pi < 0$, an increase in ψ moves inflation closer to $\pi = 0$, reducing the distortion of the firm's mark-up and therefore increasing profits and the incentive to innovate which determines the growth rate. In contrast, at $\pi > 0$, an increase in ψ raises inflation and thus exacerbates the effects of rigidity, reducing profits and economic growth.

5.3.7.6 Innovation rate μ depends negatively on price rigidity $1/\beta$

Lemma 14 An increase in the level of rigidity (i.e., decrease in β) decreases the innovation rate $\mu(L, \psi, \beta)$ for $\beta < \infty$.

Analogously to the discussion of β 's effect on the wage in section 5.3.6.1, an increase in the frequency of price adjustments β has qualitatively the same effects as a reduction in the inflation rate π : It reduces the need to have a forward-looking initial mark-up, reducing the *initial mark-up* effect of money growth by drawing the initial mark-up chosen closer to the static optimum. Graphically, an increase in β shifts the LHS-curve upward (downward) in figure 5.1 when inflation is positive (negative), which ceteris paribus decreases (increases) the partial equilibrium innovation rate.

At the same time, an increase in the frequency of price adjustment via β reduces the *mark-up erosion* effect of money growth since it shifts more weight to the discount rate for periods when prices are flexible, reducing the weight of the correction factor. Graphically, an increase in β shifts the RHS-curve upward (downward) in figure 5.1 when inflation is positive (negative), ceteris paribus increasing (decreasing) μ .

The intuition for the negative net effect of rigidity regardless of whether inflation is positive or negative is closely connected to the intuition concerning the effect of ψ : An intermediate good producer's profit is affected by rigidity only through the latter's effect on the firm's optimal and effective price. If changing prices infrequently were a profit-maximising strategy, the firm would have chosen this pricing strategy under flexibility, so there is no scope for price rigidity to increase the return to R&D.

5.4 Existence and uniqueness of the steady state equilibrium

Any solution to the equation:

$$\mu \left[L\left(\mu, \psi, \beta \right), \psi, \beta \right] = \mu \tag{5.38}$$

is a steady state equilibrium innovation rate and $L^*(\psi,\beta) := L[\mu(L,\psi,\beta),\psi,\beta]$ is the corresponding equilibrium employment level.

Remark 15 In the leading example to be presented in the next section, the steady state equilibrium is unique.

More generally, we can say the following:

Proposition 16 Given conditions (5.35)-(5.37), a steady state equilibrium exists. If the maximum feasible μ in the economy, $\mu^{max} := \mu(\overline{L}, \psi, \beta)$ is sufficiently small, then there is a unique steady state equilibrium.



Figure 5.3: Existence and uniqueness of general steady state equilibrium research intensity mu in our leading example

Remember that we explained in section 5.3.6 that the wage and employment are approximately invariant to changes in μ when μ is small in relation to $\beta + \mu$. So given continuity of $L(\mu, \psi, \beta)$ in $\mu/(\beta + \mu)$, sufficiently small in this context means that the maximum feasible innovation rate μ^{max} must be small in relation to the frequency of price adjustment β , which is the case for all plausible economies. The $L(\mu)$ -curve is then approximately linear and crosses the $\mu(L)$ -curve once. Figure 5.3 illustrates this for our leading example.

5.5 Comparative statics: Employment level and economic growth rate in general equilibrium

In this section, we discuss the comparative static properties of the output growth rate, which is proportional to the innovation rate, and of the levels of employment and output in steady state equilibrium. These properties are determined by four effects we have already discussed in the partial equilibrium analysis of sections 5.3.6 and 5.3.7: The *Initial mark-up effect*, the *Mark-up erosion effect*, the *Average mark-up effect* and the *Price dispersion effect*. Using a calibrated example, we will in particular discuss which monetary policies would be chosen by monetary authorities interested in promoting employment and economic growth, respectively, and analyse the effect of price rigidity on growth and employment. Table 5.1 lists the parameter values chosen for our leading example.

parameter	value	parameter	value
q	1.2	ρ	0.015
α	10	σ	150
β	2.5	\overline{L}	4.7250
η	2		

Table 5.1: Parameter values used for calibration of the model with endogenous labour supply

The calibration is chosen to yield realistic and empirically plausible values for the economy's endogenous variables at a baseline money growth rate $\psi =$ 0.055 per annum: The rate of economic growth is 2% while the unemployment rate is 5.3%. The mark-up chosen by firms amounts to 12.9%, while the average period during which prices are fixed is 0.40 years or 4.8 months.³⁵

5.5.1 Inflation and employment: Monetary policy for promoting employment

First, note that superneutrality continues to hold:

Proposition 17 $\lim_{\beta\to\infty} \frac{dL}{d\psi} = \lim_{\beta\to\infty} \frac{d\mu}{d\psi} = \lim_{\beta\to\infty} \frac{d\gamma}{d\psi} = 0.$

The intuition remains unchanged: With perfectly flexible prices, inflation has no influence on relative prices or average mark-ups and therefore does not influence real variables.

Regarding the effect of money growth on employment in the presence of price rigidity ($\beta < \infty$), we first present the following proposition:

Proposition 18 Starting from an equilibrium with price stability, an increase in the money growth rate ψ increases employment.

³⁵Both values are well in line with empirical estimates, see Basu and Fernald [1995, 1997] and Bils and Klenow [2004], respectively.

When we take into consideration the indirect effect of money growth on employment via the research intensity as well as the direct effects, then starting from an equilibrium with price stability, an increase in the money growth rate ψ lowers the average mark-up, so that output per labour unit, the wage and employment increase in ψ . Thus a monetary policy entailing moderate inflation is preferable to price stability for a monetary authority that wants to increase employment.

Before turning to the intuition for this result, consider more generally the shape of the function $L\left[\mu\left(\psi, L, \beta\right), \psi, \beta\right]$. Note that the total derivative of employment with respect to the money growth rate can be written as $\frac{dL}{d\psi} = \frac{L}{\psi} \left[\varepsilon_{L,\psi} + \varepsilon_{L,\mu} * \varepsilon_{\mu,\psi}\right]$ where $\varepsilon_{x,y}$ is the partial elasticity of x with respect to y. As argued in section 5.3.6.1, the elasticity of employment with respect to the innovation rate μ vanishes for $\mu/(\mu + \beta) \to 0$ and is indeed very small for all sensible calibrations since the contribution of the innovation rate to the degree of price flexibility $\beta + \mu$ is small.³⁶ Thus, $\varepsilon_{L,\mu} * \varepsilon_{\mu,\psi}$ is very small and the indirect effect of money growth on employment is negligible. We then have that Lemma (8) holds in general equilibrium:

Corollary 19 For sufficiently small $\varepsilon_{L,\mu} * \varepsilon_{\mu,\psi}$, employment is a hump-shaped function of money growth with a maximum at a money growth rate $\psi_2 > 0$ associated with a positive inflation rate $\pi(\psi_2) > 0$.

Figure 5.4 reflects this result for our leading example: The solid line depicts the function $L(\psi)$ in general equilibrium. The pointed line, which shows only the partial equilibrium effect of ψ on employment L given μ , is virtually indistinguishable from the solid line for negative and small ψ . For bigger values of ψ , the figure shows that the indirect effect through the research intensity reinforces the direct effect of money growth on employment, yet to a quantitatively negligible degree.

Thus through an increase in money growth starting from small rates of inflation, the monetary authority is successful in lowering the average

³⁶At the same time, the point elasticity of the research intensity μ with respect to the money growth rate ψ also has to be small in all realistic examples- remember that a large discrete increase in the money growth rate from 1 percentage point to 10 percentage points should lower growth by not significantly more than a quarter percentage point.



Figure 5.4: Partial and total effect of money growth on employment

mark-up in spite of the fact that intermediate good firms raise their initial mark-up in anticipation of price rigidity. This means monetary policy in this range successfully raises aggregate demand for intermediate goods which is inefficiently low due to monopolistic competition. The positive effect on aggregate demand via a lower average mark-up dominates money growth's negative effect on production efficiency at low levels of inflation, leading to a higher real wage and higher equilibrium employment than under price stability.

There is thus a range of money growth rates that entails a *Phillips-Curve*trade-off for the monetary authority: Higher employment is only to be had at the price of higher inflation. In our leading example, this is true for money growth rates up to 2.60% or equivalently, positive inflation rates of up to 0.57%. In the range of money growth rates between 2.60% and 3.15% (inflation rate of 1.12%), employment again declines in money growth but is still higher than under price stability. Yet the effect is quantitatively small: At its maximum, employment is less than 0.01% higher than in the case of flexible prices. At the same time, the effect of an increase in money growth from $\psi = 0.01$ to $\psi = 0.1$ is quite sizeable: It increases the unemployment rate by over 0.8 percentage points from 5.28% to 6.08%. Therefore monetary policy aimed at fostering employment should feature a moderate inflation rate, while high inflation should be avoided since it significantly reduces employment.

5.5.2 Inflation and the output level

By equation (5.3), we have

$$\frac{Y\left(\tau\right)}{Q\left(\tau\right)} = \alpha \widetilde{w}\left(\tau\right) L\left(\tau\right)$$

This implies that $\frac{d\frac{Y(\tau)}{Q(\tau)}}{d\psi} = \alpha \left(L(\tau) \frac{d\tilde{w}(\tau)}{d\psi} + \tilde{w}(\tau) \frac{\partial L(\tau)}{\partial \tilde{w}} \frac{d\tilde{w}(\tau)}{d\psi} \right) = \alpha \left(L + \tilde{w} \frac{\partial L}{\partial \tilde{w}} \right) \frac{d\tilde{w}}{d\psi}$ or $\varepsilon_{\frac{Y}{Q},\psi} = (1 + \varepsilon_{L,\tilde{w}}) \varepsilon_{\tilde{w},\psi}$ where $\varepsilon_{x,y}$ is again the elasticity of x with respect to y. Given $\partial L/\partial \tilde{w} > 0$, the steady state level of output in efficiency units $[Y(\tau)/Q(\tau)]^*$ as a function of money growth ψ has the same shape as the wage and employment functions. In particular, it reaches its maximum at the same money growth rate associated with positive inflation where employment peaks, and it is strongly reduced by high inflation. Starting from the maximum value at $\psi = 2.6\%$ and increasing ψ to $\psi = 0.1$ in our baseline numerical example, the reaction of employment: Output drops by 7.22\%, whereas employment decreases by only 0.86\%. Figure 5.5 shows $Y(\tau)/Q(\tau)$ as a function of the money growth rate in the leading example.

5.5.3 Inflation and economic growth: Monetary policy for promoting growth

Remember that in section 5.3.7.5 holding constant employment, we found a hump-shaped relationship between the innovation rate and money growth or inflation, respectively. The innovation rate peaked at zero inflation. Our subsequent analysis of how the additional influence of money growth on the wage and employment changes this result shows that while the hump-shaped relationship persists, the best policy for a monetary authority interested in promoting economic growth features a positive rate of inflation.



Figure 5.5: Output as a function of the money growth rate

Regarding the hump-shaped relationship between money growth and economic growth, which by equation (5.21) is a linear function of the innovation rate, we have the following proposition:

Proposition 20 For small values of the money growth rate ψ , the economic growth rate γ increases in ψ , for large values of ψ , the growth rate γ decreases in ψ .

Thus the qualitative relation is similar in partial equilibrium and general equilibrium. Yet there is one qualitative difference:

Proposition 21 The economic growth rate reaches its maximum at a positive rate of inflation.

As shown in section 5.3.7.5, *holding constant employment*, the incentive to innovate is reduced by nonzero inflation because an intermediate good producer's profits decreases due to the restriction on his price setting imposed by price rigidity: In anticipation of price rigidity, the initial mark-up chosen is higher than the static optimum. During the firm's life time, inflation erodes the mark-up. Consequently, the mark-up generically does not correspond to the optimal one, lowering profits and thus, the patent price.



Figure 5.6: Economic growth rate as a function of the money growth rate

In general equilibrium, these effects are still present. Yet at the same time, lemma 11 shows that the innovation rate increases monotonically in employment. As seen in section 5.3.6.1, at zero inflation and small positive inflation rates the wage and employment increase in the money growth rate ψ . At small positive rates of inflation, the positive indirect effect of an increase in ψ on growth via employment is stronger than the negative direct effect so that the incentive to innovate and the growth rate increase in the money growth rate.

Yet as the money growth rate ψ continues to rise, distortions increase and the wage begins to fall in ψ , which adds to the mark-up distorting effects of positive money growth in causing a fall in the economic growth rate.

Figure (5.6) shows the economic growth rate as a function of money growth in our leading example.

The economic growth rate is maximised at $\psi = 0.021$, which corresponds to the positive inflation rate $\pi = 0.07\%$. At this inflation rate, the economic growth rate is 2.03% compared to 2.0% in the baseline case. While this effect is rather small, the effect of inflation on growth is more drastic at inflation rates that are further away from the optimum: When the money growth rate increases from $\psi = 0.01$ to $\psi = 0.1$, the growth rate decreases by 0.21 percentage point, which corresponds closely to empirical estimates mentioned in the introductory chapter.

Thus while a monetary authority that wants to promote growth should allow for moderate inflation rather than aim at price stability, it should also be aware of the growth depressing effects of high inflation.

5.5.4 Limited trade-off for monetary policy between employment and growth

In the preceding two sections, we examined which monetary policy would be optimal for a monetary authority interested in promoting either employment or economic growth. We found that some inflation raises the wage and employment. Due to its effect on employment, a small positive inflation rate also fosters economic growth. At the same time, too much inflation proved to reduce both employment and inflation.³⁷

There is no strong conflict between promoting growth and raising employment for the central bank: Given perfect information about the central union's policy and given any preference structure involving the goals of employment and economic growth, the monetary authority will always choose a money growth rate from the range $\psi \in (\psi_{\gamma}, \psi_L)$ where $\psi_{\gamma} (\psi_L)$ maximises economic growth (employment). Within this range, an increase in ψ always increases employment and lowers economic growth.³⁸ The trade-off is limited in our calibrated examples where ψ_{γ} and ψ_L are very close. Figure 5.7 illustrates the trade-off for our leading example, where the range of money growth rates involved is $\psi \in (0.021, 0.026)$ which corresponds to inflation rates between 0.07% and 0.57%.

 $^{^{37}}$ This implies that a long-run version of *Okun's law*, according to which an increase in economic growth is always accompanied by a decrease in the unemployment rate, holds in our model for most money growth rates.

³⁸This follows from the fact that the inequality $\psi_{\gamma} < \psi_L$ always holds. To get intuition for this fact, remember that the total effect of an increase in money growth on economic growth comprises the sum of non-employment related effects and the employment related effect, where we know that the former are negative at positive rates of inflation. Thus, the total effect of ψ on γ $(d\gamma/d\psi)$ is always smaller than the employment related effect. Therefore, at the money growth rate that maximises employment $(dL/d\psi = 0), d\gamma/d\psi < 0$ so $\gamma(\psi)$ reaches its maximum at a smaller ψ .



Figure 5.7: Trade-off between employment and growth

5.5.5 A digression on the difference in results concerning the effect of an increase in ψ between the present model and the model of chapter 2

The comparative static results of the models of chapters 2 and 5 regarding the effects of a change in the money growth rate differ in two respects due to the modification of the sector structure: First, when intermediate goods are produced using output (labour) as in the present chapter (chapter 2), the relevant growth rate of marginal cost is the inflation rate π (the money growth rate ψ) so that the outcomes of the model under price rigidity are identical to those of the model with flexible prices only at $\pi = 0$ ($\psi = 0$). Second, in the present chapter an increase in ψ raises the incentive to innovate when the growth rate of marginal cost is negative ($\pi < 0$), whereas in chapter 2 an increase in ψ lowers growth *regardless* of the initial value of the growth rate of marginal costs.

This second more substantial difference in results vanishes when one holds constant the real wage level in efficiency units w/(PQ) in both models. Given the real wage, the incentive to innovate in both models increases (falls) in the money growth rate when the growth rate of marginal cost is negative (positive),³⁹ because any departure from constant marginal cost ($\psi = 0$ and

 $^{^{39}}$ See the intuition for proposition 3 in section 2.4.2 of chapter 2 and lemma 13 in section 5.3.7.5 of the present chapter.

 $\pi=0,$ respectively) raises the distortion of the mark-up of the intermediate goods firm.⁴⁰

In both model specifications, the real wage level depends negatively on the average monopolistic mark-up.⁴¹ The average mark-up in turn depends on the money growth rate ψ via the initial mark-up and via the deviation between initial and average mark-up caused by the fact that existing markups are eroded by money growth.⁴² At negative growth rates of marginal cost, an increase in ψ causes a decrease in the average mark-up which in turn raises the real wage given Q.⁴³ In calibrated examples of the models of chapters 2 and 5, the average mark-up (the wage) is minimised (maximised) at a small positive growth rate of marginal cost.

In the present chapter, growth depends positively on employment which in turn increases in the real wage in efficiency units. Given that the latter is maximised at a (small) positive inflation rate, so is the output growth rate.

In contrast, in the model specification of chapter 2, an intermediate goods producer upon market entry sets his price as a mark-up over marginal cost, the nominal wage, implying that the initial relative price increases in the real wage. Since demand for the new good and hence, profits fall in the initial relative price, the incentive to innovate decreases in the real wage level. Given that the real wage decreases in the money growth rate at $\psi < 0$ and at the same time, the effect of a change in the initial relative price is stronger at $\psi < 0$ than at $\psi > 0$, the economic growth rate is maximised at the lowest

⁴⁰In the context of chapter 2, the production inefficiency resulting from non-zero growth of marginal cost and optimal prices aggravates the fall in profits and hence, in the incentive to innovate.

⁴¹In the model specification of chapter 2, intermediate goods firms set prices as a markup over the nominal wage, the final good sector is perfectly competitive with zero profits and only intermediate goods are used in final good production. Hence, the real wage is one over the average mark-up.

In the specification of the present chapter, the real wage is the productivity of labour in final good production which depends positively on the amount of intermediate goods used. This amount (and therefore, the real wage level) depends negatively on the average mark-up charged by intermediate goods producers. The fact that the wage also depends on the degree of price dispersion in the specification of the present chapter does not change the results of our analysis here.

 $^{^{42}}$ See the discussion in section 5.3.6.

 $^{^{43}}$ See footnote 24 in chapter 2 and footnote 24 in section 5.3.6.1 of the present chapter.

admissible value of ψ .

Which specification is more plausible? Producing intermediate goods with constant returns to a single input is a strong assumption. Compared to this simplification, whether the unique input is labour or output (capital with full depreciation) seems to be of secondary importance. Investigating whether lower monopolistic distortions are empirically associated with higher (chapter 5) or lower growth (chapter 2) might help to distinguish between the specifications.

Note, however, that a positive influence of the real wage on labour supply in the model specification of chapter 2 would reduce the difference in results: The growth maximising value of the money growth rate would rise in this model, too, since firstly, growth depends positively on employment,⁴⁴ and secondly, the real wage is maximised at a positive value of ψ just as in the model of the present chapter.

Note finally that while there are differences concerning the implications of deflation for economic growth, the much more central policy implication that high inflation is detrimental to growth emerges consistently from both model specifications.

5.5.6 Comparative statics regarding the level of price rigidity

Unlike in partial equilibrium with constant employment where the innovation rate growth increased in β whenever $\pi \neq 0$, with endogenous labour supply price rigidity is not universally bad for innovation and economic growth. In spite of the distortions it entails, the very presence of price rigidity allows the monetary authority to implement a policy which through the lowering of the average mark-up raises employment and with it, the economic growth rate beyond its level in a world without rigidities. This is summarised in the following proposition which we prove in Appendix 5.A.4:

Proposition 22 At sufficiently low levels of positive inflation, employment

 $^{^{44}\}mathrm{See}$ equation 2.36 and footnote 46 in chapter 2.

output and economic growth are higher under price rigidity than in a world without price rigidity.

5.6 Conclusion

Endogenising employment has shown that there is a non-linear long-run Phillips-curve in the Schumpeterian model with price rigidity: Employment and output are hump-shaped functions of money growth. This result both complements and generalises the results concerning the long run found by the New Keynesian literature and reported in section 1.3.2 of the introductory chapter since this literature relied mainly on simulation exercises and on a less general specification of price rigidity. At the same time, we find that the endogeneity of growth does not significantly change the qualitative or quantitative relationship between inflation and employment or output.

In contrast, the endogeneity of employment does matter for the inflationgrowth relationship: Higher employment raises growth because the marginal productivity of intermediate goods, whose growing quality is the engine of growth, increases in employment. Thus as in the endogenous growth literature investigating the consequences of introducing money via a cash in advance constraint discussed in section 1.4.1.2 of chapter 1, inflation impacts on growth via employment because the latter influences the marginal productivity of investment. Yet fundamentally different policy recommendations emerge from their analysis of the leisure-substitution channel than from the present analysis of the sticky price transmission channel: In the models of section 1.4.1.2, growth is maximised when the monetary authority follows the Friedman rule, i.e. contracts the money supply at a rate that makes the nominal money supply zero. In contrast, in the present context where employment is influenced by inflation via the latter's influence on the average monopolistic mark-up and on price dispersion, employment and therefore growth are maximised at small positive growth rates.

Are these effects empirically relevant? A look at the data suggests they are: There are several studies investigating the relationship between inflation
and mark-ups,⁴⁵ which find evidence of a negative influence of inflation on mark-ups. At the same time, higher inflation also seems to be associated with more price dispersion empirically.⁴⁶

5.A Appendix to chapter 5

5.A.1 A new firm's market value

Following analogous reasoning about the development of the firm's price given the process for price reset signals β as in Appendix 2.A.1, the firm's market value is

$$E\left(V_{k_{j}}\left(\tau\right)\left|t_{k_{j}}=\tau\right)\right)$$

$$=\widetilde{A}\int_{\tau}^{\infty}e^{-\widetilde{\chi}\left(s-\tau\right)}\left[\int_{\tau}^{s}\beta e^{-\beta\left(s-\theta\right)}p^{*}\left(\theta\right)^{1-\alpha}d\theta+\left(1-\int_{\tau}^{s}\beta e^{-\beta\left(s-\theta\right)}d\theta\right)p^{*}\left(\tau\right)^{1-\alpha}\right]ds$$

$$-\widetilde{A}w\left(\tau\right)\int_{\tau}^{\infty}e^{-(\widetilde{\chi}-\omega)\left(s-\tau\right)}\left[\int_{\tau}^{s}\beta e^{-\beta\left(s-\theta\right)}p^{*}\left(\theta\right)^{-\alpha}d\theta+\left(1-\int_{\tau}^{s}\beta e^{-\beta\left(s-\theta\right)}d\theta\right)p^{*}\left(\tau\right)^{-\alpha}\right]ds$$

where $\widetilde{A} = \left(\frac{\alpha-1}{\alpha}A\right)^{\alpha} q^{(\alpha-1)k_j}L(\tau)$ and $\widetilde{\chi} = i + \mu - \alpha \pi$. Going through a number of steps reported in more detail in Appendix 2.A.1 yields

$$E\left(V_{k_{j}}|\tau\right) = \widetilde{A}\left(\frac{p^{*}\left(\tau\right)}{P(\tau)}\right)^{-\alpha} \frac{\widetilde{\chi} + \beta + (\alpha - 1)\pi}{\widetilde{\chi} + (\alpha - 1)\pi} \left[\frac{p^{*}\left(\tau\right)}{\widetilde{\chi} + \beta} - \frac{P\left(\tau\right)}{\widetilde{\chi} + \beta - \pi}\right]$$

Note that again, the initial mark-up $\frac{\alpha}{\alpha-1} \frac{\tilde{\chi}+\beta}{\tilde{\chi}+\beta-\pi}$ is again chosen such that the present value of revenues is identical to what it would have been if revenues had grown at the same constant rate as costs, implying that changes in relative prices only affect profits through their impact on demand, not via their effect on unit profit. Using the equation for the optimal price $p^*(\tau)$ (5.22) and reinserting $\tilde{\chi} = i + \mu - \alpha \pi$ and $\tilde{A} = \left(\frac{\alpha-1}{\alpha}A\right)^{\alpha}q^{(\alpha-1)k_j}L$ we have

 $^{^{45}}$ E. g., Benabou [1992], Banerjee and Russell [2005] and Banerjee, Mizen and Russell [2006]. More references can be found in the last mentioned paper.

⁴⁶Parks [1978] is a seminal paper in this literature. For recent contributions see Banerjee, Mizen and Russell [2006] and the references therein.

equation (5.24) in the main text.

5.A.2 Total intermediate good production

The total production of intermediate goods at time τ can be rewritten as

$$X(\tau) = \left(\frac{\alpha - 1}{\alpha}A\right)^{\alpha} L(\tau) Q(\tau) P(\tau)^{\alpha} \sum_{k=1}^{k^{\max}} d_k(\tau) \frac{q^{(\alpha - 1)k}}{Q(\tau)} \sum_{\{j|k_j = k\}} \left(p_{k_j}(\tau)\right)^{-\alpha}$$

where p_{k_j} is the price of sector j that is at quality rung k and $d_k(\tau)$ is the number of sectors at quality rung k at time τ . Analogously to the procedure of Appendix 2.A.2, the average price effective at time τ can be expressed as a weighted average of past optimal prices, using the same weights $f(s,\tau) =$ $(\mu + \beta) e^{-(\mu+\beta)(\tau-s)}$ reflecting the probability that a price valid at time τ has not been changed since time s. Using the steady state growth rate π of p^* , we have

$$X(\tau) = \left(\frac{\alpha - 1}{\alpha}A\right)^{\alpha} L(\tau) \left[\frac{p^*(\tau)}{P(\tau)}\right]^{-\alpha} Q(\tau) \int_{-\infty}^{\tau} (\beta + \mu) e^{-(\beta + \mu - \alpha\pi)(\tau - s)} ds$$

Solving the integral which converges for $\beta > \alpha \psi$ leads to (5.25) in the main text. Convergence of the integral is ensured by condition (5.37).

5.A.3 Total final good production

Total final good production can be rewritten as

$$Y(\tau) = A^{\alpha} \left(\frac{\alpha - 1}{\alpha}\right)^{\alpha - 1} L(\tau) P(\tau)^{\alpha - 1} Q(\tau) \sum_{j=1}^{N} \frac{q^{(\alpha - 1)k_j(\tau)}}{Q(\tau)} p_j(\tau)^{-(\alpha - 1)}$$

As in Appendix 5.A.2, the average intermediate good price effective at τ can be expressed as a weighted average of past optimal prices with the weights $\tilde{f}(s,\tau)$ defined in Appendix 5.A.2.

$$Y(\tau) = A^{\alpha} \left(\frac{\alpha - 1}{\alpha}\right)^{\alpha - 1} L(\tau) P(\tau)^{\alpha - 1} Q(\tau) \int_{-\infty}^{\tau} p^*(s)^{-(\alpha - 1)} \widetilde{f}(s, \tau) ds$$

Inserting $\tilde{f}(s,\tau) = (\mu + \beta) e^{-(\mu + \beta)(\tau - s)}$ and using that the optimal price grows at rate π in equilibrium, we can calculate the integral's value which gives

$$\frac{Y(\tau)}{Q(\tau)} = \left(\frac{\alpha - 1}{\alpha}\right)^{\alpha - 1} A^{\alpha} L(\tau) \left[\frac{p^*(\tau)}{P(\tau)} \left(\frac{\beta + \mu}{\beta + \mu - (\alpha - 1)\pi}\right)^{-\frac{1}{\alpha - 1}}\right]^{-(\alpha - 1)}$$

Note that convergence of the integral is ensured by assumption (5.37).

Next, we want to rewrite Y/Q as a product of efficient production with a given X/Q and a term that describes production inefficiency due to price dispersion. To find the maximum amount of final goods Y^{eff} that can be produced with a given amount X of intermediate goods when X is distributed efficiently among the intermediate good types x_j , we solve the following problem:

$$\max_{x_j} Y + \varpi \left[X - \sum_j x_j \right]$$

subject to the final good production function (5.1). The first order condition for x_j can be rewritten as

$$\left(\frac{\frac{\alpha-1}{\alpha}AL^{1/\alpha}q^{k_j\frac{\alpha-1}{\alpha}}}{\varpi}\right)^{\alpha} = x_j$$

Aggregating over the intermediate good types and solving for ϖ gives

$$\frac{\alpha - 1}{\alpha} A L^{1/\alpha} \left(\frac{X}{Q}\right)^{-1/\alpha} = \varpi$$

Reinserting this into the first order condition gives

$$q^{k_j(\alpha-1)}\left(\frac{X}{Q}\right) = x_j$$

which can in turn be reinserted in the final good production function (5.1), yielding

$$Y^{eff}(\tau) = AL(\tau)^{1/\alpha} \left(\frac{X(\tau)}{Q(\tau)}\right)^{\frac{\alpha-1}{\alpha}} Q(\tau)$$

Now multiplying and dividing actual total final good production (5.27) by Y^{eff} , replacing $X(\tau)/Q(\tau)$ in the denominator with the amount of intermediate goods actually used (5.25) and simplifying, we have equation (5.27) in the text.

5.A.4 Proofs of propositions, lemmata and corollaries (8)-(22)

Proof of lemma 8. The derivative with respect to the money growth rate ψ of the function $\widetilde{w}(\psi)$ from equation (5.28) for a given innovation rate μ is $\partial \widetilde{w}/\partial \psi = \frac{(\alpha-1)\widetilde{w}}{\beta+\mu-(\alpha-1)\pi} \left\{ -\frac{r+\beta+\mu}{r+\beta+\mu-\alpha\pi} \frac{\beta+\mu-(\alpha-1)\pi}{r+\beta+\mu-(\alpha-1)\pi} + 1 \right\}$. Examining the nulls of the derivative shows that the the function has extrema at

$$\psi_{1/2} = \frac{1}{2} \frac{1}{\alpha - 1} \left\{ \begin{bmatrix} r + \beta + \mu + 2(\alpha - 1)\gamma \end{bmatrix} - (\beta + \mu + r)^{1/2} (\beta + \mu + \frac{4 - 3\alpha}{\alpha} r)^{1/2} \right\}$$
associated with

$$\pi_{1/2} = \frac{1}{2} \frac{1}{\alpha - 1} \left(\beta + \mu + r \right)^{1/2} \left\{ \left(\beta + \mu + r \right)^{1/2} \stackrel{-}{-} \left(\beta + \mu + \frac{4 - 3\alpha}{\alpha} r \right)^{1/2} \right\}$$

with $\pi \left(\psi_2 \right) > \pi \left(\psi_1 \right) > 0.$

Examining the second derivative at ψ_1 and ψ_2 shows that $\widetilde{w}(\psi)$ has a maximum (minimum) at $\psi_1(\psi_2)$ because

$$\frac{\partial^2 \frac{w(\tau)}{P(\tau)Q(\tau)}}{\partial \psi^2} \bigg|_{\psi=\psi_1} = \frac{-4\alpha \left[\beta + \mu + \frac{4-3\alpha}{\alpha}r\right]^{1/2} (\beta + \mu + r)^{-1/2} (\alpha - 1)\widetilde{w}}{\left\{ (\beta + \mu + r)^{1/2} + \left[\beta + \mu + \frac{4-3\alpha}{\alpha}r\right]^{1/2} \right\}^2 \left\{ \frac{1}{2} \frac{\alpha - 2}{\alpha - 1} (\beta + \mu + r)^{1/2} + \frac{1}{2} \frac{\alpha}{\alpha - 1} \left[\beta + \mu + \frac{4-3\alpha}{\alpha}r\right]^{1/2} \right\}^2} < 0$$
and
$$\frac{\partial^2 \frac{w(\tau)}{P(\tau)Q(\tau)}}{\partial \psi^2} \bigg|_{\psi=\psi_2} = \frac{+4\alpha (\beta + \mu + r)^{-1/2} \left[\beta + \mu + \frac{4-3\alpha}{\alpha}r\right]^{1/2} \left[\beta + \mu + \frac{4-3\alpha}{\alpha}r\right]^{1/2} (\alpha - 1)\widetilde{w}}{\left\{ \frac{1}{2} \frac{\alpha - 2}{\alpha - 1} (\beta + \mu + r)^{1/2} - \frac{1}{2} \frac{\alpha}{\alpha - 1} \left[\beta + \mu + \frac{4-3\alpha}{\alpha}r\right]^{1/2} \right\}^2 \left\{ (\beta + \mu + r)^{1/2} - \left[\beta + \mu + \frac{4-3\alpha}{\alpha}r\right]^{1/2} \right\}^2 > 0.$$

Further, we find that $\psi_2 > \frac{1}{\alpha - 1}\beta$, which is the maximum admissible money growth rate from condition (5.37), since

$$\frac{1}{2}\frac{1}{\alpha-1}\left\{\left[r+\beta+\mu+2\left(\alpha-1\right)\gamma\right]+\left(\beta+\mu+r\right)^{1/2}\left(\beta+\mu+\frac{4-3\alpha}{\alpha}r\right)^{1/2}\right\}>\frac{1}{\alpha-1}\beta.$$

Therefore, $\widetilde{w}\left(\psi\right)$ increases (decreases) in ψ for all admissible money growth rates ψ with $\psi < \psi_1$ ($\psi > \psi_1$). The inflation rate
 $\pi\left(\psi_1\right) = \frac{1}{2}\frac{1}{\alpha-1}\left(\beta+\mu+r\right)^{1/2}\left\{\left(\beta+\mu+r\right)^{1/2}-\left(\beta+\mu+\frac{4-3\alpha}{\alpha}r\right)^{1/2}\right\}$ is positive since $\alpha > 1$ ensures that $1 > (4-3\alpha)/\alpha$.

Proof of lemma 9. Straightforward calculus shows that

$$\frac{\partial \tilde{w}}{\partial \beta} = \frac{(\alpha - 1)\tilde{w}\pi}{(\rho + \beta - \alpha\psi + \bar{\eta}\mu)[\rho + \beta - (\alpha - 1)\psi + (\bar{\eta} - \bar{q})\mu]} * \left[1 - \frac{r + \beta + \mu - \alpha\pi}{\beta + \mu} \frac{r + \beta + \mu - (\alpha - 1)\pi}{\beta + \mu - (\alpha - 1)\pi}\right].$$
 The term

in square brackets is always negative under deflation, while the fraction in front of the square brackets is negative (positive) for $\pi < 0$ ($\pi > 0$), so that $\frac{\partial \tilde{w}}{\partial \beta} > 0$ for $\psi < \psi_0$ with $\pi(\psi_0) = 0$. A very strict sufficient condition for the term in square brackets to be negative is $r > \alpha \pi$, for which $\rho/\alpha > \psi$ is again a sufficient condition so that $\frac{\partial \tilde{w}}{\partial \beta} > 0$ holds for $\rho/\alpha > \psi > \psi_0$.

Proof of lemma 10. Conditions (5.36) and (5.37) are jointly sufficient for the LHS of equation (5.33) to be concave in μ :

 $\frac{\partial^2 LHS}{\partial \mu^2} = \frac{C_1[(\overline{\eta} - \alpha \overline{q})\psi + \overline{q}(\rho + \beta)]\{(1+\alpha)[(\eta \overline{q}+1)\psi + \overline{q}(\rho + \beta)] - 2\overline{\eta}[\rho + \beta - (\alpha - 1)\psi + [(\eta + \alpha - 1)\overline{q}+1]\mu]\}}{(\rho + \beta - \alpha \psi + \overline{\eta}\mu)^3[\rho + \beta - (\alpha - 1)\psi + (\overline{\eta} - \overline{q})\mu]\left(\frac{\rho + \beta - (\alpha - 1)\psi + (\overline{\eta} - \overline{q})\mu}{\rho + \beta - \alpha \psi + \overline{\eta}\mu}\right)^{1+\alpha}} \text{ where } C_1 = \alpha \frac{L}{\lambda} A^{\alpha} \frac{1}{\alpha - 1} \left(\frac{\alpha - 1}{\alpha}\right)^{2\alpha}, \ \overline{q} = q^{\alpha - 1} - 1 > 0 \text{ and } \overline{\eta} = [(\eta + \alpha) \overline{q} + 1]. \text{ Given } \\ \text{ that } \overline{\eta} - \alpha \overline{q} = \eta \overline{q} + 1 > 0 \text{ and } \beta > \alpha \psi, \ \frac{\partial^2 LHS}{\partial \mu^2} < 0 \text{ when condition } (5.36) \\ \text{ holds. Also, } \frac{\partial^2 RHS}{\partial \mu^2} = \frac{2\alpha\beta(\eta \overline{q}+1)[(\overline{\eta} - \alpha \overline{q})\psi + \overline{q}(\rho + \beta)]}{[\rho + \beta + (\eta \overline{q} + 1)\mu]^3} > 0 \text{ when condition } (5.37) \text{ and } \\ \text{ the first inequality in condition } (5.36) \text{ hold, so these conditions are sufficient } \\ \text{ to ensure that the RHS of the equation is convex in } \mu. With condition (5.35) \\ \text{ we make sure that the value for } \mu \to 0 \text{ of the LHS of equation } (5.33) \text{ is larger } \\ \text{ than that of the RHS. Further, note that the RHS of equation } (5.33) goes to \\ \text{ infinity as } \mu \to \infty \left(\lim_{\mu \to \infty} \frac{\rho + [\eta(q^{\alpha - 1} - 1) + 1]\mu}{\rho + \beta + [\eta(q^{\alpha - 1} - 1) + 1]\mu} \left(\rho + \beta - \alpha\psi + \overline{\eta}\mu\right) = \infty \\ \text{ since } \overline{\eta} > 0 \right) \text{ while the limit of the LHS is bounded}$

 $\left(\lim_{\mu\to\infty} LHS = \frac{L}{\lambda} A^{\alpha} \frac{1}{\alpha-1} \left(\frac{\alpha-1}{\alpha}\right)^{2\alpha} \left(\frac{\overline{\eta} - \left(q^{\alpha-1} - 1\right)}{\overline{\eta}}\right)^{-\alpha} < \infty\right).$ Thus the two functions have a unique intersection with $\mu > 0$.

Proof of lemma 12. For $\beta \to \infty$, the zero profit condition (5.33) reduces to $\frac{L}{\lambda} \left(\frac{\alpha}{\alpha-1}\right)^{-2\alpha} A^{\alpha} \frac{1}{\alpha-1} = [\rho + (\eta \overline{q} + 1) \mu]$ so that the growth rate of money ψ has no influence on the equilibrium research intensity μ . Since by equation (5.21) $\gamma = (q^{\alpha-1} - 1) \mu$, the economy's real growth rate γ is independent of ψ , too.

Proof of lemma 13. Consider equation (5.33). First refer to figure 5.1 to see that given assumptions (5.35)-(5.37) and concavity of the LHS-curve, the latter's slope is always smaller than that of the RHS-curve at the equilibrium $\left(\frac{\partial LHS}{\partial \mu} - \frac{\partial RHS}{\partial \mu} < 0\right)$. Further, $\frac{\partial LHS}{\partial \psi} - \frac{\partial RHS}{\partial \psi} = -\frac{\alpha(\alpha-1)LHS(\psi-\bar{q}\mu)}{(\rho+\beta-\alpha\psi+\bar{\eta}\mu)[\rho+\beta-(\alpha-1)\psi+(\bar{\eta}-\bar{q})\mu]}$. From assumption (5.37) we have that $\beta > \alpha\psi$. Further, $\bar{\eta} = [(\eta + \alpha)\bar{q} + 1] > \bar{q}$ and from equations (5.21) and (5.18) we have $\psi - \bar{q}\mu = \pi$, so that $\frac{\partial LHS}{\partial \psi} - \frac{\partial RHS}{\partial \psi} \leq 0$ for $\pi \geq 0$. Thus we have that $\frac{d\mu}{d\psi} = -\frac{\frac{\partial LHS}{\partial \psi} - \frac{\partial RHS}{\partial \psi}}{\frac{\partial LHS}{\partial \mu} - \frac{\partial RHS}{\partial \mu}} \leq 0$ for $\pi \geq 0$.

Chapter 5. Money growth, employment and growth

Further, since $\gamma = (q^{\alpha-1} - 1) \mu$, $\frac{d\gamma}{d\psi} = (q^{\alpha-1} - 1) \frac{d\mu}{d\psi} \leq 0$ for $\pi \geq 0$. **Proof of lemma 14.**

Proof of lemma 14. From equation (5.33), $\frac{d\mu}{d\beta} = -\frac{\frac{\partial LHS}{\partial \beta} - \frac{\partial RHS}{\partial \beta}}{\frac{\partial LHS}{\partial \mu} - \frac{\partial RHS}{\partial \mu}} > 0$ since $\frac{\partial LHS}{\partial \beta} - \frac{\partial RHS}{\partial \beta} = LHS \frac{(\alpha-1)\alpha(\psi-\overline{q}\mu)^2 \{\rho+\beta-(\alpha-1)\psi+(\overline{\eta}-\overline{q})\mu\}^{-1}}{(\rho+\beta-\alpha\psi+\overline{\eta}\mu)\{\rho+\beta+(\eta\overline{q}+1)\mu\}} > 0$ is positive for $\psi - \overline{q}\mu = \pi \neq 0$ as $\beta > \alpha\psi$ given assumption (5.37) and $\overline{\eta} = [(\eta + \alpha)\overline{q} + 1] > \overline{q}$ and $\frac{\partial LHS}{\partial \mu} - \frac{\partial RHS}{\partial \mu} < 0$ from the proof of lemma 13.

Proof of proposition 16. Existence of a solution to equation (5.38) follows from the fact that for any $L \in (\frac{1}{2}\overline{L},\overline{L})$, the innovation rate $\mu(L,\psi,\beta)$ is uniquely defined (Lemma 10), and increases monotonically in L (Lemma 11) and that $L(\mu,\psi,\beta) \in (\frac{1}{2}\overline{L},\overline{L})$ is defined for each μ and is continuous in μ , with $L(\mu^{max},\psi,\beta) < \overline{L}$ since $\widetilde{w}(\mu^{max},\psi,\beta) < \infty$. Given $\lim_{[\mu/(\mu+\beta)]\to 0} \frac{\partial \widetilde{w}(\mu,\psi,\beta)}{\partial \mu} = \lim_{[\mu/(\mu+\beta)]\to 0} \frac{\partial L(\mu,\psi,\beta)}{\partial \mu} = 0$ (section 5.3.6) and continuity of L in $\mu/(\mu+\beta)$, there

always exists a μ small enough to ensure uniqueness.

Proof of proposition 17. From equation (5.28), for $\beta \to \infty$ the wage $w^{flex} = \frac{1}{\alpha} A^{\alpha} \left(\frac{\alpha}{\alpha-1}\right)^{-2(\alpha-1)}$ is constant so employment is independent of the money growth rate ψ . The equation determining the research intensity (5.38) reduces to $\frac{L(w^{flex})}{\lambda} \left[\overline{\alpha} \left(\frac{\alpha}{\alpha-1}\right)^{-\alpha}\right] = (\rho + \overline{\eta}_2 \mu)$ which is independent of ψ , too.

Proof of proposition 18. We analyse the influence at $\pi = 0$ of ψ on $L^* = L\left\{\widetilde{w}\left[\mu^*\left(\psi\right)\right],\psi\right\}$ (equation (5.30)). Here μ^* is the solution to equation (5.33) where L has been replaced by the endogenous $L\left(\mu,\psi,\beta\right)$ from equation (5.30). We then have that $\frac{dL^*}{d\psi} = \frac{\partial L^*}{\partial \psi} + \frac{\partial L^*}{\partial \mu} \frac{\partial \mu^*(\psi)}{\partial \psi} = \frac{\partial L^*}{\partial \widetilde{w}} \frac{\partial \widetilde{w}}{\partial \psi} + \frac{\partial L^*}{\partial \widetilde{w}} \frac{\partial \mu^*(\psi)}{\partial \psi} \frac{\partial \mu^*(\psi)}{\partial \psi}$. From equation (5.16), $\frac{\partial L}{\partial \widetilde{w}} > 0$ for all values of \widetilde{w} . Further, taking the derivatives of the wage in equation (5.28), we find that $\frac{d\widetilde{w}}{d\psi}\Big|_{\pi=0} = \frac{(\alpha-1)C_1r}{\left(\frac{\alpha}{\alpha-1}\right)^{\alpha-1}(\beta+\mu)(r+\beta+\mu)} > 0$ with $(C_1 = \frac{1}{\alpha}\left(\frac{\alpha-1}{\alpha}\right)^{\alpha-1}A^{\alpha}\right)$ and $\frac{d\widetilde{w}}{d\mu}\Big|_{\pi=0} = \frac{-rC_1(\alpha-1)\overline{q}}{\left(\frac{\alpha}{\alpha-1}\right)^{\alpha-1}(r+\beta+\mu)(\beta+\mu)} = -\overline{q} \frac{d\widetilde{w}}{d\psi}\Big|_{\pi=0} < 0$. Finally, $\frac{\partial \mu^*(\psi)}{\partial \psi}\Big|_{\pi=0} = \frac{\frac{\partial L(\mu,\psi,\beta)}{\partial \psi}\Big|_{\pi=0}\frac{1}{\lambda}\left(\frac{\alpha}{\alpha-1}\right)^{-2\alpha}A^{\alpha}\frac{1}{\alpha-1}}{-\left\{-(\eta\overline{q}+1)+\frac{1}{\lambda}\left(\frac{\alpha}{\alpha-1}\right)^{-2\alpha}A^{\alpha}\frac{1}{\alpha-1}}\right\}$ from equation (5.33) with endogenous $L\left(\mu,\psi,\beta\right)$. So

which using that $\frac{\partial \widetilde{w}}{\partial \mu}\Big|_{\pi=0} = -\overline{q} \left. \frac{d\widetilde{w}}{d\psi} \right|_{\pi=0}$ can be rewritten as

Chapter 5. Money growth, employment and growth

$$\frac{dL^*}{d\psi}\Big|_{\pi=0} = \frac{\partial L}{\partial \widetilde{w}}\Big|_{\pi=0} * \frac{\partial \widetilde{w}}{\partial \psi}\Big|_{\pi=0} \left\{ 1 - \overline{q} \frac{-\frac{1}{\overline{q}} \frac{\partial L(\mu,\psi,\beta)}{\partial \mu}\Big|_{\pi=0} \frac{1}{\lambda} \left(\frac{\alpha}{\alpha-1}\right)^{-2\alpha} A^{\alpha} \frac{1}{\alpha-1}}{(\eta \overline{q}+1) - \frac{1}{\lambda} \left(\frac{\alpha}{\alpha-1}\right)^{-2\alpha} A^{\alpha} \frac{1}{\alpha-1} \frac{\partial L(\mu,\psi,\beta)}{\partial \mu}\Big|_{\pi=0}} \right\} > 0$$
since the term in curly brackets is positive given $\frac{\partial L(\mu,\psi,\beta)}{\partial \mu}\Big|_{\alpha=0} < 0.$

Proof of proposition 20. We know from the proof of lemma 8 that $\partial \widetilde{w}/\partial \psi > 0$ for all admissible $\psi < \psi_1$ with $\pi(\psi_1) > 0$ and $\partial \widetilde{w}/\partial \psi < 0$ for all $\psi > \psi_1$ that are compatible with the uniqueness condition (5.37) Since employment L increases monotonically in \widetilde{w} by equation (5.16), the same applies to employment as a function of money growth ψ . From the proof of lemma 13, we further have that given employment $\frac{\partial \mu}{\partial \psi} \geq 0$ for $\psi \leq \psi_0$ with $\pi(\psi_0) = 0$. Hence we have that $d\mu[L(\psi), \psi]/d\psi > 0$ for all $\psi \leq \psi_0$, since here, both the direct and effect of money growth and its indirect effect via employment on economic growth are positive, and $d\mu \left[L(\psi), \psi\right]/d\psi < 0$ for all $\psi \geq \psi_1$, since here both effects are negative. This completes the proof. Proof of proposition 21. From the proof of lemma 13, we know that at $\pi = 0$ ($\pi < 0$), the non-employment-related effects of money growth ψ on economic growth γ are zero (positive). At the same time, from the proof of lemma 8 $\frac{\partial L}{\partial \psi} > 0$ for $\psi < \psi_1$ with $\pi(\psi_1) > 0$. Therefore, the $d\gamma \left[L^*(\psi,\beta),\psi,\beta\right]/d\psi > 0$ for $\pi \leq 0$ and the maximum growth rate is reached at a $\pi > 0$.

Proof of proposition 22. The proposition follows from the facts that first, at $\beta < \infty$ and $\pi = 0$ we have $d\gamma/d\psi > 0$ and $dL/d\psi > 0$ and second, the real outcomes of the models with price rigidity and with flexibility are identical at $\pi = 0$, which can be seen by letting $\beta \to \infty$ or setting $\pi = 0$, respectively, in equations (5.28) and (5.38).

Chapter 6

Concluding remarks

The present thesis has proposed a Schumpeterian growth model with nominal price rigidity for the analysis of the effects of money growth on the real development of an economy in the long run.

Chapter 1 has given an overview over theoretical and empirical results about the influence of money on the real side of the economy in the short and long run. In particular, the aim of the presentation was to motivate our choice of framework for the analysis of the long-run effects of money growth, which has been shown to be a synthesis of the approach of the New Keynesian business cycle literature and the literature concerned with the relationship between inflation and endogenous growth.

In chapter 2, the baseline model has been spelled out in detail. Money's non-superneutrality results from the effects of the money growth rate on relative prices, which affects firms' optimal and effective mark-ups, the level of the real wage and production efficiency. Since these level effects change the incentive to innovate for R&D firms, they affect the economy's growth rate, too.

Chapter 3 has shown that concentrating our long-run analysis on the steady state is justified since the economy converges to the steady state in all examined parameter constellations, i.e. the steady state has been found to be locally stable.

Chapter 4 has relaxed the harsh assumption made in chapter 2 that in-

cumbents' production is terminated when an innovator enters the market. It has been shown that for small quality improvements, which under nonnegative money growth are sufficient to imply innovators cannot drive incumbents out of the market by setting a monopoly price, money growth is more detrimental to growth than under the parameter constellations where monopoly pricing leads to incumbents' making non-positive profits, i.e. the scenario of chapter 2.

Finally, endogenising labour supply in chapter 5 has shown that employment and output are hump-shaped functions of money growth due to the interaction of the latter's effects on production efficiency and the average monopolistic mark-up. The functions peak at positive inflation rates. This has consequences for economic growth: Since output growth depends positively on employment in the model, the economic growth rate is maximised at a positive inflation rate, too.

The central positive goal of the research underlying this thesis was to understand the long-run effects of money growth on the real side of the economy in the presence of nominal price rigidity. The examined model has indeed shown that even moderate price rigidity can lead to a sizeable influence of money growth on employment, output and growth. The identified effects allowed money growth to influence firstly, individual mark-ups, secondly, average mark-ups, i.e. the average level of monopolistic distortions or the real wage level, and thirdly, production efficiency via its effect on price dispersion. All these effects of inflation on relative prices seem plausible and there is some evidence supporting their empirical relevance.¹ Further, the effects of money growth on output growth have turned out to be quantitatively in line with empirical estimates. We therefore believe that our analysis does indeed contribute to the understanding of money's non-superneutrality concerning economic long-run performance.

The main policy implication consistently emerging from our analysis is the clear message that excessive inflation is undesirable both from a growth and from an employment/output perspective. Our calibration exercises showed

¹See section 5.6.

that even inflation rates of around ten percent, which are not abnormal for industrialised countries, significantly reduce output growth, employment and the output level in efficiency units. Therefore, keeping inflation at low levels should be the primary goal of monetary policy concerned with the stimulation of employment and growth. While employment is maximised at a small positive inflation rate, the exact level of inflation to be targeted to maximise growth depends on the influence of the average monopolistic distortion on the incentive to innovate. If this distortion can be offset with other economic policy instruments, then there is no trade-off between the primary monetary policy goal of price stability and the fostering of high economic growth.

While we have throughout the thesis put forward policy recommendations for a monetary policy authority interested in promoting employment and growth, we have not shown that maximising growth is a welfare maximising strategy. In fact it is a priori unclear whether the output growth rate in the decentralised equilibrium is lower than the socially optimal rate that would be chosen by a benevolent central planner.

Empirically, there is some evidence that the social return to R&D is much higher than the private return and that consequently, investment in R&D and innovation-driven growth are lower than socially optimal in real economies.²

Despite this reassuring evidence, it would be a worthwhile extension to augment the model with an explicit welfare analysis that would allow to straightforwardly evaluate the consequences of different money growth rates with regard to the changes they induce in an explicit measure of welfare.

A second suggestion for further research is that it would be desirable to modify the model in a way that allows for the integration of labour supply based on first principles while preserving the model's tractability. While the current approach taken to the endogenisation of labour supply does yield some valuable insights about inflation's influence on the wage and employment, the ad-hoc and short-cut nature of the approach is somewhat unsatisfactory. Spelling out inflation's influence on the household's or labour union's

 $^{^2 \}mathrm{See}$ Jones and Williams [1998] and the references therein.

optimal behaviour would render these insights even more persuasive.

A third interesting extension would take account of the fact that the costs imposed by price rigidity under inflation might induce economic actors to find a way to change the economic environment, i.e. reduce the degree of price rigidity that we take as given in the analysis. A first possibility is the introduction of indexation of contracts, which would reduce the effects of inflation in models with price rigidity.³ However, it has been shown that perfect indexation is not an optimal strategy.⁴ In our context, the most straightforward option would therefore be to allow agents to determine the contract length endogenously against a fixed cost. As briefly discussed in the introductory chapter, this possibility has been explored by Devereux and Yetman [2002] and Graham and Snower [2004] in their analyses of the long-run behaviour of the New Keynesian model with positive steady state inflation. In their simulations, the endogenously chosen contract length implies the persistence of price rigidity – and hence, of the distorting effects of inflation – under all but extreme inflation rates. While we presume that this result carries over to our framework where distortions also affect the growth rate, it would be worthwhile to study the issue in detail.

 $^{{}^{3}}$ In fact, the effects would be eliminated in any deterministic model.

 $^{^{4}}$ See Gray [1978].

Bibliography

- Aghion, P. and Howitt, P. (1992). A model of growth through creative destruction, *Econometrica* 56: 383–396.
- Aghion, P. and Howitt, P. (1998). Endogenous Growth Theory, MIT Press, Cambridge, USA.
- Akerlof, G. A., Dickens, W. T. and Perry, G. L. (2000). Near-rational wage and price setting and the long-run Phillips curve, *Brookings Papers on Economic Activity* pp. 1–44.
- Álvarez, L. J., Dhyne, E., Hoeberichts, M., Kwapil, C., Bihan, H. L., Lünnemann, P., Martins, F., Sabbatini, R., Stahl, H., Vermeulen, P. and Vilmunen, J. (2006). Sticky prices in the Euro area: A summary of new micro evidence, *Journal of the European Economic Association* 4: 575– 584.
- Angeloni, I., Kashyap, A. K., Mojon, B. and Terlizzese, D. (2003). Monetary transmission in the euro area: Does the interest rate channel explain it all? NBER Working Paper No. 9984.
- Ascari, G. (1998). Superneutrality of money in staggered wage-setting models, *Macroeconomic Dynamics* 2: 383–400.
- Ascari, G. (2004). Staggered prices and trend inflation: Some nuisances, *Review of Economic Dynamics* **7**: 642–667.
- Ball, L. and Romer, D. (1990). Real rigidities and the nonneutrality of money, *Review of Economic Studies* 57: 183–203.

Bibliography

- Banerjee, A., Mizen, P. and Russell, B. (2006). Inflation, relative price variability and the markup: Evidence from the United States and the United Kingdom, *Economic modelling* forthcoming.
- Banerjee, A. and Russell, B. (2005). Inflation and measures of the markup, Journal of Macroeconomics 27: 289–306.
- Barro, R. J. (1996). Inflation and growth, Federal Reserve Bank of St. Louis Review 78(3): 153–169.
- Barro, R. J. (1997). Determinants of Economic Growth, MIT Press, Cambridge, Mass.
- Barro, R. J. and Sala-i-Martín, X. (2003). Economic Growth, McGraw-Hill, New York.
- Basu, S. and Fernald, J. G. (1995). Are apparent productive spillovers a figment of specification error?, *Journal of Monetary Economics* 36: 165– 188.
- Basu, S. and Fernald, J. G. (1997). Returns to scale in U.S. production: Estimates and implications, *Journal of Political Economy* **105**: 249–283.
- Bénabou, R. (1992). Inflation and markups, European Economic Review 36: 566–574.
- Benhabib, J. and Farmer, R. E. A. (1999). Indeterminacy and sunspots in macroeconomics, in J. B. Taylor and M. Woodford (eds), Handbook of Macroeconomics, Elsevier, Amsterdam, pp. 387–448.
- Benhabib, J., Schmitt-Grohé, S. and Uribe, M. (2001a). Monetary policy and multiple equilibria, American Economic Review 91(1): 167–186.
- Benhabib, J., Schmitt-Grohé, S. and Uribe, M. (2001b). Appendix to monetary policy and multiple equilibria. unpublished.
- Beyer, A. and Farmer, R. E. A. (2002). Natural rate doubts. European Central Bank Working paper No. 121.

- Bils, M. and Klenow, P. J. (2004). Some evidence on the importance of sticky prices, *Journal of Political Economy* **112**: 947–985.
- Blanchard, O. (1990). Why does money affect output? A survey, in B. Friedman and F. Hahn (eds), Handbook of Monetary Economics, North-Holland, Amsterdam, pp. 779–835.
- Blinder, A. S. (2000). Comment on akerlof, dickens and perry, SNear-rational wage and price setting and the long-run phillips curveS, Brookings Papers on Economic Activity pp. 50–55.
- Blinder, A. S., Canetti, E. R. D. and und Jeremy B. Rudd, D. E. L. (1998). Asking About Prices: A New Approach to Understanding Price Stickiness, Russel Sage Foundation, New York.
- Bruno, M. and Easterly, W. (1998). Inflation crises and long-run growth, Journal of Monetary Economics **41**(1): 3–26.
- Buiter, W. H. (1984). Saddlepoint problems in continuous time rational expectations models: A general method and some macroeconomic examples, *Econometrica* 52: 665–680.
- Bullard, J. (1999). Testing long-run monetary neutrality propositions: Lessons from the recent research, *Federal Reserve Bank of St. Louis Review* 81: 57–77.
- Calvo, G. (1983). Staggered prices in a utility-maximizing framework, Journal of Monetary Economics 12: 383–398.
- Cass, D. (1965). Optimum growth in an aggregative model of capital accumulation, *Review of Economic Studies* **32**: 233–240.
- Chang, W.-Y. (2002). Examining the long-run effect of money on economic growth: An alternative view, *Journal of Macroeconomics* **24**: 81–102.
- Chari, V. V., Jones, L. E. and Manuelli, R. E. (1996). Inflation, growth and financial intermediation, *Federal Reserve Bank of St. Louis Review* **78**(3): 41–57.

- Chari, V. V., Kehoe, P. J. and Mcgrattan, E. R. (2000). Sticky price models of the business cycle: Can the contract multiplier solve the persistence problem?, *Econometrica* **70**(5): 1151–1179.
- Christiano, L., Eichenbaum, M. and Evans, C. (1999). Monetary policy shocks: What have we learned, and to what end, in J. B. Taylor and M. Woodford (eds), *Handbook of Macroeconomics*, Elsevier, Amsterdam, pp. 65–148.
- Christiano, L., Eichenbaum, M. and Evans, C. (2005). Nominal rigidities and the dynamic effects of a shock to monetary policy, *Journal of Political Economy* 113: 1–45.
- Clarida, R., Gali, J. and Gertler, M. (1999). The science of monetary policy: A New Keynesian perspective, *Journal of Economic Literature* 37: 1661– 1707.
- Clark, T. E. (1997). Cross-country evidence on long-run growth and inflation, Economic Inquiry **35**(1): 70–81.
- Clower, R. W. (1967). A reconsideration of the microfoundation of monetary theory, Western Economic Journal 6(1): 1–8.
- Cooley, T. F. and Hansen, G. D. (1989). The inflation tax in a real business cycle model, *American Economic Review* **79**: 733–748.
- Cooley, T. F. and Hansen, G. D. (1995). Money and the business cycle, in T. F. Cooley (ed.), Frontiers of Business Cycle Research, Princeton University Press, Princeton, New Jersey, pp. 175–216.
- Crowder, W. J. (1998). The long-run link between money growth and inflation, *Economic Inquiry* 36: 229–243.
- De Gregorio, J. (1992). The effects of inflation on economic growth, *European Economic Review* **36**: 417–424.
- De Gregorio, J. (1993). Inflation, taxation and long-run growth, *Journal of* Monetary Economics **31**: 271–298.

- Devereux, M. B. and Yetman, J. (2002). Menu costs and the long-run outputinflation trade-off, *Economics Letters* **76**: 95–100.
- Dixit, A. K. and Stiglitz, J. E. (1977). Monopolistic competition and optimum product diversity, *American Economic Review* 67: 297–308.
- Dotsey, M. and Ireland, P. (1996). The welfare cost of inflation in general equilibrium, *Journal of Monetary Economics* **37**: 29–47.
- Eichenbaum, M. and Fisher, J. D. (2003). Testing the Calvo model of sticky prices, *Federal Reserve Bank of Chicago Economic Perspectives* 27: 40– 53.
- Erceg, C. J., Henderson, D. W. and Levin, A. (2000). Optimal monetary policy with staggered wage and price contracts, *Journal of Monetary Economics* 46: 281–313.
- Feenstra, R. C. (1986). Functional equivalence between liquidity costs and the utility of money, *Journal of Monetary Economics* 17: 271–291.
- Fischer, S. (1979). Capital accumulation on the transition path in a monetary optimizing model, *Econometrica* **47**: 1433–1439.
- Fischer, S. (1993). The role of macroeconomic factors in growth, Journal of Monetary Economics 32(3): 485–511.
- Fisher, M. E. and Seater, J. J. (1993). Long-run neutrality and superneutrality in an ARIMA framework, American Economic Review 83(3): 402– 415.
- Friedman, M. (1968). The Role of Monetary Policy, American Economic Review 58(1): 1–21.
- Friedman, M. (1969). The Optimum Quantity of Money and Other Essays, Aldine Publishing Co., Chicago.
- Friedman, M. and Schwartz, A. J. (1963). Monetary History of the United States, 1867-1960, Princeton University Press, Princeton.

- Fuhrer, J. C. (1997). The (Un)Importance of forward-looking behavior in price specifications, Journal of Money, Credit, and Banking 29: 338– 350.
- Fukuda, S.-I. (1996). Money and economic growth with increasing returnsto-scale. Discussion Paper No. 320, Hitotsubashi University.
- Funk, P. and Kromen, B. (2005). Inflation and innovation-driven growth. Cologne Working Paper Series in Economics No. 16.
- Funk, P. and Kromen, B. (2006). Short-term price rigidity in an endogenous growth model: Non-superneutrality and a non-vertical long-run Phillipscurve. Cologne Working Paper Series in Economics No. 29.
- Galí, G. (2003). New perspectives on monetary policy, inflation, and the business cycle, in M. Dewatripont, L. P. Hansen and S. Turnovsky (eds), Advances in Economic Theory and Econometrics, Cambridge University Press, Cambridge, pp. 151–197.
- Galí, J. and Gertler, M. (1999). Inflation dynamics: A structural econometric analysis, *Journal of Monetary Economics* 44: 195–222.
- Galí, J., Gertler, M. and López-Salido, J. D. (2001). European inflation dynamics, *European Economic Review* 45: 1237–1270.
- Gandolfo, G. (1996). *Economic Dynamics*, Springer, Berlin.
- Geweke, J. F. (1986). The superneutrality of money in the United States: An interpretation of the evidence, *Econometrica* 54: 1–22.
- Ghosh, A. and Phillips, S. (1998). Inflation, disinflation, and growth. IMF Working paper 98/68.
- Gillman, M., Harris, M. and Mátyás, L. (2004). Inflation and growth: Explaining a negative effect, *Empirical Economics* **29**(1): 149–167.
- Gillman, M. and Kejak, M. (2005a). Contrasting models of the effect of inflation on growth, *Journal of Economic Surveys* **19**: 113–134.

- Gillman, M. and Kejak, M. (2005b). Inflation and balanced-path growth with alternative payment mechanisms, *The Economic Journal* **115**: 247–270.
- Gomme, P. (1993). Money and growth: Revisited, Journal of Monetary Economics 32: 51–77.
- Goodfriend, M. and King, R. G. (1997). The New Neoclassical Synthesis and the role of monetary policy, NBER Macroeconomic Annual, MIT Press, Cambridge, Massachussetts.
- Graham, L. and Snower, D. J. (2002). The return of the long-run Phillips Curve. IZA Discussion Paper No. 646.
- Graham, L. and Snower, D. J. (2004). The real effects of money growth in dynamic general equilibrium. European Central Bank Working paper No. 412.
- Gray, J. A. (1978). On indexation and contract length, Journal of Political Economy 86: 1–18.
- Grossman, G. M. and Helpman, E. (1991). Quality ladders in the theory of growth, *Review of Economic Studies* 58: 43–61.
- Gylfason, T. and Herbertsson, T. T. (2001). Does inflation matter for growth?, Japan and the World Economy **13**(4): 405–428.
- Hamilton, J. D. (1994). *Time Series Analysis*, Princeton University Press, Princeton, New Jersey.
- Hansen, L. P. and Heckman, J. J. (1996). The empirical foundations of calibration, *Journal of Economic Perspectives* 10: 87–104.
- Hume, D. (1955). Writings in Economics, Nelson, London, chapter On Money, pp. 33–46.
- Ireland, P. N. (2006). The monetary transmission mechanism. Federal Reserve Bank of Boston Working Paper 06-1.

- Itaya, J.-I. and Mino, K. (2003). Inflation, transaction costs and indeterminacy in monetary economies with endogenous growth, *Economica* 70: 451–470.
- Jones, C. I. and Williams, J. C. (1998). Measuring the social return to R &D, Quarterly Journal of Economics 113: 1119–1135.
- Jones, L. E. and Manuelli, R. E. (1995). Growth and the effects of inflation, Journal of Economic Dynamics and Control **19**(8): 1405–1428.
- Judson, R. and Orphanides, A. (1999). Inflation, volatility and growth, International Finance 2(1): 117–138.
- Karanassou, M., Sala, H. and Snower, D. J. (2003). The european Phillips curve: Does the NAIRU exist?, Applied Economics Quarterly 49: 93– 122.
- Karanassou, M., Sala, H. and Snower, D. J. (2005). A reappraisal of the inflation-unemployment tradeoff, *European Journal of Political Economy* 21: 1–32.
- Khan, M. S. and Senhadji, A. S. (2001). Threshold effects in the relationship between inflation and growth, *IMF Staff Papers* **48**(1): 1–21.
- Kim, J. (2000). Constructing and estimating a realistic optimizing model of monetary policy, *Journal of Monetary Economics* 45: 329–359.
- Kimball, M. S. (1995). The quantitative analytics of the basic neomonetarist model, Journal of Money, Credit and Banking 27: 1241–1277.
- King, R. G. and Plosser, C. I. (1984). Money, credit, and prices in a real business cycle, *American Economic Review* 74: 363–380.
- King, R. G. and Watson, M. W. (1994). The post-war U.S. Phillips curve: A revisionist econometric history, *Carnegie-Rochester Conference Series* on Public Policy 41: 157–219.

- King, R. G. and Watson, M. W. (1997). Testing long-run neutrality, Federal Reserve Bank of Richmond Economic Quarterly pp. 69–101.
- King, R. G. and Wolman, A. L. (1996). Inflation targeting in a St. Louis model of the 21st century, *Federal Reserve Bank of St. Louis Quarterly Review* pp. 83–107.
- King, R. G. and Wolman, A. L. (1998). What should the monetary authority do when prices are sticky?, in J. Taylor (ed.), Monetary Policy Rules, University of Chicago Press, Chicago, pp. 349–398.
- Koopmans, T. C. (1965). On the Concept of Optimal Economic Growth, North-Holland, Amsterdam.
- Koustas, Z. and Serletis, A. (2003). Long-run Phillips-type trade-offs in European countries, *Economic Modelling* 20: 679–701.
- Leith, C. and Wren-Lewis, S. (2000). Interactions between monetary and fiscal policy rules, *Economic Journal* 110: C93–C108.
- Levine, R. and Renelt, D. (1992). A sensitivity analysis of cross-country growth regressions, *American Economic Review* 82: 942–963.
- Levine, R. and Zervos, S. J. (1993). What have we learned about policy and growth from cross-country regressions?, *American Economic Review* 83: 426–430.
- Linnemann, L. (1999). Sectoral and aggregate estimates of the cyclical behavior of markups: Evidence from germany, Weltwirtschaftliches Archiv (Review of World Economics) 135: 480–500.
- Lucas, R. E. (1980). Methods and problems in business cycle theory, Journal of Money, Credit and Banking 12: 696–715.
- Lucas, R. E. (1996). Nobel lecture: Monetary neutrality, Journal of Political Economy 104(4): 661–682.

- Mankiw, N. G. (1990). A quick refresher course in macroeconomics, Journal of Economic Literature 28: 1645–1660.
- Marquis, M. H. and Reffett, K. L. (1994). New technology spillovers into the payment system, *The Economic Journal* 104: 1123–1138.
- Marquis, M. H. and Reffett, K. L. (1995). Inflation taxes in a convex model of equilibrium growth, *Economica* 62: 109–122.
- McCallum, B. T. (1990). Inflation: Theory and evidence, in B. M. Friedman and F. H. Hahn (eds), Handbook of Monetary Economics, Elsevier, Amsterdam, pp. 963–1012.
- McCandless, G. T. J. and Weber, W. E. (1995). Some monetary facts, *Federal Reserve Bank of Minneapolis Quarterly Review* **19**(3).
- Motley, B. (1998). Growth and inflation: A cross-country study, *Federal Reserve Bank of San Francisco Economic Review* 1/98: 15–28.
- Orphanides, A. and Solow, R. (1990). Money, inflation and growth, in B. M. Friedman and F. H. Hahn (eds), Handbook of Monetary Economics, North Holland, Amsterdam, pp. 223–261.
- Parks, R. W. (1978). Inflation and relative price variability, Journal of Political Economy 86: 79–95.
- Pecorino, P. (1995). Inflation, human capital accumulation and long-run growth, *Journal of Macroeconomics* 17: 533–542.
- Peersman, G. and Smets, F. (2003). The monetary transmission mechanism in the euro area: More evidence from VAR analysis, in I. Angeloni, A. Kashyap and B. Mojon (eds), Monetary Policy Transmission in the Euro Area, Cambridge University Press, Cambridge.
- Phelps, E. S. (1967). Phillips curves, expectations of inflation and optimal unemployment over time, *Economica* 34: 254–281.

- Phillips, A. W. (1958). The Relation between Unemployment and the Rate of Change of Money Wage rates in the United Kingdom, 1861-1957, *Economica* 25: 343–351.
- Ragan, C. (1998). On the believable benefits of low inflation. Bank of Canada Working Paper 98-15.
- Ramsey, F. P. (1928). A mathematical theory of saving, *Economic Journal* 38: 543–559.
- Roberts, J. M. (1995). New Keynesian economics and the Phillips Curve, Journal of Money, Credit and Banking 27: 975–984.
- Romer, D. (2001). Advanced Macroeconomics, McGraw-Hill, New York.
- Romer, P. M. (1990). Endogenous technological change, Journal of Political Economy 98(5 pt. 2): S71–S102.
- Rotemberg, J. (1982). Sticky prices in the United States, Journal of Political Economy 90: 1187–1211.
- Russell, B. and Banerjee, A. (2006). The long-run Phillips Curve and nonstationary inflation. EUI ECO Working paper 2006-16.
- Samuelson, P. A. and Solow, R. M. (1960). Analytical aspects of anti-inflation policy, American Economic Review, Papers and Proceedings 50: 177– 194.
- Sarel, M. (1996). Nonlinear effects of inflation on economic growth, *IMF Staff Papers* 43(1): 199–215.
- Sbordone, A. M. (2002). Prices and unit labor costs: A new test of price stickiness, Journal of Monetary Economics 49: 265–292.
- Serletis, A. and Koustas, Z. (1998). International evidence on the neutrality of money, *Journal of Money*, *Credit and Banking* **30**: 1–25.

- Setterfield, M. and LeBlond, K. (2003). The Phillips curve and US macroeconomic performance during the 1990s, *International Review of Applied Economics* 17: 361–376.
- Sidrauski, M. (1967). Rational choice and patterns of growth in a monetary economy, *American Economic Review* **57**(2): 534–544.
- Sims, C. (1992). Interpreting the macroeconomic time series facts: The effects of. monetary policy, *European Economic Review* **36**: 975–1011.
- Solow, R. M. (1956). A contribution to the theory of economic growth, Quarterly Journal of Economics **70**(1): 65–94.
- Stock, J. H. and Watson, M. W. (2000). Business cycle fluctuations in US macroeconomic time series, in J. B. Taylor and M. Woodford (eds), *Handbook of Macroeconomics*, Vol. 1a, North Holland, Amsterdam, chapter 1, pp. 3–64.
- Stockman, A. (1981). Anticipated inflation and the capital stock in a cashin-advance economy, *Journal of Monetary Economics* 8: 387–393.
- Swan, T. W. (1956). Economic growth and capital accumulation, *Economic Record* 32: 334–361.
- Taylor, J. B. (1980). Aggregate dynamics and staggered contracts, *Journal of Political Economy* 88: 1–23.
- Taylor, J. B. (1999a). Monetary policy guidelines for employment and inflation stability, in R. M. Solow and J. B. Taylor (eds), Inflation, Unemployment, and Monetary Policy, MIT press, Cambridge, Massachusetts, pp. 29–54.
- Taylor, J. B. (1999b). Staggered Price and Wage Setting in Macroeconomics, in J. Taylor and M. Woodford (eds), Handbook of Macroeconomics, Elsevier Science B.V., Amsterdam, chapter 15, pp. 1009–1050.
- Temple, J. (2000). Inflation and growth: Stories short and tall, Journal of Economic Surveys 14(4): 395–426.

- Tobin, J. (1965). Money and economic growth, *Econometrica* **33**: 671–684.
- Walsh, C. E. (2003). Monetary Theory and Policy, The MIT Press, Cambridge, Massachusetts.
- Wang, P. and Yip, C. (1992). Examining the long-run effect of money on economic growth, *Journal of Macroeconomics* 14: 359–369.
- Weber, A. A. (1994). Testing long-run neutrality: Empirical evidence for g7 countries with special emphasis on germany, *Carnegie-Rochester Conference Series on Public Policy* **41**: 67–117.
- Wolman, A. L. (1999). Sticky prices, marginal cost, and the behavior of inflation, Federal Reserve Bank of Richmond Economic Quarterly 85: 29–48.
- Wolman, A. L. (2000). The frequency and costs of individual price adjustment, Federal Reserve Bank of Richmond Economic Quarterly 86: 1–22.
- Woodford, M. (2003). *Interest and Prices*, Princeton University Press, Princeton, New Jersey.
- Yun, T. (1996). Nominal price rigidity, money supply endogeneity, and business cycles, *Journal of Monetary Economics* 37: 345–370.